Popular and Democratic Republic of Algeria Ministry of the Higher Education and Scientific Research Djillali LIABBES University - SIDI BEL-ABBES -PoBox 89, 22000 SBA, Algeria Tel/Fax 048 54 43 44

THESIS

Prepared at the Mathematics Department of the Sciences Faculty

by

Djillali AOUED

To obtain

the rank of $\operatorname{\textbf{DOCTOR}}$

Speciality: Mathematics Option : Functional Analysis and Differential Equations (A. F. E. D.)

Existence of Solutions for Evolution Equations and Inclusions with Delay

Thesis presented and supported publicly on //2017, in front of the Jury made up of

Abdelghani OUAHAB	Prof	University of SIDI BEL-ABBES	$\mathbf{President}$
Saïd ABBAS	MCA	University of SAÏDA	Examinator
Sidi Mohammed BOUGUIMA	Prof	University of TLEMCEN	Examinator
Boualem Attou SLIMANI	MCA	University of TLEMCEN	Examinator
Mouffak BENCHOHRA	Prof	University of SIDI BEL-ABBES	Co-Supervisor
Selma BAGHLI-BENDIMERAD	Prof	University of SIDI BEL-ABBES	Supervisor

Dédicaces

Remerciements et louanges à Dieu Tout Puissant pour m'avoir donné la foi et la force d'accomplir ce modeste travail.

Prière et Salut soient sur Notre Cher Maître & Prophète "Mohammed " et sur sa famille et ses fidèles compagnons.

Je dédie ce modeste travail à :

• Ma douce mère en reconnaissance des sacrifices qu'elle a fait pour nous tracer un chemin dans cette vie. Que vie nous donne temps pour la remercier ! C'est grâce à son amour infini, sa patience, son inestimable aide et ses conseils que ma vie s'est construite !

• Mon formidable frère et guide Mohamed qui m'a ouvert les yeux sur la science et l'histoire et m'a poussé dans le domaine de recherches et du travail et qui n'a pas cessé de m'encourager tout au long de mon parcours.

• Je tiens également à mentionner et à témoigner ma reconnaissance à tous mes amis membres du laboratoire de Mathématiques et particulièrement mes camarades d'études pour leurs encouragements et leur gentillesse envers moi.

• Mes vifs remerciements vont également à tous mes amis intimes pour l'appui moral qu'ils m'ont témoigné et les encouragements qu'ils m'ont offerts. Les moments de travail que nous avons passés ensemble sont inoubliables.

• Pour terminer, j'adresse mon grand amour à mes chères parents, mes frères et ma chère sœur. C'est grâce à leurs amours et leurs sacrifices que ce mémoire a été mené à bout enfin. Mon plus grand souhait dans cette vie, c'est de les voir toujours à côté de moi, en bonne santé, heureux et que la paix soit avec eux.

Remerciements

Tout travail réussi dans la vie nécessite en premier lieu la Bénédiction de Dieu, et ensuite l'aide et le support de plusieurs personnes. Je tiens donc à remercier et à adresser ma reconnaissance à toute personne qui m'a aidé de loin ou de près afin de réaliser ce travail.

J'exprime ici ma profonde reconnaissance à l'égard de ma promotrice Mme Selma BAGHLI-BENDIMERAD. Elle a su orienter mon travail sur l'immense champ d'actualité de recherche. Les conseils et encouragements qu'elle n'a jamais cessé de prodiguer sont inestimables. Sa patience et sa compréhension m'ont permis d'avancer et de terminer ce travail.

Que le *Prof. Mouffak BENCHOHRA* trouve ici l'expression de mes remerciements pour l'intérêt qu'il a porté à ce travail. Je le remercie vivement d'avoir été mon co-encadreur il m'aidé et conseillé tout le long de ce travail.

J'adresse mes vives remerciements à *Prof. Abdelghani OUAHAB* pour ses conseils et intérêts durant toute ma formation. Je le remercie vivement d'avoir accepté de présider le jury.

Je remercie le *Prof. Sidi Mohammed BOUGUIMA* pour avoir accepté d'être membre de ce jury et d'avoir accepter d'examiner mon travail.

Que le *Dr. Saïd ABBAS* trouve ici mes sincères remerciements pour avoir accepté d'être membre de ce jury et d'avoir accepter d'examiner mon travail.

Je remercie aussi le *Dr. Boualem Attou SLIMANI* pour avoir accepté d'être membre de ce jury et d'avoir accepter d'examiner mon travail.

God created in man the ability to think ... Man's ability to think may try to obtain the desired information by combining the universals with each other, with the result that the mind obtains a universal picture that conforms to details outside. " ...

Ibn Khaldûn The Muqqaddimah.

The Science of Logic.

Abstract

In this work, we give a contribution to the study of existence and uniqueness results of mild solutions on a bounded interval for various classes of first order class and Caputo's fractional derivative order class of partial functional and neutral functional type, differential and integrodifferential, perturbed and nonperturbed *evolution* equations and inclusions with finite and infinite state-dependent delay.

To get the existence of these mild solutions ; sufficient conditions are considered in the study of different classes of *evolution* problems. Uniqueness results are also given for some classes of these problems.

The method used is to reduce the existence of these mild solutions to the search for the existence of fixed points of appropriate operators by applying different nonlinear alternatives in Fréchet and Banach spaces to entire the existence of fixed points of the above operator which are mild solutions of our problems. This method is based on famous and recent fixed point theorems and is combined with the semigroup theory.

Controllability of mild solutions is investigated for some classes of first order and fractional order of partial functional and neutral functional *evolution* equations in this work as applications.

Key words and phrases :

Evolution equations and inclusions with delay - neutral problems - Fractional derivative order - perturbed evolution problems - mild solution - state-dependent delay - existence uniqueness - controllability - fixed point - nonlinear alternative - semigroup.

AMS Subject Classification : 26A33 - 93B05 - 34G20 - 34G25 - 34K26 - 34K37 - 34K40 - 35R11.

Résumé

Ce travail est consacré à l'étude de quelques résultats d'existence et d'unicité de la solution faible sur un intervalle borné pour quelques classes d'ordre un et d'ordre fractionnaire de type Caputo d'équations et d'inclusions d'évolution partielles fonctionelles et celles de type neutre, différentielles et intégrodifférentielles, perturbées et non perturbées, avec un retard fini et infini dépendant de l'état.

Pour obtenir l'existence de ces solutions faibles ; des conditions suffisantes seront considérées dans l'étude des différentes classes de problèmes d'évolution. Des résultats d'unicité sont également donnés pour quelques classes de ces problèmes.

La méthode utilisée est de ramener la recherche de l'existence de ces solutions faibles à la recherche de l'existence des points fixes d'opérateurs appropriés en appliquant différentes alternatives non linéaires dans les espaces de Fréchet et de Banach pour entirer l'existence des points fixes de cet opérateur qui sont les solutions faibles de nos problèmes. Cette méthode est basée sur des célèbres et récents théorèmes du point fixe et est combinée avec la théorie des semi-groupes.

La contrôlabilité des solutions faibles est donnée dans ce travail pour quelques classes d'ordre fractionnaire d'équations d'*évolution* et celles de type neutre à titre d'applications.

Mots et Phrases Clefs :

Equations et inclusions d'évolution à retard - problèmes de type neutre - Ordre fractionnaire - Problèmes d'évolution perturbées - solution faible - retard dépendant de l'état - existence unicité - contrôlabilité - point fixe - alternative non linéaire - semi-groupes.

Classification AMS : 26A33 - 93B05 - 34G20 - 34G25 - 34K26 - 34K37 - 34K40 - 35R11.

Publications

1. **D. Aoued** and S. Baghli-Bendimerad, Mild solutions for perturbed evolution equations with infinite state-dependent delay, *Electronic Journal of Qualitative Theory of differential Equations*, **59**, 2013, 1-24.

 $\label{eq:http://www.math.u-szeged.hu/ejqtde/periodica.html?periodica=1¶mtipus ertek=publication¶m ertek=2328$

 D. Aoued and S. Baghli-Bendimerad, Controllability of mild solutions for evolution equations with infinite state-dependent delay, *European Journal of Pure and Applied Mathematics*, 9 (4), 2016, 383-401.

http://www.ejpam.com/index.php/ejpam/article/view/2456/485

- 3. **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Fractional partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.
- 4. **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Fractional perturbed partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.
- 5. **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Controllability of mild solutions for fractional partial functional and neutral functional evolution equations with state-dependent delay, submitted.
- 6. **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Multivalued fractional partial functional and neutral functional fractional evolution equations with state-dependent delay, submitted.

Contents

.

In	trod	uction	11
1	Pre	liminaries	19
	1.1	Notations and Definitions	19
	1.2	Some Properties in Fréchet Spaces	20
	1.3	Evolution System Generator	20
	1.4	Definition of Mild Solution	21
	1.5	Controllability of Mild Solution	22
	1.6	Phase Space $\overset{\circ}{\mathcal{B}}$	22
	1.7	State-Dependent Delay	24
	1.8	Some Properties of Fractional Calculs	25
	1.9	Definition of Multivalued Functions	26
	1.10	Fixed Point Theorems	27
-	_		• •
2		ctional Evolution Equations with State-Dependent Delay	29
	2.1		29
	2.2	Neutral Droblem with Finite Delay	3U 99
	2.3		33 90
	2.4	Partial Problem with Infinite Delay	30
	2.5	Neutral Problem with Infinite Delay	39
	2.0		43
3	Frac	ctional Integrodifferential Evolution Equations with State-Dependent De-	
	lay		47
	3.1	Introduction	47
	3.2	Partial Problem with Finite Delay	48
	3.3	Neutral Problem with Finite Delay	51
	3.4	Partial Problem with Infinite Delay	54
	3.5	Neutral Problem with Infinite Delay	58
	3.6	Examples	62
Δ	Ро	rturbed Evolution Equations with State-Dependent Delay	67
Т	4 1	Introduction	67
	ч.1 4 9	Partial Problem with Infinite Delay	68
	1.2 4.3	Neutral Problem with Infinite Delay	- 74
	ч.9 Д Д	Fractional Partial Problem with Finite Delay	80
	ч.ч 4 5	Fractional Neutral Problem with Finite Delay	83

- -

	4.6	Fractional Partial Problem with Infinite Delay	87
	4.7	Fractional Neutral Problem with Infinite Delay	93
	4.8	Examples	97
5	Con	trollability Results for Evolution Equations with State-Dependent Delay	105
	5.1	Introduction	105
	5.2	Partial Problem with Infinite Delay	107
	5.3	Neutral Problem with Infinite Delay	112
	5.4	Fractional Partial Problem with Finite Delay	116
	5.5	Fractional Neutral Problem with Finite Delay	120
	5.6	Fractional Partial Problem with Infinite Delay	124
	5.7	Fractional Neutral Problem with Infinite Delay	132
	5.8	Examples	137
6	Frac	ctional Evolution Inclusions with State-Dependent Delay	143
6	Frac 6.1	ctional Evolution Inclusions with State-Dependent Delay Introduction	143 143
6	Frac 6.1 6.2	ctional Evolution Inclusions with State-Dependent Delay Introduction	143 143 144
6	Frac 6.1 6.2 6.3	ctional Evolution Inclusions with State-Dependent DelayIntroduction	143 143 144 148
6	Frac 6.1 6.2 6.3 6.4	ctional Evolution Inclusions with State-Dependent Delay Introduction	143 143 144 148 152
6	Frac 6.1 6.2 6.3 6.4 6.5	ctional Evolution Inclusions with State-Dependent Delay Introduction	143 143 144 148 152 158
6	Frac 6.1 6.2 6.3 6.4 6.5 6.6	Control Evolution Inclusions with State-Dependent Delay Introduction	143 143 144 148 152 158 162
6 Co	Frac 6.1 6.2 6.3 6.4 6.5 6.6 onclu	Ctional Evolution Inclusions with State-Dependent Delay Introduction	 143 143 144 148 152 158 162 167
6 Cd Pe	Frac 6.1 6.2 6.3 6.4 6.5 6.6 onclu	ctional Evolution Inclusions with State-Dependent Delay Introduction	 143 143 144 148 152 158 162 167 169
6 Ca Pe Bi	Frac 6.1 6.2 6.3 6.4 6.5 6.6 onclu erspe bliog	ctional Evolution Inclusions with State-Dependent Delay Introduction	 143 143 144 148 152 158 162 167 169 170

Introduction

Fractional calculus deals with the generalization of the integrals and derivatives of noninteger order. The idea of fractional calculus and fractional order differential equations has been a subject of interest not only among mathematicians, but also among physicists and engineers. Indeed, we can find numerous applications in rheology, control, porous media, viscoelasticity, electrochemistry, electromagnetism, etc... see the books of Baleanu *et al.* [28], Hilfer [71], Oldham and Spanier [88], Podlubny [91] and Tarasov [95].

There has been a significant development in ordinary and partial fractional differential equations in recent years; see the monographs of Miller and Ross [87], Samko *et al.* [93], Podlubny [91], the papers of Abbas and Benchohra [2, 3], Benchohra *et al.* [29, 31], Kilbas and Marzan [78], Vityuk and Golushkov [97], Vityuk and Mykhailenko [98] and the references therein.

In fact, fractional differential equation is considered as an alternative model to nonlinear differential equations by Bonilla [35]. Though the concepts and the calculus of fractional derivative are few centuries old, it is realized only recently that these derivatives form an excellent framework for modeling real world problems. This in turn led to sustained study of the theory of fractional equations by Lakshmikantham *et al.* [82]. Numerical experiments for fractional models on population dynamics are discussed by Luchko and his collaborators in [86].

The first work, devoted exclusively to the subject of fractional calculus, is the book by Oldham and Spanier [88] published in 1974. One of the most recent works on the subject of fractional calculus is the book of Podlubny [91] published in 1999, which deals principally with fractional differential equations. Some of the lasted (but certainly not the last) works especially on fractional models of anomalous kinetics of complex process are the volumes edited by Carpinteri and Mainardi [39] in 1997 and by Hilfer [71] in 2000, and the book by Zaslavsky [102] published in 2005. Indeed, in the meantime, numerous other works (books, edited volumes, and conference proceeding) have also appeared. These include (for example) the monographs of Samko *et al.* [93], which was published in Russian in 1987 and in English in 1993, the book of Miller and Ross [87] in 1993, and recently the books of Baleanu and his collaborators [28], Diethelm [46], Kilbas *et al.* [79], Lakshmikantham and his collaborators [82] and Tarasov [95].

Neutral differential equations arise in many areas of applied mathematics and for this reason these equations have received much attention in last few decades. A good guide to the literature for neutral functional differential equations is in the books by Hale *et al.* [61, 62], Kolmanovskii and Myshkis [81] and the references therein.

Differential delay equations or functional differential equations, have been used in modeling scientific phenomena for many years. Often, it has been assumed that the delay is either a fixed constant or is given as an integral in which case it is called a distributed delay; see for instance the books of Benchohra and his collaborators [30], Hale *et al.* [62, 60], Hino and his collaborators [74], Kolmanovskii and Myshkis [81], Lakshmikantham *et al.* [83, 41] and Wu in [100].

When the delay is infinite, the notation of the phase space \mathcal{B} plays an important role in the study of both qualitative and quantitative theory for functional differential equations. A usual choice is a semi-normed space satisfying suitable axioms, which was introduced by Hale and Kato (see [60]). For further applications see for instance the books of Hale [62], Hino [74], Lakshmikantham *et al.* [83] and their references.

However, complicated situations in which the delay depends on the unknown functions have been proposed in modeling in recent years (see for instance Arino *et al.* [17], Rezounenko and Wu [92], Willé and Baker [99] and the references therein). These equations are frequently called equations with state-dependent delay. Existence results, among other things, were derived recently for various classes of functional differential equations when the delay is depending on the solution. We refer the reader to the papers by Ait Dads and Ezzinbi [9], Gyri and Hartung [58], Hartung [63, 64, 65] and Hernandez *et al.* [69]. Darwish and Ntouyas considered in [43] a class of semilinear functional fractional order differential equations with state-dependent delay. By means of the Banach contraction principle and the nonlinear alternative of Leray-Schauder, Abada *et al.* in [4] present some existence as well as uniqueness results for each of our problems on bounded domain.

Baghli *et al.* in [5], [20]-[26] considered existence, uniqueness and controllability of mild solutions for first order classes of semilinear partial functional and neutral functional differential and integrodifferential perturbed and nonperturbed evolution equations and inclusions with finite and infinite delay. Then they look in [27] to the case where the delay is depending on the solution for evolution equations and in [19] she gives global mild solution for evolution inclusions with state-dependent delay.

We are going here to give the existence of mild solution for first order of the perturbed class of evolution equations with state-dependent delay in [10] and we are going to study the controllability of these mild solution in [11]. Then we extend the above results for Caputo derivative fractional order of different classes of evolution equations and inclusions with state-dependent delay in [12]-[15].

So in this thesis, we give existence and even uniqueness of mild solutions on a bounded interval for different classes of the first order and the Caputo's fractional derivative order for semilinear partial functional and neutral fractional, differential and integrodifferential, perturbed and nonperturbed of *evolution* equations and inclusions with finite and infinite state-dependent delay in Fréchet and Banach spaces. Our results are based upon fixed point techniques combined with semigroup theory.

Our thesis is organized as follows:

The first Chapter contains some notations, definitions, theorems and preliminary facts that will be used throughout this thesis.

In what follows each chapter concludes with examples applying the abstract theory and this thesis is clotured by the bibliography.

Introduction

The second Chapter is devoted to give our main result, using a nonlinear alternative of Leary Schauder-type given by Frigon and Granas for contraction maps [52], combined with the semi-group theory. The existence of the unique mild solution is demonstrated in section 2.2 for the following class of fractional evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t,y_{t})}), \quad 0 < \alpha < 1 \text{ a.e. } t \in J := [0, b]$$

$$(1)$$

$$y(t) = \varphi(t), \quad t \in H := [-r, 0],$$
(2)

where b > 0, $0 < r < +\infty$, $^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative order of order $\alpha \in (0,1)$, $f: J \times C(H; E) \to E$, $\rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{0 \le t}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \le t$.

An extension of this problem for the neutral type is given in section 2.3 for the following class of fractional neutral evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J$$
(3)

$$y(t) = \varphi(t), \quad t \in H, \tag{4}$$

where $A(\cdot)$, f and φ are as in problem (2.1) – (2.2) and $g: J \times C(H; E) \to E$ is a given function.

In section 2.4 we study the fractional evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \text{ a.e. } t \in J$$
(5)

$$y_0 = \phi \in \mathcal{B},\tag{6}$$

where \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f: J \times \mathcal{B} \to E, \rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t, s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 2.5 for the following class of fractional neutral evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J$$
(7)

$$y_0 = \phi \in \mathcal{B},\tag{8}$$

where $A(\cdot)$, f and ϕ are as in problem (2.5) – (2.6) and $g: J \times \mathcal{B} \to E$ is a given function.

The third Chapter is devoted to give our main result, using a nonlinear alternative of Frigon and Granas for contraction maps [52]. The existence of the unique mild solution is demonstrated in section 3.2 for the following class of fractional integrodifferential evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t,s)f(s,y_{\rho(s,y_{s})})ds, \quad 0 < \alpha < 1 \text{ a.e. } t \in J$$
(9)

$$y(t) = \varphi(t), \quad t \in H, \tag{10}$$

where $\mathcal{K}: J \times J \to E$, $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1), f: J \times C(H; E) \to E, \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 3.3, we consider the following fractional neutral integrodifferential evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t, s)f(s, y_{\rho(s, y_{s})})ds, \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J \quad (11)$$

$$y(t) = \varphi(t), \quad t \in H, \tag{12}$$

where $A(\cdot)$, f and φ are as in problem (3.1) – (3.2) and $g: J \times C(H; E) \to E$ is a given function.

In section 3.4 we study the following fractional integrodifferential evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t,s)f(s,y_{\rho(s,y_{s})})ds, \quad 0 < \alpha < 1 \text{ a.e. } t \in J$$
(13)

$$y_0 = \phi \in \mathcal{B},\tag{14}$$

where $\mathcal{K}: J \times J \to E$, \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1)$, $f: J \times \mathcal{B} \to E$, $\rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 3.5, we consider the following fractional neutral integrodifferential evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t, s)f(s, y_{\rho(s, y_{s})})ds, \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J \quad (15)$$
$$y_{0} = \phi \in \mathcal{B}, \tag{16}$$

where $A(\cdot)$, f, \mathcal{K} and ϕ are as in problem (3.5) – (3.6) and $q: J \times \mathcal{B} \to E$ is a given function.

The fourth Chapter is devoted to study the perturbed classes using a nonlinear alternative of Avramescu for contractions maps in Fréchet spaces [18]. The existence of mild solutions on the positif real line is demonstrated in section 4.2 for the following class of first order perturbed evolution equations with infinite state-dependent delay

$$y'(t) = A(t)y(t) + f(t, y_{\rho(t,y_t)}) + h(t, y_{\rho(t,y_t)}), \quad \text{a.e.} \ t \in \mathbb{R}^+,$$
(17)

$$y_0 = \phi \in \mathcal{B} \tag{18}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $f, h : \mathbb{R}^+ \times \mathcal{B} \to E, \rho : \mathbb{R}^+ \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t\geq 0}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in\mathbb{R}^+\times\mathbb{R}^+}$ for $s \leq t$.

Introduction

An extension of this problem is given in section 4.3, we consider the following first order neutral perturbed evolution equations with infinite state-dependent delay

$$\frac{d}{dt}[y(t) - g(t, y_{\rho(t, y_t)})] = A(t)y(t) + f(t, y_{\rho(t, y_t)}) + h(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \quad t \in \mathbb{R}^+,$$
(19)

$$y_0 = \phi \in \mathcal{B} \tag{20}$$

where $A(\cdot)$, f and ϕ are as in problem (4.1) – (4.2) and $g: \mathbb{R}^+ \times \mathcal{B} \to E$ is a given function.

In section 4.4, we study the following fractional perturbed evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t,y_{t})}) + h(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \ t \in J = [0, b],$$
(21)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0], \tag{22}$$

where $b > 0, 0 < r < +\infty, {}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f, h : J \times C(H; E) \to E, \rho : J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t, s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 4.5, we consider the following fractional neutral perturbed evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}) + h(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J,$$
(23)

$$y(t) = \varphi(t), \quad t \in H, \tag{24}$$

where $A(\cdot)$, f and ϕ are as in problem (4.5) – (4.6) and $g: J \times \mathcal{B} \to E$ is a given function.

In section 4.6, we study the following fractional perturbed evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t,y_{t})}) + h(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \ t \in J,$$
(25)

$$y_0 = \phi \in \mathcal{B} \tag{26}$$

where \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f, h: J \times \mathcal{B} \to E, \rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t, s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 4.7, we consider the following fractional neutral perturbed evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}) + h(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J,$$
(27)

$$y_0 = \phi \in \mathcal{B} \tag{28}$$

where $A(\cdot)$, f and ϕ are as in problem (4.9) – (4.10) and $g: J \times \mathcal{B} \to E$ is a given function.

The fifth Chapter is devoted to controllability problems, using a nonlinear alternative of Avramescu for contractions maps in Banach spaces [18]. The controllability of mild solutions

over the real line is demonstrated in section 5.2 for the following class of first order evolution equations with infinite state-dependent delay

$$y'(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \ t \in \mathbb{R}^+,$$
(29)

$$y_0 = \phi \in \mathcal{B} \tag{30}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $f : \mathbb{R}^+ \times \mathcal{B} \to E, \rho : \mathbb{R}^+ \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions, the control function u(.) is given in $L^2(\mathbb{R}^+E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|, C$ is a bounded linear operator from E into E and $\{A(t)\}_{t\in\mathbb{R}^+}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in\mathbb{R}^+\times\mathbb{R}^+}$ for $s \leq t$.

An extension of this problem is given in section 5.3, we consider the controllability of mild solutions over the real line for the following first order neutral evolution equations with infinite state-dependent delay

$$\frac{d}{dt}[y(t) - g(t, y_{\rho(t, y_t)})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \quad t \in \mathbb{R}^+,$$
(31)

$$y_0 = \phi \in \mathcal{B} \tag{32}$$

where $A(\cdot)$, f and ϕ are as in problem (29) – (30) and $g: \mathbb{R}^+ \times \mathcal{B} \to E$ is a given function.

The controllability of mild solutions on a bounded interval is demonstrated in section 5.4 for the following class of fractional evolution equations with finite state-dependent delay

$$^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t,y_t)}), \quad \text{a.e.} \quad t \in J = [0, b]$$
(33)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0], \tag{34}$$

where $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f: J \times C(H; E) \to E, \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions, the control function u(.) is given in $L^{2}(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|$, C is a bounded linear operator from E into Eand $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 5.5, we consider the controllability of mild solutions of the following fractional neutral evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(35)

$$y(t) = \varphi(t), \quad t \in H, \tag{36}$$

where $A(\cdot)$, f, u, C and φ are as in problem (5.5) – (5.6) and $g: J \times C(H; E) \to E$ is a given function.

In section 5.6, we study the controllability of mild solutions of the following fractional evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t,y_t)}), \quad \text{a.e.} \quad t \in J$$
(37)

$$y_0 = \phi \in \mathcal{B},\tag{38}$$

where ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1)$, $f: J \times \mathcal{B} \to E$ and $\phi \in \mathcal{B}$ are given functions, the control function u(.) is given in $L^{2}(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|$, Cis a bounded linear operator from E into E and $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 5.7, we consider the controllability of mild solutions of the following fractional neutral evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(39)

$$y_0 = \phi \in \mathcal{B},\tag{40}$$

where $A(\cdot)$, f, u, C and ϕ are as in problem (5.9) – (5.10) and $g: J \times \mathcal{B} \to E$ is a given function.

The sixth Chapter is devoted to give our main result, using a nonlinear alternative of Frigon contraction multivalued maps [50]. The existence of mild solutions on a bounded interval is demonstrated in section 6.2 for the following class of fractional evolution inclusion with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) \in A(t)y(t) + F(t, y_{\rho(t,y_t)}), \quad \text{a.e. } t \in J$$
(41)

$$y(t) = \varphi(t), \quad t \in H, \tag{42}$$

where ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1), F: J \times C(H; E) \to \mathcal{P}(E)$ is a multivalued map with nonempty compact values, $\mathcal{P}(E)$ is the family of all subsets of $E, \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t\geq 0}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 6.3, we consider the following fractional neutral evolution inclusion with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t)[y(t) - g(t, y_{\rho(t, y_{t})})] \in A(t)y(t) + F(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(43)

$$y(t) = \varphi(t), \quad t \in H, \tag{44}$$

where $A(\cdot)$, F and φ are as in problem (6.1) – (6.2) and $g: J \times C(H; E) \to E$ is a given function.

In section 6.4, we study the following fractional evolution inclusions with infinite statedependent delay

$${}^{c}D_{0}^{\alpha}y(t) \in A(t)y(t) + F(t, y_{\rho(t,y_{t})}), \quad \text{a.e. } t \in J$$
(45)

$$y_0 = \phi \in \mathcal{B},\tag{46}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $F : J \times C(H; E) \to \mathcal{P}(E)$ is a multivalued map with nonempty compact values, $\rho : J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$. An extension of this problem is given in section 6.5, we consider the following fractional neutral evolution inclusion with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t)[y(t) - g(t, y_{\rho(t, y_t)})] \in A(t)y(t) + F(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \quad t \in J$$
(47)

$$y_0 = \phi \in \mathcal{B},\tag{48}$$

where $A(\cdot)$, F and φ are as in problem (6.5) – (6.6) and $g: J \times \mathcal{B} \to E$ is a given function.

Finally, we give the conclusion of our results and the bibliography based on this work.

Chapter 1

Preliminaries

The aim of this Chapter is to introduce some basic concepts, notations and elementary results that used throughout this thesis.

1.1 Notations and Definitions

Let J := [0, b] and H := [-r, 0] be two closed and bounded intervals in \mathbb{R} for the real numbers b, r > 0.

Let E be the Banach space of real numbers with the norm $\|.\|$

Consider C(H; E) be the Banach space of continuous functions from H into E with the norm

$$||y|| = \sup \{ |y(t)| : t \in H \}.$$

Let B(E) be the space of all bounded linear operators from E into E, with the norm

$$||N||_{B(E)} = \sup \{ |N(y)| : |y| = 1 \}.$$

A measurable function $y : J \to E$ is Bochner integrable if and only if |y| is Lebesgue integrable. (For the Bochner integral properties, see the classical monograph of Yosida [101]).

Let $L^1(J, E)$ denotes the Banach space of measurable functions $y: J \to E$ which are Bochner integrable normed by

$$\|y\|_{L^1} = \int_0^{+\infty} |y(t)| \, dt.$$

For any continuous function y defined on [-r, b], and for all $t \in J$, note by y_t element of $\mathcal{C}(H; E)$ defined by

$$y_t(\theta) = y(t+\theta) \quad \text{for } \theta \in H.$$

Here $y_t(\cdot)$ represent the history of the state from t - r up to the present time t.

Definition 1.1.1. A function $f: J \times E \to E$ is said to be an L^1 -Carathéodory function if it satisfies :

- (i) for each $t \in J$ the function $f(t, .) : E \to E$ is continuous;
- (ii) for each $y \in E$ the function $f(., y) : J \to E$ is measurable ;

(iii) for every positive integer k there exists $h_k \in L^1(J; \mathbb{R}^+)$ such that

 $|f(t,y)| \le h_k(t)$ for all $|y| \le k$ and almost every $t \in J$.

1.2 Some Properties in Fréchet Spaces

Let X be a Fréchet space with a family of semi-norms $\{\|\cdot\|_n\}_{n\in\mathbb{N}}$. We assume that the family of semi-norms $\{\|\cdot\|_n\}$ verifies :

 $||x||_1 \le ||x||_2 \le ||x||_3 \le \dots$ for every $x \in X$.

Let $Y \subset X$, we say that Y is bounded if for every $n \in \mathbb{N}$, there exists $\overline{M}_n > 0$ such that

 $\|y\|_n \le \overline{M}_n \quad \text{for all } y \in Y.$

To X we associate a sequence of Banach spaces $\{(X^n, \|\cdot\|_n)\}$ as follows: For every $n \in \mathbb{N}$, we consider the equivalence relation \sim_n defined by : $x \sim_n y$ if and only if $\|x - y\|_n = 0$ for $x, y \in X$. We denote $X^n = (X|_{\sim_n}, \|\cdot\|_n)$ the quotient space, the completion of X^n with respect to $\|\cdot\|_n$. To every $Y \subset X$, we associate a sequence $\{Y^n\}$ of subsets $Y^n \subset X^n$ as follows : For every $x \in X$, we denote $[x]_n$ the equivalence class of x of subset X^n and we defined $Y^n = \{[x]_n : x \in Y\}$. We denote $\overline{Y^n}$, $int_n(Y^n)$ and $\partial_n Y^n$. Respectively, the closure, the interior and the boundary of Y^n with respect to $\|\cdot\|_n$ in X^n .

The appropriate concept of contraction in X is given in the following definition.

Definition 1.2.1. [52] A function $f : X \to X$ is said to be a contraction if for each $n \in \mathbb{N}$ there exists $k_n \in (0, 1)$ such that

$$||f(x) - f(y)||_n \le k_n ||x - y||_n$$
 for all $x, y \in X$.

1.3 Evolution System Generator

In what follows, we assume that $\{A(t)\}_{t\in J}$ is a family of closed densely defined linear unbounded operators on the Banach space E and with domain D(A(t)) independent of t. Additionally, we introduce the following hypothesis:

- (P1) For $t \in [0, b]$, the domain D(A(t)) = D is independent of t and is dense on X.
- (P2) For $t \ge 0$, the resolvent $R(\lambda, A(t)) = (\lambda I A(t))^{-1}$ exists for all λ with $Re(\lambda) \le 0$, and there is a constant M independent of λ and t such that

$$||R(t, A(t))|| \le M(1 + |\lambda|)^{-1}$$
 for $Re(\lambda) \le 0$.

(P3) There exist constant L > 0 and $0 < \alpha \leq 1$ such that

$$\|(A(t) - A(\theta))A^{-1}(\tau)\| \le L|t - \tau|^{\alpha} \quad t, \theta, \tau \in J.$$

(P4) The resolvent R(t, A(t)) is compact for $t \ge 0$.

Lemma 1.3.1. ([7], p. 159) Under the assumption (P1) - (P4), the Cauchy problem

$$\begin{cases} y'(t) - A(t)y(t) = 0 & t \in J \\ y(0) = y_0 \end{cases}$$
(1.1)

has a unique evolution system $\{U(t,s)\}_{0 \le s \le t \le b}$ on E satisfying the following properties:

- 1. U(t,t) = I where I is the identity operator in E,
- 2. $U(t,s) U(s,\tau) = U(t,\tau) \text{ for } 0 \le \tau \le s \le t \le b$,
- 3. $U(t,s) \in B(E)$ the space of bounded linear operators on E, where for every $(t,s) \in \Delta$ and for each $y \in E$, the mapping $(t,s) \to U(t,s)$ y is continuous.
- 4. For $0 \le s \le t \le b$, $U(t,s) : X \longrightarrow D$ and $t \longrightarrow U(t,s)$ is strongly differentiable on E. The derivative $\frac{\partial}{\partial t}U(t,s) \in B(E)$ and it is strongly continuous on $0 \le s \le t \le b$. Moreover,

$$\frac{\partial}{\partial t}U(t,s) = A(t)U(t,s) \quad \text{for } 0 \le s \le t \le b,$$
$$\|\frac{\partial}{\partial t}U(t,s)\|_{B(E)} = \|A(t)U(t,s)\|_{B(E)} \le \frac{C}{t-s},$$
$$\|A(t)U(t,s)A^{-1}(s)\|_{B(E)} \le C \quad \text{for } 0 \le s \le t \le b.$$

5. For every $v \in D$ and $t \in (0, b], U(t, s)v$ is differentiable with respect to s on $0 \le s \le t \le b$

$$\frac{\partial}{\partial s}U(t,s)v = -U(t,s)A(s)v.$$

6. U(t,s) is a compact operator for $0 \le s \le t \le b$.

And, for each $y_0 \in E$, the Cauchy problem (1.1) has a unique classical solution $y \in C^1(J, E)$ given by

$$y(t) = U(t,0)y_0, \quad t \in J$$

More details on evolution systems and their properties could be found on the books of Ahmed [7], Engel and Nagel [47] and Pazy [90].

1.4 Definition of Mild Solution

Definition 1.4.1. We say that the function $y(\cdot) : H \cup J \to E$ is a mild solution for the following first order of partial functional differential evolution equations with delay

$$\begin{cases} y'(t) = A(t)y(t) + f(t, y_t), & a.e. \ t \in J \\ y(t) = \varphi(t), \quad t \in H \end{cases}$$
(1.2)

If $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation :

$$y(t) = U(t,0) \ \varphi(0) + \int_0^t U(t,s)f(s,y_s) \ ds \quad for \ each \ t \in J.$$
 (1.3)

Definition 1.4.2. We say that the function $y(\cdot) : H \cup J \to E$ is a mild solution for the following Caputo's fractional derivative order of partial functional differential evolution equations with delay

$$\begin{cases} {}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{t}), \quad 0 < \alpha < 1 \ a.e. \ t \in J \\ y(t) = \varphi(t), \quad t \in H \end{cases}$$
(1.4)

If $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation :

$$y(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s,y_s) \ ds \quad \text{for each} \ t \in J.$$
(1.5)

1.5 Controllability of Mild Solution

Controllability of differential equations is the origin of the study of differential inclusions. Let us consider the control problem of the following evolution equations with delay

$$\begin{cases} y'(t) = A(t)y(t) + Cu(t) + f(t, y_t), & \text{a.e. } t \in J \\ y(t) = \varphi(t), & t \in H, \end{cases}$$
(1.6)

where $f: J \times C(H; E) \to E$, $\rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{0 \le t < b}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $0 \le s \le t$. The control function u(.) is given in $L^2(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|$, C is a bounded linear operator from E into E.

Definition 1.5.1. The mild solution y of the problem evolution (1.6) is said to be controllable on the interval $H \cup J$ if for every initial function $\varphi \in H$ and $y_1 \in E$, there exists a control $u \in L^2(J; E)$ such that the mild solution $y(\cdot)$ of the problem (1.6) satisfies $y(b) = y_1$.

1.6 Phase Space \mathcal{B}

The notation of the phase space \mathcal{B} plays an important role in the study of both qualitative and quantitative theory for functional differential equations. A usual choice is a semi-normed space satisfying suitable axioms, which was introduced by Hale and Kato (see [60]). For further applications see for instance the books of Hale *et al.* [62], Hino and his collaborators [74] and Lakshmikantham [83] and their references.

For any $t \in J$ denote $B_{+\infty} = \{y :] - \infty, b] \to E : y|_J \in \mathcal{C}(J; E), y_0 \in \mathcal{B}\}$ or $y|_J$ is the restriction of y on J. Consider that the space $(\mathcal{B}, ||(.,.)||_{\mathcal{B}})$ is a seminormed linear space of functions that mapping $(-\infty, 0] \times (-\infty, 0]$ into E, and satisfying the following fundamental axioms which were adapted from those introduced by Hale and Kato for ordinary differential functional equations:

- (A₁) If $y: (-\infty, b) \to E$, b > 0, is continuous on [0, b] and $y_0 \in \mathcal{B}$, then for every $t \in [0, b)$ the following conditions hold :
 - (i) $y_t \in \mathcal{B}$;
 - (ii) There exists a positive constant H such that $|y(t)| \leq H ||y_t||_{\mathcal{B}}$;

(iii) There exist two functions $K(\cdot), M(\cdot) : \mathbb{R}_+ \to \mathbb{R}_+$ independent of y with K continuous and M locally bounded such that

$$||y_t||_{\mathcal{B}} \le K(t) \sup\{ |y(s)| : 0 \le s \le t\} + M(t) ||y_0||_{\mathcal{B}}.$$

- (A_2) For the function y in (A_1) , y_t is a \mathcal{B} -valued continuous function on [0, b].
- (A_3) The space \mathcal{B} is complete.

Denote $K_b = \sup\{K(t) : t \in [0, b]\}$ and $M_b = \sup\{M(t) : t \in [0, b]\}.$

Remark 1.6.1. 1. (ii) is equivalent to $|\phi(0)| \leq H \|\phi\|_{\mathcal{B}}$ for every $\phi \in \mathcal{B}$.

- 2. Since $\|\cdot\|_{\mathcal{B}}$ is a seminorm, two elements $\phi, \psi \in \mathcal{B}$ can verify $\|\phi \psi\|_{\mathcal{B}} = 0$ without necessarily $\phi(\theta) = \psi(\theta)$ for all $\theta \leq 0$.
- 3. From the previous equivalence in the first remark, we can see that for all $\phi, \psi \in \mathcal{B}$ such that $\|\phi \psi\|_{\mathcal{B}} = 0$: We necessarily have that $\phi(0) = \psi(0)$.

Now we indicate some examples of phase spaces. For other details we refer, for instance to the book by Hino *et al* [74].

Example 1.6.1. The spaces BC, BUC, C_1 and C_0 . Let:

BC the space of bounded continuous functions defined from $(-\infty, 0]$ to E;

BUC the space of bounded uniformly continuous functions defined from $(-\infty, 0]$ to E;

$$\begin{split} C^{\infty} &:= \left\{ \phi \in BC : \lim_{\theta \to -\infty} \phi(\theta) \text{ exist in } E \right\}; \\ C^{0} &:= \left\{ \phi \in BC : \lim_{\theta \to -\infty} \phi(\theta) = 0 \right\}, \text{ endowed with the uniform norm} \\ \|\phi\| &= \sup\{|\phi(\theta)| : \theta \leq 0\}. \end{split}$$

We have that the spaces BUC, C^{∞} and C^{0} satisfy conditions $(A_{1}) - (A_{3})$. However, BC satisfies $(A_{1}), (A_{3})$ but (A_{2}) is not satisfied.

Remark 1.6.2. $C^0 \subset C^\infty$.

Example 1.6.2. The spaces C_g , UC_g , C_g^{∞} and C_g^0 . Let g be a positive continuous function on $(-\infty, 0]$. We define

$$\begin{split} C_g &:= \left\{ \phi \in C((-\infty,0],E) : \frac{\phi(\theta)}{g(\theta)} \text{ is bounded on } (-\infty,0] \right\};\\ C_g^0 &:= \left\{ \phi \in C_g : \lim_{\theta \to -\infty} \frac{\phi(\theta)}{g(\theta)} = 0 \right\}, \text{ endowed with the uniform norm}\\ \|\phi\| &= \sup \left\{ \frac{|\phi(\theta)|}{g(\theta)} : \theta \le 0 \right\}. \end{split}$$

Then we have that the spaces C_g and C_g^0 satisfy conditions (A_3) . We consider the following condition on the function g.

(g₁) For all
$$a > 0$$
, $\sup_{0 \le t \le a} \sup \left\{ \frac{g(t+\theta)}{g(\theta)} : -\infty < \theta \le -t \right\} < \infty$.

They satisfy conditions (A_1) and (A_2) if (g_1) holds.

Remark 1.6.3. For $g \equiv 1$, we have $C_1 = BC$ and $C_1^0 = C_0$

Example 1.6.3. The space C_{γ} .

For any real constant γ , we define the functional space C_{γ} by

$$C_{\gamma} := \left\{ \phi \in C((-\infty, 0], E) : \lim_{\theta \to -\infty} e^{\gamma \theta} \phi(\theta) \text{ exists in } E \right\}$$

endowed with the following norm

$$\|\phi\| = \sup\{e^{\gamma\theta}|\phi(\theta)|: \ \theta \le 0\}.$$

Then in the space C_{γ} the axioms $(A_1) - (A_3)$ are satisfied.

1.7 State-Dependent Delay

1. Finite delay : We always assume that $\rho : J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(\mathcal{H}; E), \ \rho(s,\varphi) \le 0\}.$$

 (H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

 $\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$ for every $t \in \mathcal{R}(\rho^-)$.

Remark 1.7.1. The condition (H_{φ}) , is frequently verified by functions continuous and bounded. For more details, see for instance [1, 70, 74].

Lemma 1.7.1. ([70], Lemma 2.4) If $y : [-r, b] \to E$ is a function such that $y_0 = \varphi$, then

$$\|y_s\| \le \mathcal{L}^{\varphi} \|\varphi\| + \sup\{|y(\theta)|; \theta \in [0, \max\{0, s\}]\}, \ s \in \mathcal{R}(\rho^-) \cup J,$$

where $\mathcal{L}^{\varphi} = \sup_{t \in \mathcal{R}(\rho^{-})} \mathcal{L}^{\varphi}(t).$

2. Infinite delay : We always assume that $\rho : J \times B \longrightarrow \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}.$$

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

Remark 1.7.2. The condition (H_{ϕ}) , is frequently verified by functions continuous and bounded. For more details, see for instance [1, 70, 74].

Lemma 1.7.2. ([70], Lemma 2.4) If $y: (-\infty, b] \to E$ is a function such that $y_0 = \phi$, then

$$\|y_s\|_{\mathcal{B}} \le (M_b + \mathcal{L}^{\phi}) \|\phi\|_{\mathcal{B}} + K_b \sup\{|y(\theta)|; \theta \in [0, max\{0, s\}]\}, \ s \in \mathcal{R}(\rho^-) \cup J,$$

where $\mathcal{L}^{\phi} = \sup_{t \in \mathcal{R}(\rho^{-})} \mathcal{L}^{\phi}(t)$. K_b and M_b are as defined in the previous assumption (A1) in the condition (iii).

1.8 Some Properties of Fractional Calculs

In this section, we introduce the notations, definitions and preliminary lemmas concerning to partial fractional calculs theory.

Lemma 1.8.1. Let $v: J \to [0, \infty)$ be a real function and $\omega(.)$ be a nonnegative, locally integrable function on J. If there are constants c > 0 and $0 < \alpha < 1$ such that

$$\upsilon(t) \le \omega(t) + c \int_0^t \frac{\upsilon(s)}{(t-s)^{\alpha}} ds, \qquad (1.7)$$

then there exists a constant $\delta = \delta(\alpha)$ such that

$$v(t) \le \omega(t) + \delta c \int_0^t \frac{\omega(s)}{(t-s)^{\alpha}} ds, \qquad (1.8)$$

for every $t \in J$.

Lemma 1.8.2. The system (1.4) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s,y_s) \, ds \quad t \in J.$$
(1.9)

In other words, every solution of the integral equation (1.9) is also solution of the system (1.4) and vice versa.

Definition 1.8.1. The Riemann-Liouville fractional integral operator of order $\alpha > 0$ of a function $f : \mathbb{R}^+ \longrightarrow \mathbb{R}$ is defined as

$$I_0^{\alpha} f(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s) ds$$
 (1.10)

Provide the right hand side exists pointwise on \mathbb{R}^+ . $\Gamma(.)$ is the Euler gamma function. For instance, $I^{\alpha}f$ exists for all $\alpha > 0$. When $h \in C^0(\mathbb{R}^+) \cap L^1_{loc}(\mathbb{R}^+)$. Note also that when $h \in C^0(\mathbb{R}^+_0)$ then $I^{\alpha}f \in C^0(\mathbb{R}^+_0)$ and moreover $I^{\alpha}f(0) = 0$.

Definition 1.8.2. The fractional derivative of order $\alpha > 0$ of a function $f : \mathbb{R}^+ \longrightarrow \mathbb{R}$ in the Caputo sense is given by

$$\frac{d^{\alpha}f(t)}{dt^{\alpha}} = \frac{1}{\Gamma(m-\alpha)}\frac{d}{dt}\int_0^t (t-s)^{m-\alpha-1}f(s)ds = \frac{d}{dt}I_0^{1-\alpha}f(t).$$

1.9 Definition of Multivalued Functions

Let (X, d) be a metric space. We use the following notations :

$$\mathcal{P}_{cl}(X) = \{ Y \in \mathcal{P}(X) : Y \text{ closed} \}, \quad \mathcal{P}_{b}(X) = \{ Y \in \mathcal{P}(X) : Y \text{ bounded} \},$$

$$\mathcal{P}_{cv}(X) = \{Y \in \mathcal{P}(X) : Y \text{ convexe}\}, \quad \mathcal{P}_{cp}(X) = \{Y \in \mathcal{P}(X) : Y \text{ compact}\}$$

Consider $H_d: \mathcal{P}(X) \times \mathcal{P}(X) \longrightarrow \mathbb{R}_+ \cup \{\infty\}$, given by

$$H_d(\mathcal{A}, \mathcal{B}) = \max \left\{ \sup_{a \in \mathcal{A}} d(a, \mathcal{B}) , \sup_{b \in \mathcal{B}} d(\mathcal{A}, b) \right\},\$$

where $d(\mathcal{A}, b) = \inf_{a \in \mathcal{A}} d(a, b), d(a, \mathcal{B}) = \inf_{b \in \mathcal{B}} d(a, b)$. Then $(\mathcal{P}_{b,cl}(X), H_d)$ is a metric space and $(\mathcal{P}_{cl}(X), H_d)$ is a generalized (complete) metric space (see [80]).

Definition 1.9.1. A multivalued map $G : J \to \mathcal{P}_{cl}(X)$ is said to be measurable if for each $x \in E$, the function $Y : J \to X$ defined by

$$Y(t) = d(x, G(t)) = \inf\{|x - z| : z \in G(t)\}\$$

is measurable where d is the metric induced by the normed Banach space X.

Definition 1.9.2. A function $F : J \times C(H; E) \longrightarrow \mathcal{P}(X)$ is said to be an L^1_{loc} -Carathéodory multivalued map if it satisfies :

- (i) $x \mapsto F(t, y)$ is continuous (with respect to the metric H_d) for almost all $t \in J$;
- (ii) $t \mapsto F(t, y)$ is measurable for each $y \in C(H; E)$;

(iii) for every positive constant k there exists $h_k \in L^1_{loc}(J; \mathbb{R}^+)$ such that

 $||F(t,y)|| \le h_k(t)$ for all $||y|| \le k$ and for almost all $t \in J$.

Let $(X, \|\cdot\|)$ be a Banach space. A multivalued map $G: X \to \mathcal{P}(X)$ has convex (closed) values if G(x) is convex (closed) for all $x \in X$. We say that G is bounded on bounded sets if G(B) is bounded in X for each bounded set B of X, i.e.,

$$\sup_{x \in B} \left\{ \sup\{ \|y\| : y \in G(x) \} \right\} < \infty.$$

Finally, we say that G has a *fixed point* if there exists $x \in X$ such that $x \in G(x)$.

For each $y \in B_{+\infty}$ let the set $S_{F,y}$ known as the set of selectors from F defined by

$$S_{F,y} = \{ v \in L^1(J; E) : v(t) \in F(t, y_t) , a.e. t \in J \}.$$

For more details on multivalued maps we refer to the books of Deimling [45], Górniewicz [56], Hu and Papageorgiou [75] and Tolstonogov [96].

Definition 1.9.3. A multivalued map $F : X \to \mathcal{P}(X)$ is called an admissible contraction with constant $\{k_n\}_{n \in \mathbb{N}}$ if for each $n \in \mathbb{N}$ there exists $k_n \in (0, 1)$ such that

- *i*) $H_d(F(x), F(y)) \le k_n ||x y||_n$ for all $x, y \in X$.
- ii) for every $x \in X$ and every $\epsilon \in (0, \infty)^n$, there exists $y \in F(x)$ such that

$$||x-y||_n \leq ||x-F(x)||_n + \epsilon_n$$
 for every $n \in \mathbb{N}$

1.10 Fixed Point Theorems

In this section we give some fixed point theorems that will be used in the sequel. All the abstract results established in this thesis are based on these theorems.

1. Nonlinear Alternative in Fréchet space:

Theorem 1.10.1. (Nonlinear Alternative of Frigon-Granas, [52]). Let X be a Fréchet space and $Y \subset X$ be a closed subset and let $N : Y \to X$ be a contraction such that N(Y) is bounded. Then one of the following statements holds :

- (C1) N has a unique fixed point;
- (C2) There exists $\lambda \in [0, 1)$, $n \in \mathbb{N}$ and $x \in \partial_n Y^n$ such that $||x \lambda| N(x)||_n = 0$.

Theorem 1.10.2. (Nonlinear Alternative of Avramescu [18]). Let X be a Fréchet space and $A, B: X \longrightarrow X$ two operators satisfying:

- (1) A is a compact operator.
- (2) B is a contraction. Then either
- (C1) the operator A + B has a fixed points. Or

(C2) the set $\{x \in X, x = \lambda A(x) + \lambda B(\frac{x}{\lambda})\}$ is unbounded for $\lambda \in]0, 1[$.

Theorem 1.10.3. (Nonlinear Alternative of Frigon, [50, 51]). Let X be a Fréchet space and U an open neighborhood of the origin in X and let $N : \overline{U} \to \mathcal{P}(X)$ be an admissible multivalued contraction. Assume that N is bounded. Then one of the following statements holds :

(C1) N has a fixed point;

(C2) There exists $\lambda \in [0, 1)$ and $x \in \partial U$ such that $x \in \lambda N(x)$.

2. Nonlinear Alternative in Banach space:

Theorem 1.10.4. (Nonlinear Alternative of Frigon-Granas, [52]). Let X be a Banach space and $Y \subset X$ be a closed subset and let $N : Y \to X$ be a contraction such that N(Y) is bounded. Then one of the following statements holds :

- (C1) N has a unique fixed point;
- (C2) There exists $\lambda \in [0, 1)$ and $x \in \partial Y$ such that $||x \lambda N(x)|| = 0$.

Theorem 1.10.5. (Nonlinear Alternative of Avramescu [18]). Let X be a Banach space and $A, B: X \longrightarrow X$ two operators satisfying:

- (1) A is a compact operator.
- (2) B is a contraction. Then either
- (C1) the operator A + B has a fixed points. Or
- (C2) the set $\{x \in X, x = \lambda A(x) + \lambda B(\frac{x}{\lambda})\}$ is unbounded for $\lambda \in]0, 1[$.

Theorem 1.10.6. (Nonlinear Alternative of Frigon, [50, 51]). Let X be a Banach space and U an open neighborhood of the origin in X and let $N : \overline{U} \to \mathcal{P}(X)$ be an admissible multivalued contraction. Assume that N is bounded. Then one of the following statements holds :

- (C1) N has a fixed point;
- (C2) There exists $\lambda \in [0,1)$ and $x \in \partial U$ such that $x \in \lambda N(x)$.

Chapter 2

Fractional Evolution Equations with State-Dependent Delay

2.1 Introduction

In this chapter, we shall establish sufficient conditions for the existence of the unique mild solution for some classes of fractional for partial functional and neutral functional differential evolution equations with finite and infinite state-dependent delay involving the Caputo's fractional derivative order¹.

Using the alternative of Leary-Schauder type for contraction maps given by Frigon and Granas in Banach space (see [52]), combined with the semi-group theory.

To our Knowledge, there are very few papers devoted to fractional differential equations with delay. By using a fractional version of Gronwall's inequality, we demonstrate the existence of the unique mild solution in section 2.2 for the following class of fractional evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \text{ a.e. } t \in J = [0, b]$$

$$(2.1)$$

$$y(t) = \varphi(t), \quad t \in H = [-r, 0],$$
 (2.2)

where $b > 0, 0 < r < +\infty, {}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f: J \times C(H; E) \to E, \ \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of bounded linear operators $\{U(t, s)\}_{(t,s)\in J \times J}$ for $s \leq t$.

An extension of this problem is given in section 2.3 for the following class of fractional neutral evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J$$
(2.3)

$$y(t) = \varphi(t), \quad t \in H, \tag{2.4}$$

where $A(\cdot)$, f and φ are as in the above problem (2.1) – (2.2) and $g: J \times C(H; E) \to E$ is a given function.

¹[12] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Fractional partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.

In section 2.4, we study the fractional evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \text{ a.e. } t \in J$$
(2.5)

$$y_0 = \phi \in \mathcal{B},\tag{2.6}$$

where \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f: J \times \mathcal{B} \to E, \rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t, s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 2.5 for the following class of fractional neutral evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}), \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J$$
(2.7)

$$y_0 = \phi \in \mathcal{B},\tag{2.8}$$

where $A(\cdot)$, f and ϕ are as in problem (2.5) – (2.6) and $g: J \times \mathcal{B} \to E$ is a given function. Finally, section 2.6 is devoted to examples illustrating the abstract theory considered in provious sections.

2.2 Partial Problem with Finite Delay

Before stating and proving the main result, we give first the definition of mild solution of the fractional problem (2.1) - (2.2).

Lemma 2.2.1. The system (2.1) - (2.2) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) \, ds.$$
(2.9)

In other words, every solution of the integral equation (2.9) is also mild solution of the system (2.1) - (2.2) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (2.1) - (2.2), and using some classical results from fractional calculus to get (2.9).

Definition 2.2.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (2.1) - (2.2)if $y(t) = \varphi(t)$ for all $t \in [-r, 0]$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds \quad \text{for each } t \in J.$$
(2.10)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(H;E), \rho(s,\varphi) \le 0\}.$$

We always assume that $\rho: J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter :

(H1) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M} \quad \text{for every } (t,s) \in \Delta := \{(t,s) \in J \times J : 0 \le s \le t \le b\}.$$

(H2) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in C(H; E)$.

(H3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in C(H; E)$ with $||u|| \le R$ and $||v|| \le R$.

Theorem 2.2.1. Assume that the hypotheses (H_{φ}) and (H1) - (H3) hold and moreover

$$\frac{\widehat{M}l_n^*b^\alpha}{\Gamma(1+\alpha)} < 1 \tag{2.11}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_n^* = \sup l_n(s)$. Then the fractional evolution problem (2.1) - (2.2) has a unique mild solution on [-r, b].

Proof. Transform the problem (2.1) - (2.2) into a fixed-point problem. Consider the operator $N: C([-r,b]; E) \to C([-r,b]; E)$ defined by :

$$N(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(2.12)

Clearly, fixed points of the operator N are mild solutions of the problem (2.1) - (2.2).

Let y be a possible solution of the problem (2.1) - (2.2). Given $t \leq b$, then from (H1), (H2), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} |f(s,y_{\rho(s,y_s)})| \ ds \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|y_{\rho(s,y_s)}\| \right] ds \end{aligned}$$

It follows that

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \le \widehat{M} \|\varphi\| + \mathcal{L}^{\varphi} \|\varphi\| + \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[|y(s) + \mathcal{L}^{\varphi} \|\varphi\| \right] ds$$

 Set

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha}}{\Gamma(\alpha + 1)}$$

We consider the function μ defined by

$$\mu(t):=\sup\ \{\ |y(s)|+\mathcal{L}^{\varphi}\|\varphi\|\ :\ 0\leq s\leq t\ \},\quad 0\leq t\leq b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds \quad \text{for } t \in [0,b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}q^*b^{\alpha}}{\Gamma(\alpha+1)}\right] := \Lambda_b$$

Since $||y|| \leq \mu(t)$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Set

$$Y = \{ y \in C([-r,b];E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1\}.$$

Clearly, Y is a closed subset of C([-r, b]; E).

We shall show that $N: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H1) and (H3) for each $t \in [0, b]$

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left\| f(s,y_{\rho(s,y_s)}) - f(s,\overline{y}_{\rho(s,y_s)}) \right\| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \, l_n(s) \, \|y_{\rho(s,y_s)} - \overline{y}_{\rho(s,y_s)}\| \, ds. \end{aligned}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{\widehat{M}l_n^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &\leq \frac{\widehat{M}l_n^* b^\alpha}{\Gamma(\alpha+1)} \|y - \overline{y}\| \end{aligned}$$

Therefore,

$$\|N(y) - N(\overline{y})\| \le \frac{\widehat{M}l_n^*b^\alpha}{\Gamma(\alpha+1)} \|y - \overline{y}\|.$$

So by (6.16), the operator N is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y)$, $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator N has a unique fixed-point y^* which is the unique mild solution of the problem (2.1) - (2.2).

2.3 Neutral Problem with Finite Delay

We give here an extension to previous results for the neutral case (2.3) - (2.4). Firstly, we define its mild solution.

Lemma 2.3.1. The system (2.3) - (2.4) is equivalent to the nonlinear integral equation

$$y(t) = [\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \qquad (2.13)$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) ds \quad t \in J.$$

In other words, every solution of the integral equation (2.13) is also solution of the system (2.3) - (2.4) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (2.3) - (2.4), and using some classical results from fractional calculus to get (2.13).

Definition 2.3.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (2.3) - (2.4)if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s,y_{\rho(s,y_s)}) \, ds.$$
(2.14)

We consider the hypotheses (H_{φ}) , (H1) - (H3) and we need to introduce the following assumptions :

(H4) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H5) There exists a constant $0 < L < \frac{1}{\overline{M}_0}$ such that

$$|A(t) g(t, \varphi)| \le L (||\varphi|| + 1)$$
 for all $t \in J$ and $\varphi \in C(H; E)$.

(H6) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} (|s-\overline{s}| + ||\varphi - \overline{\varphi}||)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

Theorem 2.3.1. Suppose that the hypotheses (H1) - (H5) are satisfied and moreover

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M}L_n^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1$$
(2.15)

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_n^* = \sup l_n(s)$. Then the problem (2.3) - (2.4) has a unique mild solution on [-r, b].

Proof. Transform as below the neutral problem (2.3) - (2.4) into a fixed point problem by considering the operator $\widetilde{N}: C([-r,b];E) \to C([-r,b];E)$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ [\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(2.16)

Clearly, the fixed points of the operator \widetilde{N} are mild solutions of the problem (2.3) – (2.4).

Let y be a possible solution of the problem (2.3) - (2.4). Given $t \leq b$. Then, using (H1) - (H2) and (H4) - (H5), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq |U(t,0)[\varphi(0) - g(0,\varphi)]| + |g(t,y_{\rho(t,y_t)})| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s,y_{\rho(s,y_s)}) ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t,y_{\rho(t,y_t)})| + \widehat{M}||\varphi|| + ||U(t,0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0,\varphi)| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||y_{\rho(s,y_s)}|| \right] ds \\ &\leq \widehat{M}||\varphi|| + \overline{M}_0 L \left(||y_{\rho(t,y_t)}|| + 1 \right) + \widehat{M}\overline{M}_0 L \left(||\varphi|| + 1 \right) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||y_{\rho(s,y_s)}|| \right] ds \end{aligned}$$

Since $||y_{\rho(t,y_t)}|| \le |y(t)| + \mathcal{L}^{\varphi}||\varphi||$ we obtain

$$\begin{aligned} |y(t)| &\leq \overline{M}_0 L \left(|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \right) + \widehat{M} \|\varphi\| (1 + \overline{M}_0 L) + \widehat{M} \overline{M}_0 L + \frac{\widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\| \right) ds \end{aligned}$$

Then

$$\begin{aligned} (1 - \overline{M}_0 L)|y(t)| &\leq \left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right] \|\varphi\| + \widehat{M}\overline{M}_0 L + \frac{Mp^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi}\|\varphi\|\right) ds \end{aligned}$$

 Set

$$\beta_b := \frac{\left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right]}{(1 - \overline{M}_0 L)} \|\varphi\| + \frac{\widehat{M} \overline{M}_0 L}{(1 - \overline{M}_0 L)} + \frac{\widehat{M} p^* b^{\alpha}}{(1 - \overline{M}_0 L)\Gamma(\alpha + 1)} + \mathcal{L}^{\varphi} \|\varphi\|.$$

Thus

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \leq \beta_b + \frac{\widehat{M}q^*}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|) ds.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \beta_b + \frac{\widehat{M}q^*}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\beta_b = \beta_b(\alpha)$ such that

$$\mu(t) \leq \beta_b \times \left[1 + \frac{\widehat{M}q^*b^{\alpha}}{(1 - \overline{M}_0 L)\Gamma(\alpha + 1)} \right] := \Lambda_b$$

Since $||y|| \le \mu(t)$, we have $||y|| \le \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Now, we shall show that $\widetilde{N}: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H1) and (H3) - (H4) for each $t \in [0, b]$

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq |g(t, y_{\rho(t, y_t)}) - g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \left| f(s, y_{\rho(s, y_s)}) - f(s, \overline{y}_{\rho(s, y_s)}) \right| \, ds \\ &\leq \|A^{-1}(t)\| |A(t)g(t, y_{\rho(t, y_t)}) - A(t)g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \, l_n(s) \, \|y_{\rho(s, y_s)} - \overline{y}_{\rho(s, y_s)}\| \, ds. \end{split}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq \overline{M}_0 L_\star |y(t) - \overline{y}(t)| + \frac{\widehat{M}L_n^* b^\alpha}{\Gamma(\alpha + 1)} \|y - \overline{y}\| \\ &\leq \left[\overline{M}_0 L_\star + \frac{\widehat{M}L_n^* b^\alpha}{\Gamma(\alpha + 1)} \right] \|y - \overline{y}\| \end{split}$$

Therefore,

$$\|\widetilde{N}(y) - \widetilde{N}(\overline{y})\| \le \left[\overline{M}_0 L_{\star} + \frac{\widehat{M}L_n^* b^{\alpha}}{\Gamma(\alpha + 1)}\right] \|y - \overline{y}\|$$

So, for an appropriate choice of $\overline{M}_0 L_{\star}, L_n^*$ and b^{α} such that

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M}L_n^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1,$$

the operator \widetilde{N} is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda \widetilde{N}(y)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. A consequence of the nonlinear alternative of Frigon and Granas shows that (C1) holds. We deduce that the operator \widetilde{N} has a unique fixed-point y^* which is the unique mild solution of the problem (2.3) - (2.4).

2.4 Partial Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of mild solution of the fractional problem (2.5) - (2.6).

Lemma 2.4.1. The system (2.5) - (2.6) is equivalent to the nonlinear integral equation

$$y(t) = \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) \, ds.$$
(2.17)

In other words, every solution of the integral equation (2.17) is also solution of the system (2.5) - (2.6) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (2.5) - (2.6), and using some classical results from fractional calculus to get (2.17).

Definition 2.4.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (2.5)-(2.6) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds \quad \text{for each } t \in J.$$
(2.18)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}.$$

We always assume that $\rho: J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter :

(H01) There exists a constant $\widehat{M} \geq 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H02) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H03) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in \mathcal{B}$ with $||u|| \leq R$ and $||v|| \leq R$.
We define in $C((-\infty, b]); E)$ the semi-norms by :

$$||y|| := \sup \{ |y(t)| : t \in [0, b] \}$$

Consider the following space

$$\Omega = \left\{ y : (-\infty, b] \to E : y|_{(-\infty, 0]} \in B \text{ and } y|_{[0, b]} \text{ is continuous } \right\},$$

Theorem 2.4.1. Assume that the hypotheses (H_{φ}) and (H01) - (H03) hold and moreover

$$\frac{K_b \widehat{M} l_n^* b^{\alpha}}{\Gamma(\alpha+1)} < 1 \tag{2.19}$$

where $l_n^* = \sup l_n(s)$. Then the problem (2.5) – (2.6) has a unique mild solution on $(-\infty, b]$.

Proof. We transform the problem (2.5) - (2.6) into a fixed-point problem. Consider the operator $N : \Omega \to \Omega$ defined by :

$$N(y)(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(2.20)

Clearly, fixed points of the operator N are mild solutions of the problem (2.5) - (2.6).

For $\phi \in \mathcal{B}$, we will define the function $x(.) : (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (2.18) if and only if z satisfies $z_0 = 0$ and

$$z(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \ ds \quad \text{for } t \in J.$$

Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operator $F:\Omega^0\to\Omega^0$ by :

$$F(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds \quad \text{for } t \in J.$$
(2.21)

Obviously the operator N has a fixed point is equivalent to F has one, so it turns to prove that F has a fixed point.

Let $z \in \Omega^0$ be be a possible fixed point of the operator. By the hypotheses (H01) and (H02), we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|f(s,z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})\| ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)} \|_{\mathcal{B}} \right] ds \end{aligned}$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, b]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_b|x(s)| + (M_b + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\|U(s,0)\|_{B(E)}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\widehat{M}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_b|z(s)| + K_b\widehat{M}H\|\phi\|_{\mathcal{B}} + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi} + K_b\widehat{M}H)\|\phi\|_{\mathcal{B}} \end{aligned}$$

Set $c_b := (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}}$. It follows that

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left(K_b |z(s)| + (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}} \right) ds$$

Then, we have

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds$$

where $p^* = \sup p(s), q^* = \sup q(s)$ Then

$$K_b|z(t)| + c_b \leq \frac{K_b\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + c_b + \frac{K_b\widehat{M}q^*}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds.$$

 Set

$$\delta_b := \frac{K_b \widehat{M} p^* b^\alpha}{\Gamma(\alpha+1)} + c_b.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [0, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. By the previous inequality, we have

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds \quad \text{for } t \in [0,b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\phi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta = \delta(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^{\alpha}}{\Gamma(\alpha + 1)} \right] := \Lambda_b$$

Since $||z|| \le \mu(t)$, we have $||z|| \le \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Set

$$Z = \{ z \in \Omega^0 : \sup_{0 \le t \le b} |z(t)| \le \Theta_b + 1 \}.$$

Clearly, Z is a closed subset of Ω^0 .

We shall show that $F : Z \to \Omega^0$ is a contraction operator. Indeed, consider $z, \overline{z} \in Z$, thus using (H01) and (H03) for each $t \in [0, b]$

$$\begin{aligned} |F(z)(t) - F(\overline{z})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \right| \\ &- f(s, \overline{z}_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \right| \ ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ l_n(s) \ \|z_{\rho(s,z_s+x_s)} - \overline{z}_{\rho(s,z_s+x_s)} \|_{\mathcal{B}} \ ds. \end{aligned}$$

Using (H_{ϕ}) and Lemma 1.7.2, we obtain

$$|F(z)(t) - F(\overline{z})(t)| \leq \frac{\widehat{M}l_n^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} K_b |z(s) - \overline{z}(s)| ds$$

$$\leq \frac{K_b \widehat{M}l_n^* b^{\alpha}}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)||.$$

Therefore,

$$\|F(z) - F(\overline{z})\| \le \frac{K_b \widehat{M} l_n^* b^{\alpha}}{\Gamma(\alpha + 1)} \|z - \overline{z}\|$$

So, for $\frac{K_b \widehat{M} l_n^* b^{\alpha}}{\Gamma(\alpha+1)} < 1$ the operator F is a contraction. From the choice of Z there is no $z \in \partial Z^n$ such that $z = \lambda F(z), \ \lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator F has a unique fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), \ t \in (-\infty, b]$ is a fixed point of the operator N, which is the unique mild solution of the problem (2.5) – (2.6).

2.5 Neutral Problem with Infinite Delay

We give here an extension to previous results for the neutral case (2.7) - (2.8). Firstly, we define its mild solution.

Lemma 2.5.1. The system (2.7) - (2.8) is equivalent to the nonlinear integral equation

$$y(t) = [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \qquad (2.22)$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) ds$$

In other words, every solution of the integral equation (2.22) is also solution of the system (2.7) - (2.8).

Proof. It can be proved by applying the integral operator to both sides of the system (2.7) - (2.8), and using some classical results from fractional calculus to get (2.22).

Definition 2.5.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (2.7)-(2.8) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s,y_{\rho(s,y_s)}) \, ds, \quad t \in J.$$
(2.23)

We consider the hypotheses (H_{ϕ}) , (H01) - (H03) and we need the following assumptions : (H04) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H05) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

 $|A(t) g(t, \varphi)| \leq L (||\phi|| + 1)$ for all $t \in J$ and $\phi \in \mathcal{B}$.

(H06) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\phi) - A(\overline{s})g(\overline{s},\overline{\phi})| \le L_{\star} (|s-\overline{s}| + \|\phi - \overline{\phi}\|)$$

for all $\phi, \overline{\phi} \in \mathcal{B}$.

Theorem 2.5.1. Suppose that the hypotheses (H01) - (H05) are satisfied and moreover

$$\left[\overline{M}_{0}L_{\star}K_{b} + \frac{\widehat{M}K_{b}L_{n}^{*}b^{\alpha}}{\Gamma(\alpha+1)}\right] < 1$$
(2.24)

where $l_n^* = \sup l_n(s)$. Then the problem (2.7) – (2.8) has a unique mild solution on $(-\infty, b]$.

Proof. Consider the operator $\widetilde{N} : \Omega \to \Omega$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(2.25)

Then, fixed points of the operator \widetilde{N} are mild solutions of the problem (2.7) - (2.8).

2.5 Frac. Neutral Evo. Equa. with Infinite S.D.D.

For $\phi \in \mathcal{B}$, we consider the function $x(.): (-\infty, b] \to E$ defined as below by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (2.23) if and only if z satisfies $z_0 = 0$ and

$$z(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) ds.$$

 Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}$$

Define the operator $\widetilde{F}: \Omega^0 \to \Omega^0$ by :

$$\widetilde{F}(z)(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) ds.$$
(2.26)

Obviously the operator \widetilde{N} has a fixed point is equivalent to \widetilde{F} has one, so it turns to prove that \widetilde{F} has a fixed point.

Let $z \in \Omega^{0}$ be be a possible fixed point of the operator. Then, using (H01) - (H06), we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})| + |U(t,0)g(0,\phi)| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + ||U(t,0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0,\phi)| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||z_{\rho(s,y_s)} + x_{\rho(s,y_s)}|| \right] ds \\ &\leq \overline{M}_0 L \left(||z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}||_{\mathcal{B}} + 1 \right) + \widehat{MM}_0 L \left(||\phi||_{\mathcal{B}} + 1 \right) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||z_{\rho(s,y_s)} + x_{\rho(s,y_s)}|| \right] ds \end{aligned}$$

Since $||z_{\rho(t,y_t)} + x_{\rho(t,y_t)}||_{\mathcal{B}} \le K_b |z(t)| + c_b$ we obtain

$$|z(t)| \leq \overline{M}_0 L(K_b|z(t)| + c_b + 1) + \widehat{MM}_0 L(\|\phi\|_{\mathcal{B}} + 1) + \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha + 1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds$$

where $p^* = \sup p(s), q^* = \sup q(s)$ Then

$$(1 - \overline{M}_0 L K_b)|z(t)| \leq \overline{M}_0 L(c_b + 1) + \widehat{M} \overline{M}_0 L(\|\phi\|_{\mathcal{B}} + 1) + \frac{\widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ + \frac{\widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (K_b |z(s)| + c_b) ds$$

 Set

$$\delta_b := c_b + \frac{K_b \overline{M}_0 L(c_b+1) + K_b \widehat{M} \overline{M}_0 L(\|\phi\|_{\mathcal{B}}+1)}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha+1)} + \frac{K_b \widehat{M} p^* b^{\alpha}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha+1)}.$$

Thus

$$K_b|z(t)| + c_b \leq \delta_b + \frac{K_b \widehat{M}q^*}{(1 - \overline{M}_0 L K_b)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (K_b|z(s)| + c_b) ds.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b$$

Let $t^* \in (-\infty, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^*}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^{\alpha}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha + 1)} \right] := \Lambda_b$$

Since $||z|| \le \mu(t)$, we have $||z|| \le \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Now, we shall show that $\widetilde{F}: Z \to \Omega^0$ is a contraction operator. Indeed, consider $z, \overline{z} \in Z$, thus for each $t \in [0, b]$

$$\begin{split} |\widetilde{F}(y)(t) - \widetilde{F}(\overline{y})(t)| &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} |f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)})| \\ &+ x_{\rho(s,y_s)}) - f(s, \overline{z}_{\rho(s,y_s)} + x_{\rho(s,y_s)})| \\ &\leq \|A^{-1}(t)\| \|A(t)g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - A(t)g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} l_n(s) \|z_{\rho(s,y_s)} - \overline{z}_{\rho(s,y_s)}\|_{\mathcal{B}} ds \\ &\leq \overline{M}_0 L_{\star} \|z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}\|_{\mathcal{B}} \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} l_n(s) \|z_{\rho(s,y_s)} - \overline{z}_{\rho(s,y_s)}\|_{\mathcal{B}} ds. \end{split}$$

Since $||z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}||_{\mathcal{B}} \leq K_b |z(t) - \overline{z}(t)|$ we obtain

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq \overline{M}_0 L_{\star} K_b ||z(t) - \overline{z}(t)|| + \frac{\widetilde{M} K_b L_n^* b^{\alpha}}{\Gamma(\alpha + 1)} ||z(t) - \overline{z}(t)|| \\ &\leq \left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} K_b L_n^* b^{\alpha}}{\Gamma(\alpha + 1)} \right] ||z(t) - \overline{z}(t)|| \end{split}$$

Therefore,

$$\|\widetilde{F}(y) - \widetilde{F}(\overline{y})\| \le \left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} K_b L_n^* b^{\alpha}}{\Gamma(\alpha+1)} \|z - \overline{z}\|\right]$$

So, for an appropriate choice of $\overline{M}_0 L_{\star}, L_n^*$ and b^{α} such that

$$\left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} K_b L_b^* b^{\alpha}}{\Gamma(\alpha+1)}\right] < 1,$$

the operator \widetilde{F} is a contraction. From the choice of Z there is no $z \in \partial Z^n$ such that $z = \lambda \widetilde{F}(z)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. A consequence of the nonlinear alternative of Frigon and Granas shows that (C1) holds. We deduce that the operator \widetilde{F} has a unique fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), t \in (-\infty, b]$ is a fixed point of the operator \widetilde{N} , which is the unique mild solution of the problem (2.7) – (2.8).

2.6 Examples

We give in this section four examples to illustrate the previous results. **Example 1.** Consider the partial differential equation

$$\begin{cases} {}^{c}D_{t}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0}a_{1}(s-t)u\left[s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \end{cases}$$
(2.27)
$$u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$

where $a_1(t,\xi)$ is a continuous function and is uniformly Hölder continuous in t $0 < \alpha \leq 1$; $a_1 : [-r,0] \to \mathbb{R}$; $a_2 : [0,\pi] \to \mathbb{R}$; $\rho_i : [0,b] \to \mathbb{R}$ for i = 1,2 are continuous functions. To study this system, we consider the space $E = L^2([0,\pi],\mathbb{R})$ and the operator $A: D(A) \subset E \to E$ given by Aw = w'' with

$$D(A) := \{ w \in E : w'' \in E, w(0) = w(\pi) = 0 \}$$

It is well known that A is the infinitesimal generator of an analytic semigroup $\{T(t)\}_{t\in[0,b]}$ on E, with compact resolvent. On the domain D(A), we define the operators $A(t): D(A) \subset E \to E$ by

$$A(t)x(\xi) = Ax(\xi) + a_0(t,\xi)x(\xi)$$

By assuming that $a_0(.,.)$ is continuous and that $a_0(t,\xi) \leq -\delta_0$ ($\delta_0 > 0$) for every $t \in [0,b], \xi \in [0,\pi]$, and specific case $\alpha = 1$ it follows that the system

$$u'(t) = A(t)u(t) \quad t \ge s; \quad u(s) = x \in E,$$

has an associated evolution family given by

$$U(t,s)x(\xi) = \left[T(t-s)exp\left(\int_s^t a_0(\tau,\xi)d\tau\right)x\right](\xi).$$

From this expression, it follows that U(t,s) is a compact linear operator and that

$$||U(t,s)|| \le e^{-(1+\delta_0)(t-s)}$$
 for every $(t,s) \in [0,b] \times [0,b]$; $s \le t$.

Theorem 2.6.1. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, the functions $\rho_i : [0,b] \to \mathbb{R}$ for $i = 1, 2, a_1 : [-r,0] \to \mathbb{R}$ and $a_2 : [0,\pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (2.27).

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (2.27) into the abstract system (2.1) - (2.2). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 2.2.1. From Remark 1.7.1, we have the following result.

Corollary 2.6.1. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a unique mild solution of (2.27) on [-r, b].

Example 2. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{t}^{\alpha} \left[u(t,\xi) - \int_{-r}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(2.28)

where $a_3: [-r, 0] \to \mathbb{R}$ is a continuous function

Theorem 2.6.2. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, the functions $\rho_i : [0, b] \to \mathbb{R}$ for $i = 1, 2; a_1, a_3 : [-r, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (2.28).

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (2.28) into the abstract system (2.3) - (2.4). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 2.3.1. From Remark 1.7.1, we have the following result.

Corollary 2.6.2. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a unique mild solution of (2.28) on [-r, b].

Example 3. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0}a_{1}(s-t)u\left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(2.29)

where $a_1: (-\infty, 0] \to \mathbb{R}$ is a continuous functions.

Theorem 2.6.3. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, the functions $\rho_i : [0, +\infty[\rightarrow \mathbb{R} \text{ for } i = 1, 2; a_1 : (-\infty, 0] \rightarrow \mathbb{R} \text{ and } a_2 : [0, \pi] \rightarrow \mathbb{R}$ are continuous. Then there exists a unique mild solution of (2.29).

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (2.29) into the abstract system (2.5) - (2.6). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 2.4.1. From Remark 1.7.2, we have the following result.

Corollary 2.6.3. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a unique mild solution of (2.29) on $(-\infty, b]$.

Example 4. Consider the partial differential equation

$$\begin{aligned}
\left({}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-\infty}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\
&= \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\
&+ \int_{-\infty}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\
&\quad 0 \le t \le b, \ \xi \in [0,\pi], \\
v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\
v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi],
\end{aligned}$$
(2.30)

where $a_3: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 2.6.4. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, the functions $\rho_i : [0, b] \to \mathbb{R}$ for $i = 1, 2; a_1, a_3 : (-\infty, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (2.30).

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (2.30) into the abstract system (2.7) - (2.8). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 2.5.1. From Remark 1.7.2, we have the following result.

Corollary 2.6.4. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a unique mild solution of (2.30) on $(-\infty, b]$.

Chapter 3

Fractional Integrodifferential Evolution Equations with State-Dependent Delay

3.1 Introduction

Our attention in this chapter is to look for sufficient conditions for the existence of the unique mild solution for some classes of fractional order for partial functional and neutral functional integrodifferential evolution equations with finite and infinite state-dependent delay. Our analysis is based upon the nonlinear alternative of Frigon-Granas for contraction maps [52] combined with semigroup theory and the fractional version of Gronwall's inequality.

The existence of the unique mild solution is demonstrated in section 4.2 for the following class of fractional integrodifferential evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t,s)f(s,y_{\rho(s,y_{s})})ds, \quad 0 < \alpha < 1 \text{ a.e. } t \in J = [0,b]$$
(3.1)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0],$$
 (3.2)

where $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1)$, $\mathcal{K}: J \times J \to E, f: J \times C(H; E) \to E, \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 4.3, we consider the following fractional neutral integrodifferential evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t, s)f(s, y_{\rho(s, y_{s})})ds, \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J \quad (3.3)$$
$$y(t) = \varphi(t), \quad t \in H, \tag{3.4}$$

where $A(\cdot)$, f and φ are as in problem (3.1) – (3.2) and $g: J \times C(H; E) \to E$ is a given function.

In section 4.4 we study the following class of fractional integrodifferential evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t,s)f(s,y_{\rho(s,y_{s})})ds, \quad 0 < \alpha < 1 \text{ a.e. } t \in J$$
(3.5)

$$y_0 = \phi \in \mathcal{B},\tag{3.6}$$

where $\mathcal{K}: J \times J \to E$, \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f: J \times \mathcal{B} \to E$, $\rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 4.5, we consider the following fractional neutral integrodifferential evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + \int_{0}^{t} \mathcal{K}(t, s)f(s, y_{\rho(s, y_{s})})ds, \quad 0 < \alpha < 1 \quad \text{a.e.} \quad t \in J \quad (3.7)$$

 $y_0 = \phi \in \mathcal{B},\tag{3.8}$

where $A(\cdot)$, f, \mathcal{K} and ϕ are as in problem (3.5) – (3.6) and $g: J \times \mathcal{B} \to E$ is a given function. Finally, section 4.6 is devoted to examples illustrating the abstract theory considered in previous sections.

3.2 Partial Problem with Finite Delay

Before stating and proving the main result, we give first the definition of mild solution of the semilinear evolution problem (3.1) - (3.2).

Lemma 3.2.1. The system (3.1) - (3.2) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau ds.$$
(3.9)

In other words, every solution of the integral equation (3.9) is also solution of the system (3.1) - (3.2) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (3.1) - (3.2), and using some classical results from fractional calculus to get (3.9).

Definition 3.2.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (3.1) - (3.2)if $y(t) = \varphi(t)$ for all $t \in [-r, 0]$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_{\tau})}) d\tau \ ds \qquad \text{for each } t \in J.$$
(3.10)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(H;E), \rho(s,\varphi) \le 0\}.$$

We always assume that $\rho: J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter : $\widehat{}$

(H1) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le M$$
 for every $(t,s) \in \Delta$.

(H2) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in C(H; E)$.

(H3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v|$$

for all $u, v \in C(H; E)$ with $||u|| \le R$ and $||v|| \le R$.

(H4) For each $t \in J \mathcal{K}(t,s)$ is measurable on [0,t] and

$$\mathcal{K}(t) = ess \sup\{|\mathcal{K}(t,s)|; 0 \le s \le t\}$$

is bonded on [0, b]; let $S_b := \sup \mathcal{K}(t)$

Theorem 3.2.1. Assume that the hypotheses (H_{φ}) and (H1) - (H4) hold and moreover

$$\frac{\widehat{M}l_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} < 1 \tag{3.11}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (3.1) - (3.2) has a unique mild solution on [-r, b].

Proof. Transform the problem (3.1) - (3.2) into a fixed-point problem. Consider the operator $N: C([-r,b]; E) \to C([-r,b]; E)$ defined by :

$$N(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau \ ds, & \text{if } t \in J. \end{cases}$$

$$(3.12)$$

Clearly, fixed points of the operator N are mild solutions of the problem (3.1) - (3.2).

Let y be a possible solution of the problem (3.1) – (3.2). Given $t \leq b$, then from (H1) - (H2), (H4), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \int_0^s |\mathcal{K}(s,\tau)f(\tau,y_{\rho(\tau,y_{\tau})}) d\tau| \, ds \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t \int_0^s (t-s)^{\alpha-1} |\mathcal{K}(s,\tau)| \left[p(s) + q(s) \|y_{\rho(s,y_s)}\| \right] ds \end{aligned}$$

It follows that

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \le \widehat{M} \|\varphi\| + \mathcal{L}^{\varphi} \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^* b S_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[|y(s) + \mathcal{L}^{\varphi} \|\varphi\| \right] ds$$

 Set

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}q^*bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s)ds \quad \text{for } t \in [0,b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}q^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \right] \\ = \Lambda_b$$

Since $||y|| \leq \mu(t)$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Set

$$Y = \{ y \in C([-r, b]; E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1\}.$$

Clearly, Y is a closed subset of C([-r, b]; E).

We shall show that $N: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H1) and (H3) for each $t \in [0, b]$

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)|| \times \\ &\times \int_0^s |\mathcal{K}(s,\tau)| \left| f(\tau, y_{\rho(\tau,y_\tau)}) - f(\tau, \overline{y}_{\rho(\tau,y_\tau)}) \right| \, d\tau ds \\ &\leq \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \, l_b(s) \, ||y_{\rho(s,y_s)} - \overline{y}_{\rho(s,y_s)}|| \, ds. \end{aligned}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{\widehat{M}l_b^* b S_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &\leq \frac{\widehat{M}l_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \|y - \overline{y}\| \end{aligned}$$

Therefore,

$$\|N(y) - N(\overline{y})\| \le \frac{\widehat{M}l_n^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \|y - \overline{y}\|$$

So, the operator N is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y), \lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator N has a unique fixed-point y^* which is the unique mild solution of the problem (3.1) - (3.2).

3.3 Neutral Problem with Finite Delay

We give here an extension to previous results for the neutral case (3.3) - (3.4). Firstly, we define its mild solution.

Lemma 3.3.1. The system (3.3) - (3.4) is equivalent to the nonlinear integral equation

$$y(t) = [\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s)y(s) \, ds \qquad (3.13)$$

+
$$\frac{1}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1}\int_0^s \mathcal{K}(s,\tau)f(\tau,y_{\rho(\tau,y_{\tau})})d\tau ds.$$

In other words, every solution of the integral equation (3.22) is also solution of the system (3.3) - (3.4) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (3.3) - (3.4), and using some classical results from fractional calculus to get (3.22).

Definition 3.3.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (3.3) - (3.4)if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau ds.$$
(3.14)

We consider the hypotheses (H_{φ}) , (H1) - (H4) and we need to introduce the following assumptions :

(H5) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} \left(|s - \overline{s}| + \|\varphi - \overline{\varphi}\|\right)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

(H6) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all} \ t \in J.$$

(H7) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

 $|A(t) g(t, \varphi)| \leq L (||\varphi|| + 1)$ for all $t \in J$ and $\varphi \in C(H; E)$.

Theorem 3.3.1. Suppose that the hypotheses (H1) - (H7) are satisfied and moreover

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}\right] < 1 \tag{3.15}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (3.3) - (3.4) has a unique mild solution on [-r, b].

Proof. Transform as below the neutral problem (3.3) - (3.4) into a fixed point problem by considering the operator $\widetilde{N} : C([-r, b]; E) \to C([-r, b]; E)$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \left[\varphi(0) - g(0,\varphi)\right] + g(t, y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau ds, & \text{if } t \in J. \end{cases}$$
(3.16)

Clearly, the fixed points of the operator \widetilde{N} are mild solutions of the problem (3.3) - (3.4).

Let y be a possible solution of the problem (3.3) - (3.4). Given $t \leq b$. Then, using (H1) - (H5), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq |g(t, y_{\rho(t, y_t)})| + |U(t, 0)[\varphi(0) - g(0, \varphi)]| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \int_0^s \mathcal{K}(s, \tau) f(\tau, y_{\rho(\tau, y_\tau)}) d\tau ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t, y_{\rho(t, y_t)})| + ||U(t, 0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0, \varphi)| \\ &+ \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s) ||y_{\rho(s, y_s)}|| \right] ds \\ &\leq \widehat{M} ||\varphi|| + \overline{M}_0 L \left(||y_{\rho(t, y_t)}|| + 1 \right) + \widehat{M}\overline{M}_0 L \left(||\varphi|| + 1 \right) \\ &+ \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s) ||y_{\rho(s, y_s)}|| \right] ds \end{aligned}$$

Since $||y_{\rho(t,y_t)}|| \le |y(t)| + \mathcal{L}^{\varphi}||\varphi||$ we obtain

$$\begin{aligned} |y(t)| &\leq \overline{M}_0 L \left(|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \right) + \widehat{M} \|\varphi\| (1 + \overline{M}_0 L) + \widehat{M} \overline{M}_0 L + \frac{\widehat{M} p^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \\ &+ \frac{\widehat{M} q^* b S_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\| \right) ds \end{aligned}$$

Then

$$(1 - \overline{M}_{0}L)|y(t)| \leq \left[\overline{M}_{0}L\mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_{0}L)\right] \|\varphi\| + \widehat{M}\overline{M}_{0}L + \frac{\widehat{M}p^{*}b^{\alpha+1}S_{b}}{\Gamma(\alpha+1)} \\ + \frac{\widehat{M}q^{*}bS_{b}}{\Gamma(\alpha)} \int_{0}^{t} (t - s)^{\alpha-1} \left(|y(s)| + \mathcal{L}^{\varphi}\|\varphi\|\right) ds$$

 Set

$$\delta_b := \frac{\left[1 - \overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 - \overline{M}_0 L)\right]}{(1 - \overline{M}_0 L)} \|\varphi\| + \frac{\widehat{M}\overline{M}_0 L}{(1 - \overline{M}_0 L)} + \frac{\widehat{M}p^* b^{\alpha+1} S_b}{(1 - \overline{M}_0 L)\Gamma(\alpha+1)}.$$

Thus

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \leq \beta_b + \frac{\widehat{M}q^* bS_b}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|) ds$$

with $\beta_b = \mathcal{L}^{\varphi} \|\varphi\| + \delta_b$.

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \beta_b + \frac{\widehat{M}q^*bS_b}{(1-\overline{M}_0L)\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s)ds \quad \text{for } t \in [0,b]$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \beta_b \times \left[1 + \frac{\widehat{M}q^* b^{\alpha+1} S_b}{(1 - \overline{M}_0 L)\Gamma(\alpha+1)} \right]$$
$$= \Lambda_b$$

Since $||y|| \le \mu(t)$, we have $||y|| \le \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Now, we shall show that $\widetilde{N}: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus for each $t \in [0, b]$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq \overline{M}_0 L_\star |y(t) - \overline{y}(t)| + \frac{\widehat{M} L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \|y - \overline{y}\| \\ &\leq \left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \right] \|y - \overline{y}\| \end{split}$$

Therefore,

$$\|\widetilde{N}(y) - \widetilde{N}(\overline{y})\| \le \left[\overline{M}_0 L_\star + \frac{\widehat{M}L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}\right] \|y - \overline{y}\|.$$

So, for an appropriate choice of $\overline{M}_0 L_{\star}, L_b^*$ and b^{α} such that

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}\right] < 1,$$

the operator \widetilde{N} is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda \widetilde{N}(y)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. A consequence of the nonlinear alternative of Frigon and Granas shows that (C1) holds. We deduce that the operator \widetilde{N} has a unique fixed-point y^* which is the unique mild solution of the problem (3.3) - (3.4).

3.4 Partial Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of mild solution of the semilinear evolution problem (3.5) - (3.6).

Lemma 3.4.1. The system (3.5) - (3.6) is equivalent to the nonlinear integral equation

$$y(t) = \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau ds.$$
(3.17)

In other words, every solution of the integral equation (3.17) is also solution of the system (3.5) - (3.6) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (3.5) - (3.6), and using some classical results from fractional calculus to get (3.17).

Definition 3.4.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (3.5)-(3.6) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_{\tau})}) d\tau \ ds \qquad \text{for each } t \in J.$$
(3.18)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}.$$

We always assume that $\rho: J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter :

(H01) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H02)~ There exist two functions $p,q\in L^1_{loc}(J;\mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H03) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in \mathcal{B}$ with $||u|| \leq R$ and $||v|| \leq R$.

(H04) For each $t \in J \mathcal{K}(t,s)$ is measurable on [0,t] and

$$\mathcal{K}(t) = ess \sup\{|\mathcal{K}(t,s)|; 0 \le s \le t\}$$

is bonded on [0, b]; let $S_b := \sup \mathcal{K}(t)$

Consider the following space

$$\Omega = \left\{ y : (-\infty, b] \to E : y|_{(-\infty, 0]} \in B \text{ and } y|_{[0, b]} \text{ is continuous } \right\},$$

Theorem 3.4.1. Assume that the hypothesis (H_{φ}) and (H01) - (H03) hold and moreover

$$\frac{K_b \widehat{M} l_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} < 1 \tag{3.19}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (3.5) - (3.6) has a unique mild solution on $(-\infty, b]$.

Proof. We transform the problem (3.5) - (3.6) into a fixed-point problem. Consider the operator $N : \Omega \to \Omega$ defined by :

$$\int \phi(t), \quad \text{if } t \le 0;$$

$$N(y)(t) = \begin{cases} U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_{\tau})}) d\tau \ ds, & \text{if } t \in J. \end{cases}$$
(3.20)

Clearly, fixed points of the operator N are mild solutions of the problem (3.5) - (3.6).

For $\phi \in \mathcal{B}$, we will define the function $x(.): (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \in [0, b]; \\ U(t, 0) \ \phi(0), & \text{if } t \in (-\infty, 0]. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (3.18) if and only if z satisfies $z_0 = 0$ and

$$z(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) d\tau ds \quad \text{for } t \in J.$$

Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operator $F: \Omega^0 \to \Omega^0$ by :

$$F(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) d\tau ds \quad \text{for } t \in J.$$
(3.21)

Obviously the operator N has a fixed point is equivalent to F has one, so it turns to prove that F has a fixed point.

Let $z \in \Omega^{0}$ be be a possible fixed point of the operator. By the hypotheses (H01) and (H02), we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) \right| d\tau ds \\ &\leq \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)} \|_{\mathcal{B}} \right] ds \end{aligned}$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, b]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_b|x(s)| + (M_b + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\|U(s,0)\|_{B(E)}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\widehat{M}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_b|z(s)| + K_b\widehat{M}H\|\phi\|_{\mathcal{B}} + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi} + K_b\widehat{M}H)\|\phi\|_{\mathcal{B}} \end{aligned}$$

It follows that

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha+1}S_b}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left(K_b|z(s)| + (M_b + \mathcal{L}^\phi + K_b\widehat{M}H)\|\phi\|_{\mathcal{B}}\right) ds$$

Set $c_b := (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}}$. Then, we have

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha+1}S_b}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)|+c_b) ds$$

Then

$$K_b|z(t)| + c_b \leq \frac{K_b\widehat{M}p^*b^{\alpha+1}S_b}{\Gamma(\alpha+1)} + c_b + \frac{K_b\widehat{M}q^*bS_b}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds.$$

 Set

$$\delta_b := \frac{K_b \widehat{M} p^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} + c_b$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [0, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. By the previous inequality, we have

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^* b S_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds \quad \text{for } t \in [0,b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\phi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \right]$$
$$= \Lambda_b$$

Since $||y|| \leq \mu(t)$, we have $||y|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

 Set

$$Z = \{ z \in \Omega^0 : \sup_{0 \le t \le b} |z(t)| \le \Theta_b + 1 \}.$$

Clearly, Z is a closed subset of Ω^0 .

We shall show that $F: \mathbb{Z} \to \Omega^0$ is a contraction operator. Indeed, consider $z, \overline{z} \in \mathbb{Z}$, thus using (H01) and (H03) – (H04) for each $t \in [0, b]$

$$\begin{aligned} |F(z)(t) - F(\overline{z})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\| \int_0^s |\mathcal{K}(s,\tau)| \times \\ &\times \left| f(\tau, z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) - f(\tau, \overline{z}_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) \right| \ d\tau ds \\ &\leq \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ l_b(s) \ \|z_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)}\|_{\mathcal{B}} \ ds. \end{aligned}$$

Using (H_{ϕ}) and Lemma 1.7.2, we obtain

$$|F(z)(t) - F(\overline{z})(t)| \leq \frac{\widehat{M}bS_b l_b^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} K_b ||z(s) - \overline{z}(s)| ds$$

$$\leq \frac{K_b \widehat{M} l_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)||.$$

Therefore,

$$\|F(z) - F(\overline{z})\| \le \frac{K_b \widehat{M} l_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \|z - \overline{z}\|$$

So, the operator F is a contraction. From the choice of Z there is no $z \in \partial Z^n$ such that $z = \lambda F(z), \ \lambda \in (0,1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator F has a unique fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), \ t \in (-\infty, b]$ is a fixed point of the operator N, which is the unique mild solution of the problem (3.5) – (3.6).

3.5 Neutral Problem with Infinite Delay

We give here an extension to previous results for the neutral case (3.7) - (3.8). Firstly, we define its mild solution.

Lemma 3.5.1. The system (3.7) - (3.8) is equivalent to the nonlinear integral equation

$$y(t) = [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \qquad (3.22)$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau \, ds$$

In other words, every solution of the integral equation (3.22) is also solution of the system (3.7) - (3.8) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (3.7) - (3.8), and using some classical results from fractional calculus to get (3.22).

Definition 3.5.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (3.7)-(3.8) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau \ ds.$$
(3.23)

We consider the hypotheses (H_{ϕ}) , (H01) - (H04) and we need the following assumptions (H05) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} (|s-\overline{s}| + \|\varphi - \overline{\varphi}\|)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

(H06) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{ for all } t \in J.$$

(H07) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

$$|A(t) g(t, \varphi)| \le L (||\varphi|| + 1)$$
 for all $t \in J$ and $\varphi \in C(H; E)$.

Theorem 3.5.1. Suppose that the hypotheses (H01) - (H07) are satisfied and moreover

$$\left[\overline{M}_{0}L_{\star}K_{b} + \frac{\widehat{M}K_{b}L_{b}^{*}b^{\alpha+1}S_{b}}{\Gamma(\alpha+1)}\right] < 1$$
(3.24)

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (3.7) - (3.8) has a unique mild solution on $(-\infty, b]$.

Proof. Consider the operator $\widetilde{N} : \Omega \to \Omega$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, y_{\rho(\tau,y_\tau)}) d\tau ds, & \text{if } t \in J. \end{cases}$$
(3.25)

Then, fixed points of the operator \widetilde{N} are mild solutions of the problem (3.7) - (3.8).

For $\phi \in \mathcal{B}$, we consider the function $x(.): (-\infty, b] \to E$ defined as below by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (3.23) if and only if z satisfies $z_0 = 0$ and

$$z(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) d\tau ds.$$

 Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operator $\widetilde{F}: \Omega^0 \to \Omega^0$ by :

$$\widetilde{F}(z)(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) d\tau ds.$$
(3.26)

Obviously the operator \widetilde{N} has a fixed point is equivalent to \widetilde{F} has one, so it turns to prove that \widetilde{F} has a fixed point.

Let $z \in \Omega^0$ be be a possible fixed point of the operator. Then, using (H01) - (H04), we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})| + |U(t,0)g(0,\phi)| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) d\tau ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + ||U(t,0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0,\phi)| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \int_0^s \mathcal{K}(s,\tau) f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) d\tau ds \right| \\ &\leq \overline{M}_0 L \left(||z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}||_{\mathcal{B}} + 1 \right) + \widehat{M} \overline{M}_0 L \left(||\phi||_{\mathcal{B}} + 1 \right) \\ &+ \left| \frac{\widehat{M} bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||z_{\rho(s,y_s)} + x_{\rho(s,y_s)} || \right] ds \end{aligned}$$

Since $||z_{\rho(t,y_t)} + x_{\rho(t,y_t)}||_{\mathcal{B}} \le K_b |z(t)| + c_b$ we obtain

$$\begin{aligned} |z(t)| &\leq \overline{M}_0 L(K_n | z(t)| + c_b + 1) + \widehat{MM}_0 L(\|\phi\|_{\mathcal{B}} + 1) + \frac{\widehat{M} p^* b^{\alpha + 1} S_b}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M} q^* b S_b}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (K_b | z(s)| + c_b) ds \end{aligned}$$

Then

$$(1 - \overline{M}_0 L K_b)|z(t)| \leq \overline{M}_0 L(c_b + 1) + \widehat{M} \overline{M}_0 L(\|\phi\|_{\mathcal{B}} + 1) + \frac{\widehat{M} p^* b^{\alpha + 1} S_b}{\Gamma(\alpha + 1)} + \frac{\widehat{M} q^* b S_b}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (K_b |z(s)| + c_b) ds$$

 Set

$$\delta_b := c_b + \frac{K_b \overline{M}_0 L(c_b+1) + K_b \widehat{M} \overline{M}_0 L(\|\phi\|_{\mathcal{B}}+1)}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha+1)} + \frac{K_b \widehat{M} p^* b^{\alpha+1} S_b}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha+1)}$$

Thus

$$K_b|z(t)| + c_b \leq \delta_b + \frac{K_b \widehat{M}q^* b S_b}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (K_b|z(s)| + c_b) ds$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in (-\infty, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^* b S_b}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b]$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^{\alpha+1} S_b}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha + 1)} \right] := \Lambda_b$$

Since $||z|| \le \mu(t)$, we have $||z|| \le \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Now, we shall show that $\widetilde{F}: Z \to \Omega^0$ is a contraction operator. Indeed, consider $z, \overline{z} \in Z$, thus for each $t \in [0, b]$

$$\begin{split} |\widetilde{F}(y)(t) - \widetilde{F}(\overline{y})(t)| &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\| \int_0^s |\mathcal{K}(s,\tau)| \left| f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) \right| \\ &- f(\tau, \overline{z}_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) \right| d\tau ds \\ &\leq \|A^{-1}(t)\| |A(t)g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - A(t)g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\| \int_0^s |\mathcal{K}(s,\tau)| \left| f(\tau, z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) \right| \\ &- f(\tau, \overline{z}_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)}) \right| d\tau ds \\ &\leq \overline{M}_0 L_{\star} \|z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}\|_{\mathcal{B}} \\ &+ \frac{\widehat{M}bS_b}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} l_b(s) \|z_{\rho(s,y_s)} - \overline{z}_{\rho(s,y_s)}\|_{\mathcal{B}} ds. \end{split}$$

Since $||z_{\rho(t,y_t)}||_{\mathcal{B}} \leq K_b|z(t)| + c_b$ we obtain

$$\begin{aligned} |\widetilde{F}(y)(t) - \widetilde{F}(\overline{y})(t)| &\leq \overline{M}_0 L_{\star} k_b ||z(t) - \overline{z}(t)|| + \frac{\widehat{M} K_b L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)|| \\ &\leq \left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} k_b L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)} \right] ||z(t) - \overline{z}(t)|| \end{aligned}$$

Therefore,

$$\|\widetilde{F}(y) - \widetilde{F}(\overline{y})\| \le \left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} K_b L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}\right] \|z - \overline{z}\|$$

So, for an appropriate choice of $\overline{M}_0 L_\star, L_b^\star$ and b^α such that

$$\left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} K_b L_b^* b^{\alpha+1} S_b}{\Gamma(\alpha+1)}\right] < 1,$$

the operator \widetilde{F} is a contraction. From the choice of Z there is no $z \in \partial Z^n$ such that $z = \lambda \widetilde{F}(z)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. A consequence of the nonlinear alternative of Frigon and Granas shows that (C1) holds. We deduce that the operator \widetilde{F} has an unique fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), t \in (-\infty, b]$ is a fixed point of the operator \widetilde{N} , which is the unique mild solution of the problem (3.7) – (3.8).

3.6 Examples

We give in this section four examples to illustrate the previous results. **Example 1.** Consider the partial differential equation

$$\begin{cases} {}^{c}D_{t}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{t}\eta(t,\tau)\int_{-r}^{0}a_{1}(s-t)u\left[s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]d\tau ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \\ -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(3.27)

where $a_1(t,\xi)$ is a continuous function and is uniformly Hölder continuous in t $0 < \alpha \leq 1$; $a_1 : [-r,0] \to \mathbb{R}$; $a_2 : [0,\pi] \to \mathbb{R}$; $\rho_i : [0,b] \to \mathbb{R}$ for i = 1,2 are continuous functions.

To study this system, we consider the space $E = L^2([0, \pi], \mathbb{R})$ and the operator $A : D(A) \subset E \to E$ given by Aw = w'' with

$$D(A) := \{ w \in E : w'' \in E, w(0) = w(\pi) = 0 \}$$

It is well known that A is the infinitesimal generator of an analytic semigroup $\{T(t)\}_{t\in[0,b]}$ on E, with compact resolvent. On the domain D(A), we define the operators $A(t): D(A) \subset E \to E$ by $A(t)x(\xi) = Ax(\xi) + a_0(t,\xi)x(\xi)$.

By assuming that $a_0(.,.)$ is continuous and that $a_0(t,\xi) \leq -\delta_0$ ($\delta_0 > 0$) for every $t \in [0,b], \xi \in [0,\pi]$, and specific case $\alpha = 1$ it follows that the system

$$u'(t) = A(t)u(t) \quad t \ge s; \quad u(s) = x \in E,$$

has an associated evolution family given by

$$U(t,s)x(\xi) = \left[T(t-s)exp\left(\int_s^t a_0(\tau,\xi)d\tau\right)x\right](\xi).$$

From this expression, it follows that U(t,s) is a compact linear operator and that

$$||U(t,s)|| \le e^{-(1+\delta_0)(t-s)}$$
 for every $(t,s) \in [0,b] \times [0,b]$; $s \le t$.

Theorem 3.6.1. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, the functions $\rho_i : [0,b] \to \mathbb{R}$ for $i = 1, 2; a_1 : [-r,0] \to \mathbb{R}$ and $a_2 : [0,\pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (3.27).

Proof. From the assumptions, we have that

$$\mathcal{K}(t,s) = \eta(t,s)$$
$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2 d\theta\right),\,$$

are well defined functions, which permit to transform system (3.27) into the abstract system (3.1) - (3.2). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 3.2.1. From Remark 1.7.1, we have the following result.

Corollary 3.6.1. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a unique mild solution of (3.27) on [-r, b].

Example 2. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{t}^{\alpha} \left[u(t,\xi) - \int_{-r}^{0} a_{3}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{t} \eta(t,\tau) \int_{-r}^{0} a_{1}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) d\tau ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(3.28)

where $a_3: [-r, 0] \to \mathbb{R}$ is a continuous function.

Theorem 3.6.2. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, the functions $\rho_i : [0,b] \to \mathbb{R}$ for i = 1,2; $a_i : [-r,0] \to \mathbb{R}$ for i = 1,3 and $a_2 : [0,\pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (3.28).

Proof. From the assumptions, we have that

$$\mathcal{K}(t,s) = \eta(t,s)$$

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$

$$g(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right)$$

are well defined functions, which permit to transform system (3.28) into the abstract system (3.3) - (3.4). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 3.3.1. From Remark 1.7.1, we have the following result.

Corollary 3.6.2. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a unique mild solution of (3.28) on [-r, b].

Example 3. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{t}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{t}\eta(t,\tau)\int_{-\infty}^{0}a_{1}(s-t)u\left[s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]d\tau ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \\ -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(3.29)

where $a_1: (-\infty, 0] \to \mathbb{R}$ is continuous function.

Theorem 3.6.3. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, the functions $\rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}$ for $i = 1, 2; a_1 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (3.29) on $] - \infty, b]$.

Proof. From the assumptions, we have that

$$\mathcal{K}(t,s) = \eta(t,s)$$
$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right)$$

are well defined functions, which permit to transform system (3.29) into the abstract system (3.5) - (3.6). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 3.4.1. From Remark 1.7.2, we have the following result.

Corollary 3.6.3. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (3.29) on $] - \infty, b]$.

Example 4. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-\infty}^{0} a_{3}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{t} \eta(t,\tau) \int_{-\infty}^{0} a_{1}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) d\tau ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(3.30)

where $a_3: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 3.6.4. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, the functions $\rho_i : [0, b] \to \mathbb{R}$ for i = 1, 2; $a_i : (-\infty, 0] \to \mathbb{R}$ for i = 1, 3 and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a unique mild solution of (3.30).

Proof. From the assumptions, we have that

$$\mathcal{K}(t,s) = \eta(t,s)$$
$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (3.30) into the abstract system (3.7) - (3.8). Moreover, the function f is bounded linear operator. Now, the existence of the unique mild solution can be deduced from a direct application of Theorem 3.5.1. From Remark 1.7.2, we have the following result.

Corollary 3.6.4. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a unique mild solution of (3.30) on $(-\infty, b]$.

Chapter 4

Perturbed Evolution Equations with State-Dependent Delay

4.1 Introduction

In this chapter, we will demonstrate the existence of mild solutions for some classes of first order of partial functional and neutral functional differential perturbed evolution equations with infinite state-dependent delay on Fréchet spaces¹, then we will show the existence of mild solutions for some classes of Caputo's fractional derivative order for partial functional and neutral functional perturbed evolution equations with finite and infinite state-dependent delay² by using the nonlinear alternative of Avramescu for the sum of contraction and completely continuous operators on Banach spaces.

The existence of mild solutions on the real positif interval is demonstrated in section 3.2 for the following class of perturbed evolution equations with infinite state-dependent delay

$$y'(t) = A(t)y(t) + f(t, y_{\rho(t,y_t)}) + h(t, y_{\rho(t,y_t)}), \quad \text{a.e.} \ t \ge 0,$$
(4.1)

$$y_0 = \phi \in \mathcal{B} \tag{4.2}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $f, h : \mathbb{R}^+ \times \mathcal{B} \to E, \rho : \mathbb{R}^+ \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{0 \le t < +\infty}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in\mathbb{R}^+\times\mathbb{R}^+}$ for $s \le t$.

An extension of this problem is given in section 3.3, we consider the following class of neutral perturbed evolution equations with infinite state-dependent delay

$$\frac{d}{dt}[y(t) - g(t, y_{\rho(t, y_t)})] = A(t)y(t) + f(t, y_{\rho(t, y_t)}) + h(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \quad t \ge 0,$$
(4.3)

$$y_0 = \phi \in \mathcal{B} \tag{4.4}$$

where $A(\cdot)$, f, h and ϕ are as in problem (4.1) – (4.2) and $g: \mathbb{R}^+ \times \mathcal{B} \to E$ is a given function.

¹[10] **D. Aoued** and S. Baghli-Bendimerad, Mild solutions for perturbed evolution equations with infinite state-dependent delay, *Electronic Journal of Qualitative Theory of Differential Equations* 2013, No. 59, 1–24.

²[13] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Fractional perturbed partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.

In section 3.4, we give the existence of the unique mild solution of the following class of fractional perturbed evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t,y_{t})}) + h(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \ t \in J = [0, b],$$
(4.5)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0],$$
(4.6)

where $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0,1)$, $f,h: J \times C(H;E) \to E, \rho: J \times C(H;E) \to \mathbb{R}$ and $\varphi \in C(H;E)$ are given functions and $\{A(t)\}_{t\in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 3.5, we consider the following class of fractional neutral perturbed evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}) + h(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J,$$
(4.7)

$$y(t) = \varphi(t), \quad t \in H, \tag{4.8}$$

where $A(\cdot)$, f and ϕ are as in problem (4.5) – (4.6) and $g: J \times C(H; E) \to E$ is a given function.

In section 3.6, we investigate the existence of mild solutions of the following class of fractional perturbed evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + f(t, y_{\rho(t,y_{t})}) + h(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \ t \in J,$$

$$(4.9)$$

$$y_0 = \phi \in \mathcal{B} \tag{4.10}$$

where \mathcal{B} is an abstract *phase space* to be specified later, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $f, h: J \times \mathcal{B} \to E, \rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generates an evolution system of operators $\{U(t, s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 3.7, we consider the following class of fractional neutral perturbed evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + f(t, y_{\rho(t, y_{t})}) + h(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J,$$
(4.11)

$$y_0 = \phi \in \mathcal{B} \tag{4.12}$$

where $A(\cdot)$, f and ϕ are as in problem (4.9) – (4.10) and $g: J \times \mathcal{B} \to E$ is a given function. Finally, in section 3.8 we give examples to illustrate the abstract theory presented in previous sections.

4.2 Partial Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of a mild solution of the perturbed semilinear evolution problem (4.1) - (4.2).

Definition 4.2.1. We say that the function $y : \mathbb{R} \to E$ is a mild solution of (4.1) - (4.2) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \phi(0) + \int_0^t U(t,s) \ [f(s, y_{\rho(s,y_s)}) + h(s, y_{\rho(s,y_s)})] \ ds \quad a.e. \ t \ge 0.$$
(4.13)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in \mathbb{R}^{+} \times \mathcal{B}, \rho(s,\phi) \le 0\}$$

We always assume that $\rho : \mathbb{R}^+ \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter

- (H0) U(t,s) is compact for t-s > 0.
- (H1) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $0 \le s \le t < +\infty$.

(H2) There exists a function $p \in L^1_{loc}(J; \mathbb{R}_+)$ and a continuous nondecreasing function ψ : $\mathbb{R}_+ \to (0, \infty)$ and such that

 $|f(t, u)| \le p(t) \ \psi(||u||_{\mathcal{B}})$ for a.e. $t \ge 0$ and each $u \in \mathcal{B}$.

(H3) For all R > 0, there exists $l_R \in L^1_{loc}(\mathbb{R}^+; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||_{\mathcal{B}}$$

for all $u, v \in \mathcal{B}$ with $||u||_{\mathcal{B}} \leq R$ and $||v||_{\mathcal{B}} \leq R$.

(H4) There exists a function $\eta \in L^1(J, \mathbb{R}_+)$ such that

 $|h(t,u) - h(t,v)| \le \eta(t) \quad ||u - v||_{\mathcal{B}} \quad a.e. \ t \in J \ et \ \forall u, v \in \mathcal{B}.$

Consider the following space

$$B_{+\infty} = \left\{ y : \mathbb{R} \to E : y|_{[0,T]} \text{ continuous for } T > 0 \text{ and } y_0 \in \mathcal{B} \right\},\$$

where $y|_{[0,T]}$ is the restriction of y to the real compact interval [0,T].

Let us fix $\tau > 1$. For every $n \in \mathbb{N}$, we define in $B_{+\infty}$ the semi-norms by

$$||y||_n := \sup \{ e^{-\tau L_n^*(t)} |y(t)| : t \in [0, n] \}$$

where $L_n^*(t) = \int_0^t \overline{l}_n(s) \, ds$, $\overline{l}_n(t) = K_n \widehat{M} l_n(t)$ and l_n is the function from (H3). Then $B_{+\infty}$ is a Fréchet space with the family of semi-norms $\|\cdot\|_{n\in\mathbb{N}}$.

Proposition 4.2.1. By (H_{ϕ}) , and Lemma 1.7.2 and the assumption (A_1) , we have for each $t \in [0, n]$ and $n \in \mathbb{N}$

$$\|y_{\rho(t,y_t)}\| \le K_n |y(t)| + (M_n + \mathcal{L}^{\varphi}) \|y_0\|_{\mathcal{B}}$$

Theorem 4.2.1. Assume that the hypotheses (H_{ϕ}) , (H0) - (H2) and (H4) hold and moreover for all $n \in \mathbb{N}$, we have

$$\int_{\sigma_n}^{+\infty} \frac{ds}{s + \psi(s)} > K_n \widehat{M} \int_0^n \max(p(s); \eta(s)) \, ds.$$
(4.14)

with $\sigma_n = (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}} + K_n \widehat{M} \int_0^n |h(s,0)| \, ds.$ Then the problem (4.1) – (4.2) has a mild solution on $(-\infty, +\infty)$.

Proof. We transform the problem (4.1) - (4.2) into a fixed-point problem. Consider the operator $N: B_{+\infty} \to B_{+\infty}$ defined by

$$N(y)(t) = \begin{cases} \phi(t), & \text{if } t \in \mathbb{R}^{-}; \\ U(t,0) \ \phi(0) + \int_{0}^{t} U(t,s) \ f(s, y_{\rho(s,y_{s})}) \ ds, \\ + \int_{0}^{t} U(t,s) \ h(s, y_{\rho(s,y_{s})}) \ ds, & \text{if } t \ge 0. \end{cases}$$

Clearly, fixed points of the operator N are mild solutions of the problem (4.1) - (4.2).

For $\phi \in \mathcal{B}$, we will define the function $x(.) : \mathbb{R} \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0), & \text{if } t \ge 0. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in B_{+\infty}$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (4.13) if and only if z satisfies $z_0 = 0$ and

$$z(t) = \int_0^t U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds + \int_0^t U(t,s) h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds.$$

Let

$$B^0_{+\infty} = \{ z \in B_{+\infty} : z_0 = 0 \}.$$

Define the operators $F,G:B^0_{+\infty}\to B^0_{+\infty}$ by

$$F(z)(t) = \int_0^t U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds$$

and

$$G(z)(t) = \int_0^t U(t,s) h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds.$$

Obviously the operator N has a fixed point is equivalent to F + G has one, so it turns to prove that F + G has a fixed point.

First, show that F is continuous and compact. **Step 1**: F continuous. Let $(z_n)_{n \in \mathbb{N}}$ a sequence in $B^0_{+\infty}$ such that $z_n \to z$ in $B^0_{+\infty}$. By the hypothesis (H1), we have

$$\begin{aligned} |F(z_n)(t) - F(z)(t)| &\leq \\ &\leq \int_0^t \|U(t,s)\|_{B(E)} |f(s, z_{n\rho(s, z_{ns} + x_s)} + x_{\rho(s, z_{ns} + x_s)}) - f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})| ds \\ &\leq \widehat{M} \int_0^t \left| f(s, z_{n\rho(s, z_{ns} + x_s)} + x_{\rho(s, z_{ns} + x_s)}) - f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \right| ds. \end{aligned}$$

Since f is continuous, by dominated convergence theorem of Lebesgue, we get

$$|F(z_n)(t) - F(z)(t)| \longrightarrow 0 \text{ when } n \longrightarrow +\infty$$

So F is continuous.

Step 2: Show that F transforms any bounded of $B_{+\infty}^0$ in a bounded set. For each d > 0, there exists a positive constant ξ such that for all $z \in B_d = \{z \in B_{+\infty}^0 : ||z||_n \leq d\}$ we get $||F(z)||_n \leq \xi$. Soit $z \in B_d$, from assumption (H1) and (H2), we have for each $t \in [0, n]$

$$|F(z)(t)| \leq \int_0^t ||U(t,s)||_{B(E)} |f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| ds$$

$$\leq \widehat{M} \int_0^t |p(s)| \psi(||z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}||_{\mathcal{B}}) ds.$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, n]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + (M_n + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_n |x(s)| + (M_n + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + K_n \|U(s,0)\|_{B(E)} |\phi(0)| + (M_n + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + K_n \widehat{M} |\phi(0)| + (M_n + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_n |z(s)| + K_n \widehat{M}H \|\phi\|_{\mathcal{B}} + (M_n + \mathcal{L}^{\phi}) \|\phi\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + (M_n + \mathcal{L}^{\phi} + K_n \widehat{M}H) \|\phi\|_{\mathcal{B}} \end{aligned}$$

Set $c_n := (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}}$ and $\delta_n := K_n d + c_n$. Then

$$\|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \le K_n |z(s)| + c_n \le \delta_n.$$
(4.15)

Using the nondecreasing character of ψ , we get for each $t \in [0, n]$

$$|F(z)(t)| \le \widehat{M}\psi(\delta_n) ||p||_{L^1} := \varrho.$$

So there is a positive constant ρ such that $||F(z)||_n \leq \rho$. Then $F(B_d) \subset B_{\rho}$.

Step 3 : F maps bounded sets into equi-continuous sets of $B^0_{+\infty}$. We consider B_d as in Step 2 and we show that $F(B_d)$ is equi-continuous. Let $\tau_1, \tau_2 \in J$ with $\tau_1 < \tau_2$ and $z \in B_d$.

$$\begin{aligned} |F(z)(\tau_2) - F(z)(\tau_1)| &\leq \int_0^{\tau_1} \|U(\tau_2, s) - U(\tau_1, s)\|_{B(E)} \|f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})\| ds \\ &+ \int_{\tau_1}^{\tau_2} \|U(\tau_2, s)\|_{B(E)} \|f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})\| ds. \end{aligned}$$

Then by (5.41) and the nondecreasing character of ψ we get

$$\begin{aligned} |F(z)(\tau_2) - F(z)(\tau_1)| &\leq \psi(\delta_n) \int_0^{\tau_1} \|U(\tau_2, s) - U(\tau_1, s)\|_{B(E)} \ p(s) \ ds \\ &+ \widehat{M}\psi(\delta_n) \int_{\tau_1}^{\tau_2} \ p(s) \ ds. \end{aligned}$$

Noting that $|F(z)(\tau_2) - F(z)(\tau_1)| \longrightarrow 0$ tends to zero as $\tau_2 - \tau_1 \longrightarrow 0$ independently of $z \in B_d$. The right-hand of the above inequality tends to zero as $\tau_2 - \tau_1 \longrightarrow 0$, since U(t, s) is a strongly continuous operator and the compactness of U(t, s) for t > s, implies the continuity in the uniform operator topology (voir [7, 90]). As a consequence of Steps 1 to 3 together with the Arzelá-Ascoli theorem it suffices to show that the operator F maps B_d into a precompact set in E.

Let $t \in J$ be fixed and let ε be a real number satisfying $0 < \varepsilon < t$. For $z \in B_d$ we define

$$F_{\varepsilon}(z)(t) = U(t, t - \varepsilon) \int_0^{t-\varepsilon} U(t - \varepsilon, s) C u_{z+x}(s) \, ds.$$

Since U(t,s) is a compact operator, the set $Z_{\varepsilon}(t) = \{F_{\varepsilon}(z)(t) : z \in B_d\}$ is pre-compact in E for every ε , $0 < \varepsilon < t$. Moreover, using the definition of w, we get

$$|F_{(z)}(t) - F_{\varepsilon}(z)(t)| \le \int_{t-\varepsilon}^{t} \|U(t,s)\|_{B(E)} \|f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})\| ds$$

Therefore the set $Z(t) = \{F(z)(t) : z \in B_d\}$ is totally bounded.

Hence the set $\{F(z)(t) : z \in B_d\}$. So, we deduce from Steps 1, 2 and 3 that F is a compact operator.

Step 4 : G is a contraction. Let $z, \overline{z} \in B^0_{+\infty}$. By the hypotheses (H1) and (H4), we get for each $t \in [0, n]$ and $n \in \mathbb{N}$

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \\ &\leq \int_0^t \|U(t,s)\|_{B(E)} \|h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) - h(s, \overline{z}_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})\| ds \\ &\leq \widehat{M} \int_0^t \eta(s) \|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)} - x_{\rho(s, z_s + x_s)}\|_{\mathcal{B}} ds \\ &\leq \widehat{M} \int_0^t \eta(s) \|z_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)}\|_{\mathcal{B}} ds. \end{aligned}$$
Use the inequality (4.15), to get

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \int_0^t \widehat{M} K_n \eta(s) |z(s) - (\overline{z})(s)| \, ds \\ &\leq \int_0^t \left[\overline{L}_n(s) e^{\tau L_n^*(s)} \right] \left[e^{-\tau L_n^*(s)} |z(s) - \overline{z}(s)| \right] \, ds \\ &\leq \int_0^t \left[\frac{e^{\tau L_n^*(s)}}{\tau} \right]' \, ds \, \|z - \overline{z}\|_n \\ &\leq \frac{1}{\tau} e^{\tau L_n^*(t)} \, \|z - \overline{z}\|_n. \end{aligned}$$

Therefore

$$\|G(z) - G(\overline{z})\|_n \le \frac{1}{\tau} \|z - \overline{z}\|_n.$$

Then the operator G is a contraction for all $n \in \mathbb{N}$.

Step 5: For applying Theorem (1.10.2), we must check (Av2): i.e. it remains to show that the set

$$\Sigma = \left\{ z \in B^0_{+\infty} : z = \lambda F(z) + \lambda G\left(\frac{z}{\lambda}\right) \quad \text{for some } \lambda \in]0,1[\right\}.$$

is bounded.

Let $z \in \Sigma$. By (H1) - (H2) and (H4), we have for each $t \in [0, n]$

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \int_0^t \|U(t,s)\|_{B(E)} \|f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})\| ds \\ &+ \int_0^t \|U(t,s)\|_{B(E)} \|h\left(s, \frac{z_{\rho(s, z_s + x_s)}}{\lambda}\right) - h(s, 0) + h(s, 0)\| ds \\ &\leq \widehat{M} \int_0^t \|p(s)\|\psi(\|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})\|_{\mathcal{B}} ds) \\ &+ \|\widehat{M} \int_0^t \|\eta(s)\| \|\frac{z_{\rho(s, z_s + x_s)}}{\lambda}\|_{\mathcal{B}} ds + \widehat{M} \int_0^t \|h(s, 0)\| ds. \end{aligned}$$

Use Proposition 4.2.1 and inequality (4.15)

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \widehat{M} \int_0^t p(s) \ \psi(K_n |z(s)| + c_n) \ ds \\ &+ \widehat{M} \int_0^t \eta(s) \ \left(\frac{K_n}{\lambda} |z(s)| + c_n\right) \ ds + \widehat{M} \int_0^t \ |h(s,0)| \ ds. \end{aligned}$$

We consider the function $u(t) := \sup_{\theta \in [0,t]} |z(\theta)|$. The use of nondecreasing character of ψ gives with the fact that $0 < \lambda < 1$

$$\frac{K_n}{\lambda}u(t) + c_n \leq c_n + K_n\widehat{M}\int_0^t p(s) \ \psi\left(\frac{K_n}{\lambda}u(s) + c_n\right) \ ds \\
+ K_n\widehat{M}\int_0^t \eta(s) \ \left(\frac{K_n}{\lambda}u(s) + c_n\right) \ ds + \widehat{M}\int_0^t |h(s,0)| \ ds.$$

Set $\sigma_n := c_n + K_n \widehat{M} \int_0^n |h(s, 0)| \, ds$. Then, we have

$$\frac{K_n}{\lambda}u(t) + c_n \leq \sigma_n + K_n \widehat{M} \int_0^t p(s) \psi\left(\frac{K_n}{\lambda}u(s) + c_n\right) ds
+ K_n \widehat{M} \int_0^t \eta(s) \left(\frac{K_n}{\lambda}u(s) + c_n\right) ds.$$

We consider the function μ defined by

 $\mu(t) = \sup \{ K_n u(s) + c_n : 0 \le s \le t \}, \quad 0 \le t \le +\infty$

Let $t^* \in [0, t]$ such that $\mu(t) = K_n u(t^*) + c_n$. From the previous inequality, we have for all $t \in [0, n]$

$$\mu(t) \le \sigma_n + K_n \widehat{M} \int_0^t p(s)\psi(\mu(s))ds + K_n \widehat{M} \int_0^t \eta(s)\mu(s)ds.$$

Let us take the right-hand side of the above inequality as v(t). Then, we have

$$\mu(t) \le v(t) \quad \forall t \in [0, n].$$

From the definition of v, we have

$$v(0) = \sigma_n$$
 and $v'(t) = K_n \widehat{M} \left[p(t)\psi(\mu(t)) + \eta(t)\mu(t) \right]$ a.e. $t \in [0, n]$.

Using the nondecreasing character of ψ we get

$$v'(t) \le K_n \widehat{M}[p(t)\psi(v(t)) + \eta(t)v(t)] \quad a.e. \ t \in [0, n]$$

This implies that for each $t \in [0, n]$ and using (4.14), we get

$$\int_{\sigma_n}^{v(t)} \frac{ds}{s + \psi(s)} \leq K_n \widehat{M} \int_0^t \max(p(s); \eta(s)) ds$$
$$\leq K_n \widehat{M} \int_0^n \max(p(s); \eta(s)) ds$$
$$< \int_{c_n}^{+\infty} \frac{ds}{s + \psi(s)}.$$

Thus, for every $t \in [0, n]$, there exists a constant Λ_n such that $v(t) \leq \Lambda_n$ and hence $\mu(t) \leq \Lambda_n$. Since $||z||_n \leq \mu(t)$, we have $||z||_n \leq \Lambda_n$. This shows that the set Σ is bounded. Then the statement (Av2) in Theorem 1.10.2 does not hold. The nonlinear alternative of Avramescu implies that (Av1) is satisfied, we deduce that the operator F + G has a fixed point z^* . Then $y^*(t) = z^*(t) + x(t), t \in] -\infty, +\infty[$ is a fixed point of the operator N which is a mild solution of the problem (4.1) - (4.2).

4.3 Neutral Problem with Infinite Delay

In this section, we give an existence result of mild solution for the problem (4.3) - (4.4). Firstly, we define the concept of the mild solution for that problem. **Definition 4.3.1.** We say that the function $y(\cdot) : \mathbb{R} \to E$ is a mild solution of (4.3) – (4.4) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \int_0^t U(t,s)A(s) \ g(s, y_{\rho(s,y_s)}) \ ds + \int_0^t U(t,s) \ [f(s, y_{\rho(s,y_s)}) + h(s, y_{\rho(s,y_s)})] \ ds \quad \forall t \ge 0.$$

$$(4.16)$$

We consider the hypotheses (H_{ϕ}) , (H0) - (H4) and we need the following assumptions (H5) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \ge 0.$$

(H6) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_n}$, such that

$$|A(t) g(t,\phi)| \leq L (||\phi||_{\mathcal{B}} + 1)$$
 for all $t \geq 0$ and $\phi \in \mathcal{B}$.

(H7) There exists a constant $L_* > 0$ such that

$$|A(s) g(s,\phi) - A(\overline{s}) g(\overline{s},\overline{\phi})| \le L_* (|s - \overline{s}| + \|\phi - \overline{\phi}\|_{\mathcal{B}})$$

for all $s, \overline{s} \ge 0$ and $\phi, \overline{\phi} \in \mathcal{B}$.

(H8) The function g is completely continuous and for any bounded set $Q \subset \mathcal{B}$ the set $\{t \longrightarrow g(t, x_{\rho(t,y_t)})\}$ is equi-continuous in \mathcal{B} .

Theorem 4.3.1. Suppose that the hypotheses (H_{ϕ}) , (H0) - (H8) are satisfied and moreover for all $n \in \mathbb{N}$, we have

$$\int_{\xi_n}^{+\infty} \frac{ds}{s + \psi(s)} > \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^n \max(L + \eta(s), p(s)) \ ds \tag{4.17}$$

with $c_n = (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}}$ and

$$\xi_n = c_n + K_n \frac{\left(\widehat{M} + 1\right) \overline{M}_0 L + \widehat{M} Ln + \overline{M}_0 L \left(c_n + \widehat{M}\right) \|\phi\|_{\mathcal{B}} + \widehat{M} \int_0^n |h(s, 0)| \, ds}{1 - \overline{M}_0 L K_n}$$

Then the problem (4.3) - (4.4) has a mild solution.

Proof. Consider the operator $\widetilde{N}: B_{+\infty} \to B_{+\infty}$ defined by

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \in \mathbb{R}^{-1}; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) \\ + \int_0^t U(t,s)A(s) \ g(s,y_{\rho(s,y_s)}) \ ds, \\ + \int_0^t U(t,s) \ [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] \ ds & \text{if } t \ge 0. \end{cases}$$

Then, fixed points of the operator \widetilde{N} are mild solutions of the problem (4.3) – (4.4).

For $\phi \in \mathcal{B}$, we consider the function $x(.) : \mathbb{R} \to E$ defined as below by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0), & \text{if } t \ge 0. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in B_{+\infty}$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (4.16) if and only if z satisfies $z_0 = 0$ and

$$\begin{aligned} z(t) &= g(t, z_{\rho(s, z_t + x_t)} + x_{\rho(s, z_t + x_t)}) - U(t, 0)g(0, \phi) \\ &+ \int_0^t U(t, s)A(s) \ g(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds \\ &+ \int_0^t U(t, s) \ f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds \\ &+ \int_0^t U(t, s) \ h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds. \end{aligned}$$

Let

$$B^0_{+\infty} = \{ z \in B_{+\infty} : z_0 = 0 \}.$$

Define the operators $F, \widetilde{G}: B^0_{+\infty} \to B^0_{+\infty}$ by

$$F(z)(t) = \int_0^t U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \ ds$$

And

$$\widetilde{G}(z)(t) = g(t, z_{\rho(s, z_t + x_t)} + x_{\rho(s, z_t + x_t)}) - U(t, 0)g(0, \phi) + \int_0^t U(t, s)A(s) \ g(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds + \int_0^t U(t, s) \ h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds.$$

Obviously the operator \widetilde{N} has a fixed point is equivalent to $F + \widetilde{G}$ has one, so it turns to prove that $F + \widetilde{G}$ has a fixed point.

We have shown that the operator F is continuous and compact as in Section 3.2 Remains to show that the operator \tilde{G} is a contraction.

Let
$$z, \overline{z} \in B_{+\infty}^0$$
. By $(H1)$, $(H4)$, $(H5)$ and $(H7)$, we have for each $t \in [0, n]$ and $n \in \mathbb{N}$
 $|\widetilde{G}(z)(t) - \widetilde{G}(\overline{z})(t)| \leq \leq |g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}) - g(t, \overline{z}_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + \int_0^t ||U(t,s)||_{B(E)} |A(s)[g(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) - g(s, \overline{z}_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})]| ds + \int_0^t ||U(t,s)||_{B(E)} |h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) - h(s, \overline{z}_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| ds \leq ||A^{-1}(t)||_{B(E)} |A(t)g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}) - A(t)g(t, \overline{z}_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + \int_0^t \widehat{M} |A(s)g(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) - A(s)g(s, \overline{z}_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| ds + \int_0^t \widehat{M} |h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) - h(s, \overline{z}_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| ds \leq \overline{M_0}L_* ||z_{\rho(t,z_t+x_t)} - \overline{z}_{\rho(t,z_t+x_t)} ||_{\mathcal{B}} + \int_0^t \widehat{M} L_* ||z_{\rho(s,z_s+x_s)} - \overline{z}_{\rho(s,z_s+x_s)} - \overline{z}_{\rho(s,z_s+x_s)} ||_{\mathcal{B}} ds + \int_0^t \widehat{M} \eta(s) ||z_{\rho(s,z_s+x_s)} - \overline{z}_{\rho(s,z_s+x_s)} ||_{\mathcal{B}} ds.$

Use the inequality (4.15), we get

$$\begin{aligned} |\widetilde{G}(z)(t) - \widetilde{G}(\overline{z})(t)| &\leq \overline{M_0} L_* K_n |z(t) - \overline{z}(t)| + \int_0^t \widehat{M} L_* K_n |z(s) - \overline{z}(s)| \ ds \\ &+ \int_0^t \widehat{M} K_n \ \eta(s) |z(s) - \overline{z}(s)| \ ds \\ &\leq \overline{M_0} L_* K_n |z(t) - \overline{z}(t)| + \int_0^t \widehat{M} K_n [L_* + \eta(s)] |z(s) - \overline{z}(s)| \ ds. \end{aligned}$$

Set $\overline{l}_n(t) = \widehat{M}K_n[L_* + \eta(t)]$ for the family of semi-norms $\{\|\cdot\|_{n\in\mathbb{N}}\}$, then

$$\begin{split} |\widetilde{G}(z)(t) - \widetilde{G}(\overline{z})(t)| &\leq \overline{M}_{0}L_{*}K_{n} |z(t) - \overline{z}(t)| + \int_{0}^{t} \overline{l}_{n}(s) |z(s) - \overline{z}(s)| ds \\ &\leq [\overline{M}_{0}L_{*}K_{n} \ e^{\tau \ L_{n}^{*}(t)}] \ \left[e^{-\tau \ L_{n}^{*}(t)} \ |z(t) - \overline{z}(t)|\right] \\ &+ \int_{0}^{t} \left[\overline{l}_{n}(s) \ e^{\tau \ L_{n}^{*}(s)}\right] \ \left[e^{-\tau \ L_{n}^{*}(s)} \ |z(s) - \overline{z}(s)|\right] ds \\ &\leq \overline{M}_{0}L_{*}K_{n} \ e^{\tau \ L_{n}^{*}(t)} \ \|z - \overline{z}\|_{n} + \int_{0}^{t} \left[\frac{e^{\tau \ L_{n}^{*}(s)}}{\tau}\right]' \ ds \ \|z - \overline{z}\|_{n} \\ &\leq \left[\overline{M}_{0}L_{*}K_{n} + \frac{1}{\tau}\right] \ e^{\tau \ L_{n}^{*}(t)} \ \|z - \overline{z}\|_{n}. \end{split}$$

Therefore

$$\|\widetilde{G}(z) - \widetilde{G}(\overline{z})\|_n \le \left[\overline{M_0}L_*K_n + \frac{1}{\tau}\right] \|z - \overline{z}\|_n$$

Let us fix $\tau > 0$ and assume that

$$\overline{M}_0 L_* K_n + \frac{1}{\tau} < 1,$$

then the operator \widetilde{G} is a contraction for all $n \in \mathbb{N}$.

For applying Theorem (1.10.2), we must check (Av2) i.e. it remains to show that the set

$$\widetilde{\Sigma} = \left\{ z \in B^0_{+\infty} : z = \lambda F(z) + \lambda \widetilde{G}\left(\frac{z}{\lambda}\right) \text{ pour } 0 < \lambda < 1 \right\}.$$

is bounded.

Let $z \in \widetilde{\Sigma}$. By (H1) - (H2), we have for each $t \in [0, n]$

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq \|A^{-1}(t)\| |A(t)g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)})| + \widehat{M}\|A^{-1}(0)\| |A(0)g(0, \phi)| \\ &+ \widehat{M} \int_0^t |A(s)g(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})| ds \\ &+ \widehat{M} \int_0^t p(s) \psi\left(\left\| z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)} \right\| \right) ds \\ &+ \widehat{M} \int_0^t \left| h\left(s, \frac{z_{\rho(s, \frac{z_s}{\lambda} + x_s)}}{\lambda} + x_{\rho(s, \frac{z_s}{\lambda} + x_s)}\right) - h(s, 0) \right| ds + \widehat{M} \int_0^n |h(s, 0)| ds \end{aligned}$$

Using assumptions (H5) - (H6) and (H4)

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq \overline{M}_{0}L(\|z_{\rho(t,z_{t}+x_{t})} + x_{\rho(t,z_{t}+x_{t})}\|_{\mathcal{B}} + 1) + \widehat{MM}_{0}L(\|\phi\|_{\mathcal{B}} + 1) \\ &+ \widehat{M}L \int_{0}^{t} (\|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}} + 1) ds \\ &+ \widehat{M}\int_{0}^{t} p(s) \psi(\|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}} ds + \widehat{M}\int_{0}^{t} |h(s,0)| ds \\ &+ \widehat{M}\int_{0}^{t} \eta(s) \left\|\frac{\widetilde{z}_{\rho(s,z_{s}+x_{s})}}{\lambda} + x_{\rho(s,z_{s}+x_{s})}\right\|_{\mathcal{B}} ds + \widehat{M}\int_{0}^{t} |h(s,0)| ds \\ &\leq (\widehat{M}+1)\overline{M}_{0}L + \widehat{M}Ln + \widehat{MM}_{0}L\|\phi\|_{\mathcal{B}} + \widehat{M}\int_{0}^{n} |h(s,0)| ds \\ &+ \overline{M}_{0}L\|z_{\rho(t,z_{t}+x_{t})} + x_{\rho(t,z_{t}+x_{t})}\|_{\mathcal{B}} + \widehat{M}L\int_{0}^{t} \|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}} ds \\ &+ \widehat{M}\int_{0}^{t} p(s) \psi(\|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}} ds \\ &+ \widehat{M}\int_{0}^{t} \eta(s) \left\|\frac{\widetilde{z}_{\rho(s,\overline{\lambda}} + x_{s})}{\lambda} + x_{\rho(s,\overline{\lambda} + x_{s})}\right\|_{\mathcal{B}} ds. \end{aligned}$$

Use Proposition (4.2.1) and inequality (4.15) for get

$$\begin{aligned} \left\| \frac{z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}}{\lambda}+x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}\right\|_{\mathcal{B}} &\leq \frac{1}{\lambda} \left\| z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}\right\|_{\mathcal{B}}+\left\| x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}\right\|_{\mathcal{B}} \\ &\leq \frac{K_{n}}{\lambda}|z(s)|+\frac{M_{n}+\mathcal{L}^{\phi}}{\lambda}\|z_{0}\|_{\mathcal{B}}+K_{n}|x(s)|+\left(M_{n}+\mathcal{L}^{\phi}\right)\|x_{0}\|_{\mathcal{B}} \\ &\leq \frac{K_{n}}{\lambda}|z(s)|+K_{n}\|U(s,0)\|_{\mathcal{B}(E)}|\phi(0)|+\left(M_{n}+\mathcal{L}^{\phi}\right)\|\phi\|_{\mathcal{B}} \\ &\leq \frac{K_{n}}{\lambda}|z(s)|+K_{n}\widehat{M}H\|\phi\|_{\mathcal{B}}+\left(M_{n}+\mathcal{L}^{\phi}\right)\|\phi\|_{\mathcal{B}} \\ &\leq \frac{K_{n}}{\lambda}|z(s)|+\left(M_{n}+\mathcal{L}^{\phi}+K_{n}\widehat{M}H\right)\|\phi\|_{\mathcal{B}}. \end{aligned}$$

Hence

$$\left\|\frac{z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}}{\lambda}+x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}\right\|_{\mathcal{B}} \leq \frac{K_{n}}{\lambda}|z(s)|+c_{n}.$$
(4.18)

Use the function $u(\cdot)$ and the nondecreasing character of ψ to get

$$\frac{u(t)}{\lambda} \leq \left(\widehat{M}+1\right)\overline{M}_{0}L + \widehat{M}Ln + \widehat{M}\overline{M}_{0}L \|\phi\|_{\mathcal{B}} + \widehat{M}\int_{0}^{n} |h(s,0)| ds
+ \overline{M}_{0}L \left(K_{n}u(t)+c_{n}\right) + \widehat{M}L\int_{0}^{t} \left(K_{n}u(s)+c_{n}\right) ds
+ \widehat{M}\int_{0}^{t} p(s) \psi\left(K_{n}u(s)+c_{n}\right) ds + \widehat{M}\int_{0}^{t} \eta(s) \left(\frac{K_{n}}{\lambda}u(s)+c_{n}\right) ds.$$

Then

$$\frac{u(t)}{\lambda} \leq \left(\widehat{M}+1\right)\overline{M}_{0}L + \widehat{M}Ln + \overline{M}_{0}L\left[M_{n} + \mathcal{L}^{\phi} + \widehat{M}(1+K_{n}H)\right] \|\phi\|_{\mathcal{B}} \\
+ \widehat{M}\int_{0}^{n} |h(s,0)| \, ds + \overline{M}_{0}L \, \frac{K_{n}}{\lambda}u(t) + \widehat{M}L\int_{0}^{t} \left(\frac{K_{n}}{\lambda}u(s) + c_{n}\right) \, ds \\
+ \widehat{M}\int_{0}^{t} p(s) \, \psi\left(\frac{K_{n}}{\lambda}u(s) + c_{n}\right) \, ds + \widehat{M}\int_{0}^{t} \eta(s) \, \left(\frac{K_{n}}{\lambda}u(s) + c_{n}\right) \, ds.$$

 Set

$$\zeta_n := \left(\widehat{M} + 1\right) \overline{M}_0 L + \widehat{M} L n + \overline{M}_0 L \left[M_n + \mathcal{L}^\phi + \widehat{M}(1 + K_n H)\right] \|\phi\|_{\mathcal{B}} + \widehat{M} \int_0^n |h(s, 0)| \, ds$$

 So

$$\frac{K_n}{\lambda} \left(1 - \overline{M}_0 L K_n \right) u(t) \leq K_n \zeta_n + K_n \widehat{M} \int_0^t \left[L + \eta(s) \right] \left(\frac{K_n}{\lambda} u(s) + c_n \right) ds \\
+ K_n \widehat{M} \int_0^t p(s) \psi \left(\frac{K_n}{\lambda} u(s) + c_n \right) ds.$$

Set $\xi_n := c_n + \frac{K_n \zeta_n}{1 - \overline{M}_0 L K_n}$. Then

$$\frac{K_n}{\lambda}u(t) + c_n \leq \xi_n + \frac{K_n\widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^t [L + \eta(s)] \left(\frac{K_n}{\lambda}u(s) + c_n\right) ds \\
+ \frac{K_n\widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^t p(s) \psi\left(\frac{K_n}{\lambda}u(s) + c_n\right) ds.$$

We consider the function μ defined by

$$\mu(t) = \sup \left\{ \frac{K_n}{\lambda} u(s) + c_n : 0 \le s \le t \right\}, \quad 0 \le t \le +\infty$$

Let $t^* \in [0,t]$ be such that $\mu(t) = \frac{K_n}{\lambda}u(t^*) + c_n$. By the previous inequality, we have for $t \in [0,n]$

$$\mu(t) \le \xi_n + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^t \left[L + \eta(s) \right] \ \mu(s) \ ds + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^t \ p(s) \ \psi(\mu(s)) \ ds.$$

Let us take the right-hand side of the above inequality as v(t). Then we have

$$\mu(t) \le v(t) \quad \forall t \in [0, n].$$

From the definition of v, we get $v(0) = \xi_n$ and

$$v'(t) = \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \ [L + \eta(t)] \ \mu(t) + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \ p(t) \ \psi(\mu(t)) \quad \text{a.e.} \ t \in [0, n].$$

Using the nondecreasing character of ψ we have

$$v'(t) \leq \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \quad [L + \eta(t)] \quad v(t) + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \quad p(t) \quad \psi(v(t)) \quad \text{ a.e. } t \in [0, n].$$

This implies that for each $t \in [0, n]$ and using (4.17) we get

$$\int_{\xi_n}^{v(t)} \frac{ds}{s + \psi(s)} \leq \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \int_0^t \max(L + \eta(s), p(s)) ds$$
$$\leq \frac{\widehat{M} K_n}{1 - \overline{M}_0 L K_n} \int_0^n \max(L + \eta(s), p(s)) ds$$
$$< \int_{\xi_n}^{+\infty} \frac{ds}{s + \psi(s)}.$$

Thus, for every $t \in [0, n]$, there exists a constant Λ_n such that $v(t) \leq \Lambda_n$ and hence $\mu(t) \leq \Lambda_n$. Since $\|z\|_n \leq \mu(t)$, we have $\|z\|_n \leq \Lambda_n$. This shows that the set $\widetilde{\Sigma}$ is bounded. Then the statement (Av2) in Theorem 1.10.2 does not hold. The nonlinear alternative of Avramescu implies that (Av1) is satisfied. we deduce that the operator $F + \widetilde{G}$ has a fixed point z^* . Then $y^*(t) = z^*(t) + x(t), t \in] -\infty, +\infty[$ is a fixed point of the operator N which is a mild solution of the problem (4.3) - (4.4).

4.4 Fractional Partial Problem with Finite Delay

Before stating and proving the main result, we give first the definition of the unique mild solution of the perturbed fractional problem (4.5) - (4.6).

Lemma 4.4.1. The system (4.5) - (4.6) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} [f(s, y_{\rho(s,y_s)}) + h(s, y_{\rho(s,y_s)})] \, ds.$$
(4.19)

In other words, every solution of the integral equation (4.19) is also solution of the system (4.5) - (4.6) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (4.5) - (4.6), and using some classical results from fractional calculus to get (4.19).

Definition 4.4.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (4.5) - (4.6)if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] \ ds \quad \text{for each } t \in J.$$
(4.20)

 Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(H;E), \rho(s,\varphi) \le 0\}.$$

We always assume that $\rho: J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter

- (H'0) U(t,s) is compact for t-s > 0.
- $(H'1)\,$ There exists a constant $\widehat{M}\geq 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $s \le t$.

(H'2) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

 $|f(t,u)| \le p(t) + q(t) ||u||_B$ for a.e. $t \in J$ and each $u \in C(H; E)$.

(H'3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in C(H; E)$ with $||u|| \leq R$ and $||v|| \leq R$.

(H'4) There exists a function $\eta \in L^1(J, \mathbb{R}_+)$ such that

$$|h(t, u) - h(t, v)| \le \eta(t) ||u - v||$$
 a.e. $t \in J$ et $\forall u, v \in C(H; E)$.

Theorem 4.4.1. Assume that the hypotheses (H_{φ}) , (H'0) - (H'4) hold and moreover for all $n \in \mathbb{N}$,

$$\frac{\widehat{M}(l_b^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)} < 1 \tag{4.21}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $\eta^* = \sup \eta(s)$. Then the problem (4.5) - (4.6) has a mild solution on [-r, b].

Proof. We transform the problem (4.5) - (4.6) into a fixed-point problem. Consider the operator $N: C([-r,b]; E) \to C([-r,b]; E)$ defined by

$$N(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) h(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$

Clearly, fixed points of the operator N are mild solutions of the problem (4.1) - (4.2).

Let y be a possible solution of the problem (4.5)–(4.6). Given $t \leq b$, then from (H'1)-(H'4), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} |f(s,y_{\rho(s,y_s)})| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} |h(s,y_{\rho(s,y_s)})| \, ds. \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|y_{\rho(s,y_s)} \| \right] ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \|y_{\rho(s,y_s)}\| ds. \end{aligned}$$

It follows that

$$\begin{aligned} |y(t)| + \mathcal{L}^{\varphi} \|\varphi\| &\leq (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} q(s) \left[|y(s)| + \mathcal{L}^{\varphi} \|\varphi\| \right] ds \\ &+ \int_0^t (t-s)^{\alpha-1} \eta(s) (|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|) ds \end{aligned}$$

 Set

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha}}{\Gamma(\alpha + 1)}.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}(q^* + \eta^*)}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.7.2 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}(q^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $\|y\|_n \leq \mu(t)$, we have $\|y\|_n \leq \max\{\|\varphi\|, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $\|y\|_n \leq \max\{\|\varphi\|, \Lambda_b\} := \Theta_b$.

Set

$$Y = \{ y \in C([-r, b]; E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1\}.$$

Clearly, Y is a closed subset of C([-r, b]; E).

We shall show that $N: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H'1) and (H'3) - (H'4) for each $t \in [0, b]$

$$\begin{split} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s, y_{\rho(s,y_s)}) - f(s, \overline{y}_{\rho(s,y_s)}) \right| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| h(s, y_{\rho(s,y_s)}) - h(s, \overline{y}_{\rho(s,y_s)}) \right| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \, l_{b}(s) \, \|y_{\rho(s,y_s)} - \overline{y}_{\rho(s,y_s)}\| \, ds. \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \, \eta(s) \, \|y_{\rho(s,y_s)} - \overline{y}_{\rho(s,y_s)}\| \, ds. \end{split}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$|N(y)(t) - N(\overline{y})(t)| \leq \frac{\widehat{M}(l_b^* + \eta^*)}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} |y(s) - \overline{y}(s)| ds$$
$$\leq \frac{\widehat{M}(l_b^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)} ||y - \overline{y}||_n$$

Therefore,

$$\|N(y) - N(\overline{y})\|_n \le \frac{\widehat{M}(l_b^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)} \|y - \overline{y}\|_n$$

So, the operator N is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y), \lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator N has a unique fixed-point y^* which is the unique mild solution of the problem (4.5) - (4.6).

4.5 Fractional Neutral Problem with Finite Delay

Before stating and proving the main result, we give first the definition of a mild solution of the perturbed semilinear evolution problem (4.7) - (4.8).

Lemma 4.5.1. The system (4.7) - (4.8) is equivalent to the nonlinear integral equation

$$y(t) = [\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \tag{4.22}$$

$$+\frac{1}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] ds.$$

In other words, every solution of the integral equation (4.22) is also solution of the system (4.7) - (4.8) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (4.7) - (4.8), and using some classical results from fractional calculus to get (4.22).

Definition 4.5.1. We say that the function $y : [-r, b] \to E$ is a mild solution of (4.7) – (4.8) if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)[f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] ds$$
(4.23)

We consider the hypotheses (H_{φ}) , (H'0) - (H'4) and we need the following assumptions (H'5) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} \left(|s - \overline{s}| + \|\varphi - \overline{\varphi}\|\right)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

(H'6) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H'7) There exists a constant $0 < L < \frac{1}{\overline{M}_0}$ such that

 $|A(t) \ g(t,\varphi)| \le L \ (\|\varphi\|+1) \text{ for all } t \in J \text{ and } \varphi \in C(H;E).$

Theorem 4.5.1. Suppose that the hypotheses (H_{ϕ}) , (H'0) - (H'7) are satisfied and moreover

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M}(L_b^\star + \eta^\star)b^\alpha}{\Gamma(\alpha + 1)}\right] < 1$$
(4.24)

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $\eta^* = \sup \eta(s)$. Then the problem (4.7) - (4.8) has a unique mild solution on [-r, b].

Proof. Transform as below the neutral problem (4.7) - (4.8) into a fixed point problem by considering the operator $\widetilde{N} : C([-r, b]; E) \to C([-r, b]; E)$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ [\varphi(0) - g(0,\varphi)] + g(t, y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ [f(s, y_{\rho(s,y_s)}) + h(s, y_{\rho(s,y_s)})] \ ds & \text{if } t \in J. \end{cases}$$

Clearly, the fixed points of the operator \widetilde{N} are mild solutions of the problem (4.7) – (4.8).

Let y be a possible solution of the problem (4.7) – (4.8). Given $t \leq b$. Using (H'1) - (H'2), (H'6) - (H'7), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq |g(t, y_{\rho(t, y_t)})| + |U(t, 0)[\varphi(0) - g(0, \varphi)]| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) f(s, y_{\rho(s, y_s)}) ds \right| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) h(s, y_{\rho(s, y_s)}) ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t, y_{\rho(t, y_t)})| + ||U(t, 0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0, \varphi)| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s) ||y_{\rho(s, y_s)}|| \right] ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) ||y_{\rho(s, y_s)}|| ds. \end{aligned}$$

Since $||y_{\rho(t,y_t)}|| \le |y(t)| + \mathcal{L}^{\varphi}||\varphi||$, we obtain

$$\begin{aligned} |y(t)| &\leq \overline{M}_0 L\left(|y(t)| + \mathcal{L}^{\varphi} \|\varphi\|\right) + \widehat{M} \|\varphi\| (1 + \overline{M}_0 L) + \frac{\widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) \ \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|\right) \ ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) \ \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|\right) \ ds \end{aligned}$$

Then

$$\begin{aligned} (1 - \overline{M}_0 L)|y(t)| &\leq \left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right] \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M}(q^* + \eta^*)}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|\right) ds \end{aligned}$$

 Set

$$\delta_b := \mathcal{L}^{\varphi} \|\varphi\| + \frac{\left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right]}{(1 - \overline{M}_0 L)} \|\varphi\| + \frac{\widehat{M} p^* b^{\alpha}}{(1 - \overline{M}_0 L) \Gamma(\alpha + 1)}$$

Thus

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \leq \delta_b + \frac{\widehat{M}(q^* + \eta^*)}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\| \right) ds$$

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}(q^* + \eta^*)}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}(q^* + \eta^*)b^{\alpha}}{(1 - \overline{M}_0 L)\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||y|| \le \mu(t)$, we have $||y|| \le \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Now, we shall show that $\widetilde{N}: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus for each $t \in [0, b]$

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq |g(t, y_{\rho(t, y_t)}) - g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \left| f(s, y_{\rho(s, y_s)}) - f(s, \overline{y}_{\rho(s, y_s)}) \right| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \left| h(s, y_{\rho(s, y_s)}) - h(s, \overline{y}_{\rho(s, y_s)}) \right| \, ds \\ &\leq \|A^{-1}(t)\| |A(t)g(t, y_{\rho(t, y_t)}) - A(t)g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \, l_b(s) \, \|y_{\rho(s, y_s)} - \overline{y}_{\rho(s, y_s)}\| \, ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \, \eta(s) \, \|y_{\rho(s, y_s)} - \overline{y}_{\rho(s, y_s)}\| \, ds \end{split}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{split} |\widetilde{N}(y)(t) - \widetilde{N}(\overline{y})(t)| &\leq \overline{M}_0 L_\star \|y - \overline{y}\| + \frac{\widehat{M}(L_b^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)} \|y - \overline{y}\| \\ &\leq \left[\overline{M}_0 L_\star + \frac{\widehat{M}(L_b^* + \eta^*)b^{\alpha}}{\Gamma(\alpha + 1)}\right] \|y - \overline{y}\| \end{split}$$

Therefore,

$$\|\widetilde{N}(y) - \widetilde{N}(\overline{y})\|_{n} \leq \left[\overline{M}_{0}L_{\star} + \frac{\widehat{M}(L_{b}^{*} + \eta^{*})b^{\alpha}}{\Gamma(\alpha + 1)}\right]\|y - \overline{y}\|_{T}$$

So, for an appropriate choice of $\overline{M}_0L_\star,L_b^*,\eta^*$ and b^α such that

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M}(L_b^* + \eta^*)b^\alpha}{\Gamma(\alpha + 1)}\right] < 1,$$

the operator \widetilde{N} is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda \widetilde{N}(y)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. A consequence of the nonlinear alternative of Frigon and Granas shows that (C1) holds. We deduce that the operator \widetilde{N} has a unique fixed-point y^* which is the unique mild solution of the problem (4.7) - (4.8).

4.6 Fractional Partial Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of a mild solution of the perturbed semilinear evolution problem (4.9) - (4.10).

Lemma 4.6.1. The system (4.9) - (4.10) is equivalent to the nonlinear integral equation

$$y(t) = \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \tag{4.25}$$

$$+\frac{1}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] ds.$$

In other words, every solution of the integral equation (4.25) is also solution of the system (4.9) - (4.10) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (4.9) - (4.10), and using some classical results from fractional calculus to get (4.25).

Definition 4.6.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (4.9) – (4.10) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] \ ds \quad \text{for each } t \in J.$$
(4.26)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}.$$

We always assume that $\rho: J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter

(H'0) U(t,s) is compact for t-s > 0.

(H'01) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H'02) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H'03) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in \mathcal{B}$ with $||u|| \leq R$ and $||v|| \leq R$.

(H'04) There exists a function $\eta \in L^1(J, \mathbb{R}_+)$ such that

$$|h(t,u) - h(t,v)| \le \eta(t) \quad ||u - v||_{\mathcal{B}} \quad a.e. \ t \in J \quad et \quad \forall u, v \in \mathcal{B}.$$

Consider the following space

$$\Omega = \left\{ y : (-\infty, b] \to E : y|_{(-\infty, 0]} \in B \text{ and } y|_{[0, b]} \text{ is continuous } \right\},$$

Theorem 4.6.1. Assume that the hypotheses (H_{ϕ}) , (H'0) - (H'04) hold and moreover for all $n \in \mathbb{N}$,

$$\frac{K_b \widehat{M} \eta^* b^{\alpha}}{\Gamma(\alpha+1)} < 1 \tag{4.27}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $\eta^* = \sup \eta(s)$. Then the problem (4.9) - (4.10) has a unique mild solution on $(-\infty, b]$.

Proof. We transform the problem (4.9) - (4.10) into a fixed-point problem. Consider the operator $N : \Omega \to \Omega$ defined by

$$N(y)(t) = \begin{cases} \phi(t), & \text{if } t \in \mathbb{R}^{-}; \\ U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \ U(t,s) f(s, y_{\rho(s,y_s)}) \ ds, \\ + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \ U(t,s) h(s, y_{\rho(s,y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$

Clearly, fixed points of the operator N are mild solutions of the problem (4.9) - (4.10).

For $\phi \in \mathcal{B}$, we will define the function $x(.): (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \in (-\infty, 0]; \\ U(t, 0) \ \phi(0), & \text{if } t \in [0, b]. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (4.26) if and only if z satisfies $z_0 = 0$ and

$$z(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds$$

Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operators $F, G: \Omega^0 \to \Omega^0$ by

$$F(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \, ds.$$

and

$$G(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)h(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds$$

Obviously the operator N has a fixed point is equivalent to F + G has one, so it turns to prove that F + G has a fixed point.

First, show that F is continuous and compact.

Step 1: *F* continuous. Let $(z_n)_{n \in \mathbb{N}}$ a sequence in Ω^0 such that $z_n \to z$ in Ω^0 . By the hypothesis (H1), we have

$$|F(z_n)(t) - F(z)(t)| \leq \frac{1}{\Gamma(\alpha)} \int_0^t |(t-s)^{\alpha-1}| ||U(t,s)||_{B(E)}$$

 $\times |f(s, z_{n\rho(s, z_{ns}+x_s)} + x_{\rho(s, z_{ns}+x_s)}) - f(s, z_{\rho(s, z_s+x_s)} + x_{\rho(s, z_s+x_s)})| ds$
 $\leq \frac{\widehat{M}b^{\alpha} ||f(., z_{n\rho(.,.)} + x_{\rho(.,.)}) - f(., z_{\rho(.,.)} + x_{\rho(.,.)})||_{\infty}}{\Gamma(\alpha+1)}.$

$$\begin{aligned} |F(z)(t) - F(\overline{z})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} |f(s, z_{n\rho(s, z_{ns}+x_s)} + x_{\rho(s, z_{ns}+x_s)})| \\ &- f(s, z_{\rho(s, z_s+x_s)} + x_{\rho(s, z_s+x_s)})| ds \\ &\leq \frac{\widehat{M}b^{\alpha} ||f(., z_{n\rho(.,.)} + x_{\rho(.,.)}) - f(., z_{\rho(.,.)} + x_{\rho(.,.)})||_{\infty}}{\Gamma(\alpha+1)}. \end{aligned}$$

Since f is continuous, by dominated convergence theorem of Lebesgue, we get

$$|F(z_n)(t) - F(z)(t)| \longrightarrow 0 \text{ when } n \longrightarrow +\infty$$

So F is continuous.

Step 2 : Show that F transforms any bounded of Ω^0 in a bounded set. For each d > 0, there exists a positive constant ξ such that for all $z \in B_d = \{z \in \Omega^0 : ||z||_n \leq d\}$ we get $||F(z)||_n \leq \xi$. Soit $z \in B_d$, from assumption (H'01) and (H'02), we have for each $t \in [0, b]$

$$\begin{aligned} |F(z)(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} |f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| \ ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t p(s) \left[p(s) + q(s) ||z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}||_{\mathcal{B}} \right] \ ds \end{aligned}$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, b]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_b|x(s)| + (M_b + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\|U(s,0)\|_{B(E)}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\widehat{M}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_b |z(s)| + K_b \widehat{M}H \|\phi\|_{\mathcal{B}} + (M_b + \mathcal{L}^{\phi}) \|\phi\|_{\mathcal{B}} \\ &\leq K_b |z(s)| + (M_b + \mathcal{L}^{\phi} + K_b \widehat{M}H) \|\phi\|_{\mathcal{B}} \end{aligned}$$

Set $c_b := (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}}$ and $\delta_b := K_b d + c_b$. Then

$$||z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}||_{\mathcal{B}} \le K_b|z(s)| + c_b \le \delta_b.$$
(4.28)

For each $t \in [0, b]$, it follows that

$$|F(z)(t)| \leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)|+c_b) ds$$

$$\leq \frac{\widehat{M}b^{\alpha}(p^*+q^*\delta_b)}{\Gamma(\alpha+1)}$$

$$|F(z)(t)| \leq \frac{\widehat{M}b^{\alpha}(p^*+q^*\delta_b)}{\Gamma(\alpha+1)} := \varrho.$$

So there is a positive constant ρ such that $||F(z)||_n \leq \rho$. Then $F(B_d) \subset B_{\rho}$.

Step 3 : F maps bounded sets into equi-continuous sets of Ω^0 . We consider B_d as in Step 2 and we show that $F(B_d)$ is equi-continuous. Let $\tau_1, \tau_2 \in J$ with $\tau_1 < \tau_2$ and $z \in B_d$.

$$\begin{aligned} |F(z)(\tau_{2}) - F(z)(\tau_{1})| &\leq \frac{1}{\Gamma(\alpha)} \int_{0}^{\tau_{1}} |(\tau_{2} - s)^{\alpha - 1} - (\tau_{1} - s)^{\alpha - 1}| \|U(\tau_{2}, s) - U(\tau_{1}, s)\|_{B(E)} \\ &\quad |f(s, z_{\rho(s, z_{s} + x_{s})} + x_{\rho(s, z_{s} + x_{s})})| \\ &\quad + \frac{1}{\Gamma(\alpha)} \int_{\tau_{1}}^{\tau_{2}} |(\tau_{2} - s)^{\alpha - 1}| \|U(\tau_{2}, s)\|_{B(E)} |f(s, z_{\rho(s, z_{s} + x_{s})} + x_{\rho(s, z_{s} + x_{s})})| ds. \end{aligned}$$

Then by (5.41) we get

$$\begin{aligned} |F(z)(\tau_2) - F(z)(\tau_1)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^{\tau_1} |(\tau_2 - s)^{\alpha - 1} - (\tau_1 - s)^{\alpha - 1}| \|U(\tau_2, s) - U(\tau_1, s)\|_{B(E)} \\ &\quad [p(s) + q(s)\delta_b] ds. \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_{\tau_1}^{\tau_2} |(\tau_2 - s)^{\alpha - 1}| [p(s) + q(s)\delta_b] ds. \end{aligned}$$

Noting that $|F(z)(\tau_2) - F(z)(\tau_1)| \longrightarrow 0$ tends to zero as $\tau_2 - \tau_1 \longrightarrow 0$ independently of $z \in B_d$. The right-hand of the above inequality tends to zero as $\tau_2 - \tau_1 \longrightarrow 0$, since U(t, s) is a strongly continuous operator and the compactness of U(t, s) for t > s, implies the continuity in the uniform operator topology (voir [7, 90]). As a consequence of Steps 1 to 3 together with the Arzelá-Ascoli theorem it suffices to show that the operator F maps B_d into a precompact set in E.

Let $t \in J$ be fixed and let ε be a real number satisfying $0 < \varepsilon < t$. For $z \in B_d$ we define

$$F_{\varepsilon}(z)(t) = \frac{1}{\Gamma(\alpha)} U(t, t-\varepsilon) \int_0^{t-\varepsilon} (t-s)^{\alpha-1} U(t-\varepsilon, s) C u_{z+x}(s) \, ds.$$

Since U(t,s) is a compact operator, the set $Z_{\varepsilon}(t) = \{F_{\varepsilon}(z)(t) : z \in B_d\}$ is pre-compact in E for every ε , $0 < \varepsilon < t$. Moreover, using the definition of w, we get

$$|F(z)(t) - F_{\varepsilon}(z)(t)| \le \frac{1}{\Gamma(\alpha)} \int_{t-\varepsilon}^{t} (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} |f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})| ds$$

Therefore the set $Z(t) = \{F(z)(t) : z \in B_d\}$ is totally bounded. Hence the set $\{F(z)(t) : z \in B_d\}$. So we deduce from Steps 1, 2 and 3 that F is a compact operator.

Step 4 : G is a contraction. Let $z, \overline{z} \in \Omega^0$. By the hypotheses (H'01) and (H'04), we get for each $t \in [0, b]$ and $n \in \mathbb{N}$

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \\ &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \\ &-h(s, \overline{z}_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \|ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)} - x_{\rho(s, z_s + x_s)} \|_{\mathcal{B}} ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \|z_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)} \|_{\mathcal{B}} ds. \end{aligned}$$

Use the inequality (4.28), for get

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} K_b \eta(s) |z(s) - \overline{z}(s)| \ ds \\ &\leq \frac{\widehat{M} K_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)|| \end{aligned}$$

Therefore

$$\|G(z) - G(\overline{z})\|_n \le \frac{\widehat{M}K_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)} \|z(t) - \overline{z}(t)\|.$$

Then the operator G is a contraction.

Step 5: For applying Theorem (1.10.2), we must check (Av2): i.e. it remains to show that the set

$$\Sigma = \left\{ z \in \Omega^0 : z = \lambda F(z) + \lambda G\left(\frac{z}{\lambda}\right) \quad \text{for some } \lambda \in]0, 1[\right\}.$$

is bounded.

Let $z \in \Sigma$. By (H'01) - (H'02) and (H'04), we have for each $t \in [0, b]$

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \right| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| h\left(s, \frac{z_{\rho(s,\frac{z_s}{\lambda}+x_s)}}{\lambda}\right) - h(s,0) + h(s,0) \right| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t \left[p(s) + q(s) \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)} \|_{\mathcal{B}} \right] \, ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t \left(t-s\right)^{\alpha-1} \eta(s) \left\| \frac{z_{\rho(s,\frac{z_s}{\lambda}+x_s)}}{\lambda} \right\|_{\mathcal{B}} \, ds + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t \left(t-s\right)^{\alpha-1} |h(s,0)| \, ds \end{aligned}$$

Use Proposition (4.2.1) and inequality (4.15) to get

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} q(s) \left(\frac{K_b}{\lambda} |z(s)| + c_b\right) ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \left(\frac{K_b}{\lambda} |z(s)| + c_b\right) ds + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |h(s,0)| ds. \end{aligned}$$

We consider the function $u(t) := \sup_{\theta \in [0,t]} |z(\theta)|$ and use the fact that $0 < \lambda < 1$

$$\frac{K_b}{\lambda}u(t) + c_b \leq c_b + \frac{K_b\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)| + c_b\right) ds \\
+ \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}\eta(s)\left(\frac{K_b}{\lambda}|z(s)| + c_b\right) ds + \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1}|h(s,0)| ds$$

Set $\sigma_b := c_b + \frac{K_b \widehat{M} p^* b^{\alpha}}{\Gamma(\alpha+1)} + \frac{K_b \widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} |h(s,0)| ds$. Then, we have

$$\frac{K_b}{\lambda}u(t) + c_b \leq \sigma_b + \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)| + c_b\right) ds \\
+ \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}\eta(s)\left(\frac{K_b}{\lambda}|z(s)| + c_b\right) ds.$$

We consider the function μ defined by

$$\mu(t) = \sup \{ \frac{K_b}{\lambda} u(s) + c_b : 0 \le s \le t \}, \ 0 \le t \le b$$

Let $t^* \in [0, t]$ such that $\mu(t) = \frac{K_b}{\lambda} u(t^*) + c_b$. From the previous inequality, we have

$$\mu(t) \le \sigma_b + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds + \frac{K_b \widehat{M} \eta^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds.$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = ||\phi||$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M}(q^* + \eta^*) b^{\alpha}}{\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||z|| \leq \mu(t)$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Since $||z|| \leq \mu(t)$, we have $||z|| \leq \Lambda_b$. This shows that the set Σ is bounded. Then the statement (Av2) in Theorem 1.10.5 does not hold. The nonlinear alternative of Avramescu implies that (Av1) is satisfied, we deduce that the operator F + G has a fixed point z^* . Then $y^*(t) = z^*(t) + x(t), t \in]-\infty, b]$ is a fixed point of the operator N which is a mild solution of the problem (4.9) - (4.10).

4.7 Fractional Neutral Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of a mild solution of the perturbed semilinear evolution problem (4.11) - (4.12).

Lemma 4.7.1. The system (4.11) - (4.12) is equivalent to the nonlinear integral equation

$$y(t) = [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s)y(s) \ ds \tag{4.29}$$

$$+\frac{1}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} [f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)})] \, ds.$$

In other words, every solution of the integral equation (4.29) is also solution of the system (4.11) - (4.12) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (4.11) - (4.12), and using some classical results from fractional calculus to get (4.29).

Definition 4.7.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (4.11) – (4.12) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)})$$

$$+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)[f(s, y_{\rho(s,y_s)}) + h(s, y_{\rho(s,y_s)})] ds$$
(4.30)

We consider the hypotheses (H_{ϕ}) , (H'0) - (H'04) and we need the following assumptions (H'05) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\phi) - A(\overline{s})g(\overline{s},\overline{\phi})| \le L_{\star} (|s-\overline{s}| + \|\phi - \overline{\phi}\|)$$

for all $\phi, \overline{\phi} \in \mathcal{B}$.

(H'06) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H'07) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

$$|A(t) g(t, \varphi)| \le L (||\phi|| + 1)$$
 for all $t \in J$ and $\phi \in \mathcal{B}$.

Theorem 4.7.1. Suppose that the hypotheses (H_{ϕ}) , (H'0) - (H'02) and (H'04) - (H'07) are satisfied and moreover

$$\left[\overline{M}_{0}L_{\star}K_{b} + \frac{\widehat{M}k_{b}\eta^{*}b^{\alpha}}{\Gamma(\alpha+1)}\right] < 1$$
(4.31)

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $\eta^* = \sup \eta(s)$. Then the problem (4.11) – (4.12) has a unique mild solution on $(-\infty, b]$.

Proof. Consider the operator $\widetilde{N}: \Omega \to \Omega$ defined by

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \in \mathbb{R}^-; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ \left[f(s,y_{\rho(s,y_s)}) + h(s,y_{\rho(s,y_s)}) \right] \ ds & \text{if } t \in J. \end{cases}$$

Then, fixed points of the operator \widetilde{N} are mild solutions of the problem (4.11) - (4.12).

For $\phi \in \mathcal{B}$, we consider the function $x(.): (-\infty, b] \to E$ defined as below by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (4.30) if and only if z satisfies $z_0 = 0$ and

$$\begin{aligned} z(t) &= g(t, z_{\rho(s, z_t + x_t)} + x_{\rho(s, z_t + x_t)}) - U(t, 0)g(0, \phi) \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds \end{aligned}$$

Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operators $F, \widetilde{G}: \Omega^0 \to \Omega^0$ by

$$F(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \ ds$$

And

$$\widetilde{G}(z)(t) = g(t, z_{\rho(s, z_t + x_t)} + x_{\rho(s, z_t + x_t)}) - U(t, 0)g(0, \phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) ds.$$

Obviously the operator \widetilde{N} has a fixed point is equivalent to $F + \widetilde{G}$ has one, so it turns to prove that $F + \widetilde{G}$ has a fixed point.

We have shown that the operator F is continuous and compact as in Section 3.6 Remains to show that the operator \widetilde{G} is a contraction.

Let $z, \overline{z} \in \Omega^0$. By (H'01) and (H'04) we have for each $t \in [0, b]$

$$\begin{split} |\widetilde{G}(z)(t) - \widetilde{G}(\overline{z})(t)| &\leq |g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)}) - g(t, \overline{z}_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} |h(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) - \\ &\quad h(s, \overline{z}_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})| ds \\ &\leq \|A^{-1}(t)\| \|A(t)g(t, z_{\rho(t, y_t)} + x_{\rho(t, y_t)}) - A(t)g(t, \overline{z}_{\rho(t, y_t)} + x_{\rho(t, y_t)})\| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) \|z_{\rho(s, z_s + x_s)} - \overline{z}_{\rho(s, z_s + x_s)}\|_{\mathcal{B}} ds. \end{split}$$

4.7 Frac. Neutral Perturbed Evo. Equa. with Infinite S.D.D.

Use the inequality (4.15), and (H'05) - (H'06), we get

$$\begin{aligned} |\widetilde{G}(z)(t) - \widetilde{G}(\overline{z})(t)| &\leq \overline{M}_0 L_{\star} K_b ||z(t) - \overline{z}(t)|| + \frac{\widehat{M} K_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)} \int_0^t (t-s)^{\alpha-1} |z(s) - \overline{z}(s)| \ ds \\ &\leq \overline{M}_0 L_{\star} K_b ||z(t) - \overline{z}(t)|| + \frac{\widehat{M} K_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)|| \\ &\leq \left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} k_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)}\right] ||z(t) - \overline{z}(t)||. \end{aligned}$$

Therefore

$$\|\widetilde{G}(z) - \widetilde{G}(\overline{z})\|_{n} \leq \left[\overline{M}_{0}L_{\star}K_{b} + \frac{\widehat{M}k_{b}\eta^{*}b^{\alpha}}{\Gamma(\alpha+1)}\right]\|z(t) - \overline{z}(t)\|$$

So, for an appropriate choice of $\overline{M}_0 L_{\star}$, $\eta^* K_b$ and b^{α} such that

$$\left[\overline{M}_0 L_{\star} K_b + \frac{\widehat{M} k_b \eta^* b^{\alpha}}{\Gamma(\alpha+1)}\right] < 1,$$

operator \widetilde{G} is a contraction.

For applying Theorem (1.10.2), we must check (Av2) i.e. it remains to show that the set

$$\widetilde{\Sigma} = \left\{ z \in B^0_{+\infty} : z = \lambda F(z) + \lambda \widetilde{G}\left(\frac{z}{\lambda}\right) \text{ pour } 0 < \lambda < 1 \right\}$$

is bounded.

Let $z \in \widetilde{\Sigma}$. By (H'01) - (H'02), we have for each $t \in [0, b]$

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})| + |U(t,0)g(0,\phi)| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \right| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| h\left(s, \frac{z_{\rho(s,z_s+x_s)}}{\lambda}\right) - h(s,0) + h(s,0) \right| \\ &\leq \|A^{-1}(t)\| \, |A(t)g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + \|U(t,0)\|_{B(E)} \|A^{-1}(0)\| \, |A(0)g(0,\phi)| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,y_s)} + x_{\rho(s,y_s)} \| \right] \, ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \, \left\| \frac{z_{\rho(s,z_s+x_s)}}{\lambda} + x_{\rho(s,z_s+x_s)} \right\|_{\mathcal{B}} \, ds + \widehat{M} \int_0^t |h(s,0)| \, ds.ds \end{aligned}$$

Using assumptions (H'04), (H'06) and (H'07)

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq \overline{M}_0 L(\|z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}\|_{\mathcal{B}} + 1) + \widehat{MM}_0 L(\|\phi\|_{\mathcal{B}} + 1) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,y_s)} + x_{\rho(s,y_s)}\| \right] ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \left\| \frac{z_{\rho(s,\frac{z_s}{\lambda}+x_s)}}{\lambda} + x_{\rho(s,\frac{z_s}{\lambda}+x_s)} \right\|_{\mathcal{B}} ds + \widehat{M} \int_0^t |h(s,0)| ds. \end{aligned}$$

Use Proposition (4.2.1) and inequality (4.15) for get

$$\begin{aligned} \left\| \frac{z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}}{\lambda} + x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)} \right\|_{\mathcal{B}} &\leq \frac{1}{\lambda} \left\| z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)} \right\|_{\mathcal{B}} + \left\| x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)} \right\|_{\mathcal{B}} \\ &\leq \frac{K_{b}}{\lambda} |z(s)| + \frac{M_{b}+\mathcal{L}^{\phi}}{\lambda} \| z_{0} \|_{\mathcal{B}} + K_{b} |x(s)| + \left(M_{b}+\mathcal{L}^{\phi}\right) \| x_{0} \|_{\mathcal{B}} \\ &\leq \frac{K_{b}}{\lambda} |z(s)| + K_{b} \| U(s,0) \|_{\mathcal{B}(E)} |\phi(0)| + \left(M_{b}+\mathcal{L}^{\phi}\right) \| \phi \|_{\mathcal{B}} \\ &\leq \frac{K_{b}}{\lambda} |z(s)| + K_{b} \widehat{M}H \| \phi \|_{\mathcal{B}} + \left(M_{b}+\mathcal{L}^{\phi}\right) \| \phi \|_{\mathcal{B}} \\ &\leq \frac{K_{b}}{\lambda} |z(s)| + \left(M_{b}+\mathcal{L}^{\phi}+K_{b} \widehat{M}H\right) \| \phi \|_{\mathcal{B}}. \end{aligned}$$

Hence

$$\left\|\frac{z_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}}{\lambda}+x_{\rho\left(s,\frac{z_{s}}{\lambda}+x_{s}\right)}\right\|_{\mathcal{B}} \leq \frac{K_{b}}{\lambda}|z(s)|+c_{b}.$$
(4.32)

Use the function $u(\cdot)$ to get

$$\frac{u(t)}{\lambda} \leq \overline{M}_0 L(K_b | u(t) | + c_b) + \overline{M}_0 L\left(1 + \widehat{M}(\|\phi\| + 1)\right) \\
+ \frac{\widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} |h(s, 0)| \, ds + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) \left(K_b u(s) + c_b\right) \, ds \\
+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) \left(\frac{K_b}{\lambda} u(s) + c_b\right) \, ds.$$

Then

$$\frac{u(t)}{\lambda} \leq \overline{M}_0 L(\frac{K_b}{\lambda}u(t) + c_b) + \overline{M}_0 L\left(1 + \widehat{M}(\|\phi\| + 1)\right) + \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha + 1)} \\
+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t - s)^{\alpha - 1} |h(s, 0)| \, ds + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) \left(\frac{K_b}{\lambda}u(s) + c_b\right) \, ds \\
+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) \left(\frac{K_b}{\lambda}u(s) + c_b\right) \, ds.$$

Set

$$\zeta_b := \overline{M}_0 L c_b + \overline{M}_0 L \left(1 + \widehat{M}(\|\phi\| + 1) \right) + \frac{\widehat{M} p^* b^\alpha}{\Gamma(\alpha + 1)} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t - s)^{\alpha - 1} |h(s, 0)| \, ds.$$

So

$$\frac{K_b}{\lambda} \left(1 - \overline{M}_0 L K_b \right) u(t) \leq K_b \zeta_b + \frac{K_b \widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} q(s) \left(\frac{K_b}{\lambda} u(s) + c_b \right) ds \\
+ \frac{K_b \widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \eta(s) \left(\frac{K_b}{\lambda} u(s) + c_b \right) ds.$$

Set
$$\xi_b := c_b + \frac{K_b \zeta_b}{(1 - \overline{M}_0 L K_b)}$$
. Then

$$\frac{K_b}{\lambda} u(t) + c_b \leq \xi_b + \frac{K_b \widehat{M}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) \left(\frac{K_b}{\lambda} u(s) + c_b\right) ds$$

$$+ \frac{K_b \widehat{M}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \eta(s) \left(\frac{K_b}{\lambda} u(s) + c_b\right) ds.$$

We consider the function μ defined by

$$\mu(t) = \sup \left\{ \frac{K_b}{\lambda} u(s) + c_b : 0 \le s \le t \right\}, \quad 0 \le t \le b$$

Let $t^* \in [0, t]$ be such that $\mu(t) = \frac{K_b}{\lambda} u(t^*) + c_b$. By the previous inequality, we have for $t \in [0, b]$

$$\mu(t) \leq \xi_b + \frac{K_b \widehat{M}(q^* + \eta^*)}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = ||\phi||$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M}(q^* + \eta^*) b^{\alpha}}{\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||z|| \leq \mu(t)$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Since $||z|| \leq \mu(t)$, we have $||z|| \leq \Lambda_b$. This shows that the set Σ is bounded. Then the statement (Av2) in Theorem 1.10.5 does not hold. The nonlinear alternative of Avramescu implies that (Av1) is satisfied, we deduce that the operator F + G has a fixed point z^* . Then $y^*(t) = z^*(t) + x(t), t \in]-\infty, b]$ is a fixed point of the operator N which is a mild solution of the problem (4.11) - (4.12).

4.8 Examples

To illustrate the previous results, we give in this section six examples.

Example 1. Consider the partial differential equation

$$\begin{pmatrix}
\frac{\partial u}{\partial t}(t,\xi) = \frac{\partial^2 u(t,\xi)}{\partial \xi^2} + a_0(t,\xi)u(t,\xi) \\
+ \int_{-\infty}^0 a_1(s-t)u \left[s - \rho_1(t)\rho_2\left(\int_0^{\pi} a_2(\theta)|u(t,\theta)|^2d\theta\right),\xi\right] ds \\
+ \int_{-\infty}^0 a_3(s-t)u \left[s - \rho_1(t)\rho_2\left(\int_0^{\pi} a_2(\theta)|u(t,\theta)|^2d\theta\right),\xi\right] ds, \\
t \ge 0, \ \xi \in [0,\pi], \\
u(t,0) = u(t,\pi) = 0, \qquad t \ge 0, \\
u(\theta,\xi) = u_0(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi].
\end{cases}$$
(4.33)

where $a_0(t,\xi)$ is a continuous function and is uniformly Hölder continuous in $t; a_1, a_3 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0,\pi] \to \mathbb{R}, \rho_i : [0,+\infty[\to \mathbb{R} \text{ are continuous functions } i=1,2.$

To study this system, we consider the space $E = L^2([0, \pi], \mathbb{R})$ and the operator $A : D(A) \subset E \to E$ given by Aw = w'' with

$$D(A) := \{ w \in E : w'' \in E, w(0) = w(\pi) = 0 \}$$

It is well known that A is the infinitesimal generator of an analytic semigroup $\{T(t)\}_{t\geq 0}$ on E, with compact resolvent. On the domain D(A), we define the operators $A(t) : D(A) \subset E \to E$ by

$$A(t)x(\xi) = Ax(\xi) + a_0(t,\xi)x(\xi).$$

By assuming that $a_0(.,.)$ is continuous and that $a_0(t,\xi) \leq -\delta_0$ ($\delta_0 > 0$) for every $t \in \mathbb{R}, \xi \in [0,\pi]$, and specific case $\alpha = 1$ it follows that the system

$$u'(t) = A(t)u(t) \quad t \ge s; \quad u(s) = x \in E,$$

has an associated evolution family given by

$$U(t,s)x(\xi) = \left[T(t-s)exp\left(\int_s^t a_0(\tau,\xi)d\tau\right)x\right](\xi).$$

From this expression, it follows that U(t,s) is a compact linear operator and that

$$||U(t,s)|| \le e^{-(1+\delta_0)(t-s)} \text{ for every } (t,s) \in \mathbb{R}^+ \times \mathbb{R}^+ \quad ; \quad s \le t.$$

Theorem 4.8.1. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}, i = 1, 2$, are continuous and the functions $a_1, a_3 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (4.33) on $] - \infty, +\infty[$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$h(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (4.33) into the abstract system (4.1) - (4.2). Moreover, the function f and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.2.1. From Remark 1.7.2, we have the following result

Corollary 4.8.1. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (4.33) on $] - \infty, +\infty[$.

Example 2. Consider the partial differential equation

$$\begin{cases} \frac{\partial}{\partial t} \left[u(t,\xi) - \int_{-\infty}^{0} a_{4}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \\ + \int_{-\infty}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\ t \ge 0, \ \xi \in [0,\pi], \end{cases}$$
(4.34)
$$v(t,0) = v(t,\pi) = 0, \qquad t \ge 0, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi]. \end{cases}$$

 $a_4: \mathbb{R}_- \to \mathbb{R}$ is a continuous function.

Theorem 4.8.2. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $d : [0, \pi] \to E, \ \rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}$ for $i = 1, 2, \ a_1, a_3, a_4 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (4.34) on $] - \infty, +\infty[$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$
$$h(t,\psi)(\xi) = \int_{-r}^{0} a_4(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (4.34) into the abstract system (4.3) - (4.4). Moreover, the function f, g and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.3.1.

From Remark 1.7.2, we have the following result

Corollary 4.8.2. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (4.34) on $] - \infty, +\infty[$.

,

Example 3. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0}a_{1}(s-t)u \left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right] ds \\ + \int_{-r}^{0}a_{3}(s-t)u \left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right] ds, \\ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(4.35)

where $a_1, a_3 : [-r, 0] \to \mathbb{R}$ is a continuous function.

Theorem 4.8.3. Let $\varphi \in C(H, E)$. Assume that the condition (H_{φ}) holds, $\rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}$, i = 1, 2, are continuous and the functions $a_1, a_3 : [-r, 0] \rightarrow \mathbb{R}$ and $a_2 : [0, \pi] \rightarrow \mathbb{R}$ are continuous. Then there exists a mild solution of (4.35) on] - r, b].

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$h(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right)$$

are well defined functions, which permit to transform system (4.35) into the abstract system (4.1) - (4.2). Moreover, the function f and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.4.1. From Remark 1.7.1, we have the following result

Corollary 4.8.3. Let $\varphi \in C(H, E)$ be continuous and bounded. Then there exists a mild solution of (4.35) on] - r, b].

Example 4. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-r}^{0} a_{4}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \\ + \int_{-r}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\ (4.36) \\ & , \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi]. \end{cases}$$

where $a_4: [-r, 0] \to \mathbb{R}$ is a continuous function.

Theorem 4.8.4. Let $\varphi \in C(H, E)$. Assume that the condition (H_{φ}) holds, $d : [0, \pi] \to E$, $\rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}$ for $i = 1, 2, a_1, a_3, a_4 : [-r, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (4.36) on] - r, b].

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$
$$h(t,\psi)(\xi) = \int_{-r}^{0} a_4(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right)$$

are well defined functions, which permit to transform system (4.36) into the abstract system (4.3) - (4.4). Moreover, the function f, g and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.5.1.

,

From Remark 1.7.1, we have the following result

Corollary 4.8.4. Let $\varphi \in C(H, E)$ be continuous and bounded. Then there exists a mild solution of (4.36) on] - r, b].

Example 5. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0}a_{1}(s-t)u \left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right] ds \\ + \int_{-\infty}^{0}a_{3}(s-t)u \left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right] ds, \\ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(4.37)

where $a_1, a_3 : (-\infty, 0] \to \mathbb{R}$ are continuous functions.

Theorem 4.8.5. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}, i = 1, 2$, are continuous and the functions $a_1, a_3 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (4.37) on $] -\infty, b]$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds$$

$$h(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (4.37) into the abstract system (4.1) - (4.2). Moreover, the function f and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.6.1. From Remark 1.7.2, we have the following result

Corollary 4.8.5. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (4.37) on $] - \infty, b]$.

Example 6. Consider the partial differential equation

where $a_4: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 4.8.6. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $d : [0, \pi] \to E, \ \rho_i : \mathbb{R}_+ \longrightarrow \mathbb{R}$ for $i = 1, 2, \ a_1, a_3, a_4 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (4.38) on $] - \infty, b]$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$

$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$

$$h(t,\psi)(\xi) = \int_{-\infty}^{0} a_4(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (4.38) into the abstract system (4.3) - (4.4). Moreover, the function f, g and h is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 4.7.1.

From Remark 1.7.2, we have the following result

Corollary 4.8.6. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (4.38) on $] - \infty, b]$.

Chapter 5

Controllability Results for Evolution Equations with State-Dependent Delay

5.1 Introduction

In this chapter, we will prove the controllability of mild solutions for some classes of first order of partial functional and neutral functional differential evolution equations with infinite state-dependent delay in Fréchet spaces¹, then we will show the controllability of mild solutions for some classes of Caputo's fractional derivative order of partial functional and neutral functional differential evolution equations with finite and infinite state-dependent delay² by using the nonlinear alternative of Avramescu for the sum of contraction and completely continuous operators on Banach spaces.

The controllability of mild solutions on the real positif interval is demonstrated in section 5.2 for the following class of evolution equations with infinite state-dependent delay :

$$y'(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_t)}), \text{ a.e. } t \ge 0$$
(5.1)

$$y_0 = \phi \in \mathcal{B},\tag{5.2}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $f : \mathbb{R}^+ \times \mathcal{B} \to E, \rho : \mathbb{R}^+ \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions, the control function u(.) is given in $L^2(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|, C$ is a bounded linear operator from E into E and $\{A(t)\}_{0 \leq t < +\infty}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in\mathbb{R}^+\times\mathbb{R}^+}$ for $s \leq t$.

An extension of this problem is given in section 5.3, we consider the following class of neutral evolution equations with infinite state-dependent delay

$$\frac{d}{dt}[y(t) - g(t, y_{\rho(t, y_t)})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_t)}), \quad \text{a.e.} \quad t \ge 0$$
(5.3)

$$y_0 = \phi \in \mathcal{B},\tag{5.4}$$

¹[11] **D. Aoued** and S. Baghli-Bendimerad, Controllability of mild solutions for evolution equations with infinite state-dependent delay, *Europian Journal of Pure and Applied Mathematics*, **9** (4), 2016, 383-401.

²[14] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Controllability of mild solutions for fractional partial functional and neutral functional evolution equations with state-dependent delay, submitted.

where $A(\cdot)$, f, ϕ and C are as in problem (5.1) – (5.2) and $g: \mathbb{R}^+ \times \mathcal{B} \to E$ is a given function.

In section 5.4, we give the controllability of the unique mild solution of the following class of fractional evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \quad t \in J = [0, b]$$
(5.5)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0],$$
 (5.6)

where $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1]$, $f: J \times C(H; E) \to E, \rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions, the control function u(.) is given in $L^{2}(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|$, C is a bounded linear operator from E into Eand $\{A(t)\}_{0 \leq t \leq b}$ is a family of linear closed (not necessarily bounded) operators from E into Ethat generate an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $0 \leq s \leq t \leq b$.

An extension of this problem is given in section 5.5, we consider the following class of fractional neutral evolution equations with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(5.7)

$$y(t) = \varphi(t), \quad t \in H, \tag{5.8}$$

where $A(\cdot)$, f, u, C and φ are as in problem (5.5) – (5.6) and $g: J \times C(H; E) \to E$ is a given function.

In section 5.6, we investigate the controllability of mild solutions of the following class of fractional evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) = A(t)y(t) + Cu(t) + f(t, y_{\rho(t,y_{t})}), \quad \text{a.e.} \quad t \in J$$
(5.9)

$$y_0 = \phi \in \mathcal{B},\tag{5.10}$$

where ${}^{c}D_{t}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1]$, $f: J \times \mathcal{B} \to E$ and $\phi \in \mathcal{B}$ are given functions, the control function u(.) is given in $L^{2}(J; E)$, the Banach space of admissible control function with E is a real separable Banach space with the norm $|\cdot|$, Cis a bounded linear operator from E into E and $\{A(t)\}_{0 \leq t \leq b}$ is a family of linear closed (not necessarily bounded) operators from E into E that generate an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $0 \leq s \leq t \leq b$.

An extension of this problem is given in section 5.7, we consider the following class of fractional neutral evolution equations with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}[y(t) - g(t, y_{\rho(t, y_{t})})] = A(t)y(t) + Cu(t) + f(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(5.11)

$$y_0 = \phi \in \mathcal{B},\tag{5.12}$$

where $A(\cdot)$, f, u, C and ϕ are as in problem (5.11) – (5.12) and $g : J \times \mathcal{B} \to E$ is a given function. Finally, section 5.8 is devoted to examples illustrating the abstract theory considered in previous sections.

5.2 Partial Problem with Infinite Delay

Before stating and proving our first main result, we define firstly the corresponding mild solution of the semilinear evolution problem (5.1)-(5.2) then we define the concept of controllability for that problem and finally we expose the properties of state-dependent delay.

Definition 5.2.1. We say that the function $y : \mathbb{R} \to E$ is a mild solution of (5.1) - (5.2) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies for each $t \geq 0$ the following integral equation

$$y(t) = U(t,0)\phi(0) + \int_0^t U(t,s)Cu(s)ds + \int_0^t U(t,s) f(s,y_{\rho(s,y_s)})ds.$$
(5.13)

Definition 5.2.2. The evolution problem (5.1) - (5.2) is said to be controllable if for every initial function $\phi \in \mathcal{B}$, $y^* \in E$ and for some $n \in \mathbb{N}$, there is some control $u \in L^2([0, n]; E)$ such that the mild solution $y(\cdot)$ of (5.1) - (5.2) satisfies the terminal condition $y(n) = y^*$.

Set $\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \leq 0\}$. We always assume that $\rho : J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $L^{\phi} : \mathcal{R}(\rho^-) \to (0, +\infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq L^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

Lemma 5.2.1. ([70]) If $y: (-\infty, b] \to E$ is a function such that $y_0 = \phi$, then

$$\|y_s\|_{\mathcal{B}} \le (M_b + L^{\phi}) \|\phi\|_{\mathcal{B}} + K_b \sup\{|y(\theta)|; \theta \in [0, max\{0, s\}]\}, \ s \in \mathcal{R}(\rho^-) \cup J,$$

where $L^{\phi} = \sup_{t \in \mathcal{R}(\rho^{-})} L^{\phi}(t).$

Proposition 5.2.1. From (H_{ϕ}) , (A_1) and Lemma (5.2.1), for all $t \in [0, n]$ and $n \in \mathbb{N}$ we have

$$||y_{\rho(t,y_t)}||_{\mathcal{B}} \le K_n |y(t)| + (M_n + L^{\phi}) ||\phi||_{\mathcal{B}}.$$

We will need to introduce the following hypothesis which are assumed thereafter :

- (H0) U(t,s) is compact for t-s > 0.
- (H1) There exists a constant $\widehat{M} \geq 1$ such that $||U(t,s)||_{B(E)} \leq \widehat{M}$ for every $(t,s) \in \Delta$.
- (H2) There exists a function $p \in L^1_{loc}(J; \mathbb{R}_+)$ and a continuous nondecreasing function $\psi : \mathbb{R}_+ \to (0, +\infty)$ and such that

$$|f(t, u)| \le p(t) \ \psi(||u||_{\mathcal{B}})$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||_{\mathcal{B}}$$

for all $u, v \in \mathcal{B}$ with $||u||_{\mathcal{B}} \leq R$ and $||v||_{\mathcal{B}} \leq R$.

(H4) For each $n \in \mathbb{N}$, the linear operator $W: L^2([0,n]; E) \to E$ is defined by

$$Wu = \int_0^n U(n,s) Cu(s) ds$$

has a pseudo invertible operator \tilde{W}^{-1} which takes values in $L^2([0,n];E)/\ker W$ and there exists positive constants \widetilde{M} and \widetilde{M}_1 such that $||C|| \leq \widetilde{M}$ and $||\widetilde{W}^{-1}|| \leq \widetilde{M}_1$.

For the construction of \tilde{W}^{-1} see the paper of Carmichael *et al.* [38].

Consider the following space $B_{+\infty} = \{ y : \mathbb{R} \to E : y|_{[0,b]} \text{ continuous for } b > 0 \text{ and } y_0 \in \mathcal{B} \},\$ where $y|_{[0,b]}$ is the restriction of y to the real compact interval [0,b].

Let us fix $\tau > 1$. For every $n \in \mathbb{N}$, we define in $B_{+\infty}$ the semi-norms by :

$$||y||_n := \sup_{t \in [0,n]} e^{-\tau L_n^*(t)} |y(t)|$$

where $L_n^*(t) = \int_0^t \overline{l}_n(s) \, ds$, $\overline{l}_n(t) = K_n \widehat{M} l_n(t)$ and l_n is the function from (H3).

Then $B_{+\infty}$ is a Fréchet space with the family of semi-norms $\|\cdot\|_{n\in\mathbb{N}^+}$

Theorem 5.2.1. Assume that the hypotheses (H_{ϕ}) and (H0) - (H4) hold and moreover for each $n \in \mathbb{N}$, there exists a constant $M^n_{\star} > 0$ such that

$$\frac{M_{\star}^{n}}{\alpha_{n} + K_{n}\widehat{M}(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1) \ \psi(M_{\star}^{n}) \ \|p\|_{L^{1}}} > 1,$$
(5.14)

with $\alpha_n = K_n \widehat{M} \widetilde{M} \widetilde{M}_1 n |y^*| + \left[M_n + L^{\phi} + K_n \widehat{M} H\left(\widehat{M} \widetilde{M} \widetilde{M}_1 n + 1 \right) \right] \|\phi\|_{\mathcal{B}}$. Then the evolution problem (5.1) - (5.2) is controllable on \mathbb{R} .

Proof. We transform the problem (5.1) - (5.2) into a fixed-point problem. Consider the operator $N: B_{+\infty} \to B_{+\infty}$ defined by :

$$N(y)(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0) + \int_0^t U(t,s) \ C \ u_y(s) ds + \int_0^t U(t,s) \ f(s, y_{\rho(s,y_s)}) ds, & \text{if } t \in J. \end{cases}$$

Clearly, fixed points of the operator N are mild solutions of the problem (5.1) - (5.2).

Using assumption (H4), for arbitrary function $y(\cdot)$, we define the control

$$u_y(t) = \tilde{W}^{-1} \left[y^* - U(n,0) \ \phi(0) - \int_0^n U(n,\tau) \ f(\tau, y_{\rho(\tau,y_\tau)}) \ d\tau \right] (t)$$

Applying (H2), we get

$$|u_y(t)| \leq \widetilde{M}_1 \left[|y^*| + \widehat{M}H \|\phi\|_{\mathcal{B}} + \widehat{M} \int_0^n p(\tau) \ \psi(\|y_{\rho(\tau,y_\tau)}\|_{\mathcal{B}}) d\tau \right].$$
(5.15)

We shall show that using this control the operator N has a fixed point $y(\cdot)$. Then $y(\cdot)$ is a mild solution of the evolution system (2.1).
For $\phi \in \mathcal{B}$, we will define the function $x(.) : \mathbb{R} \to E$ by $x(t) = \phi(t)$ for $t \leq 0$ and $x(t) = U(t,0) \phi(0)$ for $t \in J$. Then $x_0 = \phi$. For each function $z \in B_{+\infty}$ with z(0) = 0, we denote by \overline{z} the function defined by $\overline{z}(t) = 0$ for $t \leq 0$ and $\overline{z}(t) = z(t)$ for $t \in J$.

If $y(\cdot)$ satisfies (5.13), we can decompose it as $y(t) = z(t) + x(t), t \ge 0$, which implies $y_t = z_t + x_t$, for every $t \in J$ and the function $z(\cdot)$ satisfies for $t \in J$

$$z(t) = \int_0^t U(t,s) \ C \ u_{z+x}(s)ds + \int_0^t \ U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)})ds.$$

Let $B^0_{+\infty} = \{z \in B_{+\infty} : z_0 = 0 \in \mathcal{B}\}$. For any $z \in B^0_{+\infty}$ we have $||z||_{+\infty} = \sup_{s \ge 0} |z(s)|$. Thus $(B^0_{+\infty}, ||\cdot||_{+\infty})$ is a Banach space. We define the operators $F, G : B^0_{+\infty} \to B^0_{+\infty}$ by $F(z)(t) = \int_0^t U(t,s) \ C \ u_{z+x}(s) ds \text{ and } G(z)(t) = \int_0^t U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds.$ Obviously the operator N has a fixed point is equivalent to F + G has one, so it turns to

prove that F + G has a fixed point. The proof will be given in several steps. First we show that F is continuous and compact.

Step 1 : F is continuous. Let $(z_n)_{n\in\mathbb{N}}$ be a sequence in $B^0_{+\infty}$ such that $z_n \to z$ in $B^0_{+\infty}$. By (H1), (H4) and (5.15), we get for every $t \in [0, n]$

$$\begin{aligned} |F(z_n)(t) - F(z)(t)| &\leq \widehat{MM} \int_0^t |u_{z_n+x}(s) - u_{z+x}(s)| ds \\ &\leq \widehat{M}^2 \widetilde{M} \widetilde{M}_1 \int_0^t \int_0^n |f(\tau, z_{n\rho(\tau, z_{n\tau}+x_{\tau})} + x_{\rho(\tau, z_{n\tau}+x_{\tau})}) \\ &\quad -f(\tau, z_{\rho(\tau, z_{\tau}+x_{\tau})} + x_{\rho(\tau, z_{\tau}+x_{\tau})})| d\tau ds \\ &\leq \widehat{M}^2 \widetilde{M} \widetilde{M}_1 n \int_0^n |f(s, z_{n\rho(s, z_{ns}+x_s)} + x_{\rho(s, z_{ns}+x_s)}) - f(s, z_{\rho(s, z_s+x_s)} + x_{\rho(s, z_s+x_s)})| ds. \end{aligned}$$

Since f is continuous, we obtain by the Lebesgue Dominated Convergence theorem

$$|F(z_n)(t) - F(z)(t)| \to 0$$
 as $n \to +\infty$.

Thus F is continuous.

Step 2 : F maps bounded sets of $B^0_{+\infty}$ into bounded sets. For any d > 0, there exists a positive constant ℓ such that for each $z \in B_d = \{z \in B^0_{+\infty} : ||z||_n \leq d\}$ one has $||F(z)||_n \leq \ell$. Let $z \in B_d$. By (H1), (H2) and (5.15), we have for each $t \in [0, n]$

$$\begin{aligned} |F(z)(t)| &\leq \widehat{M}\widetilde{M} \int_0^t \widetilde{M}_1 \left[|\widehat{y}| + \widehat{M}H \| \phi \|_{\mathcal{B}} \\ &\quad + \widehat{M} \int_0^n p(\tau) \ \psi(\|z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)} \|_{\mathcal{B}}) \ d\tau \right] ds \\ &\leq \widehat{M}\widetilde{M}\widetilde{M}_1 n \left[|\widehat{y}| + \widehat{M}H \| \phi \|_{\mathcal{B}} + \widehat{M} \int_0^n p(s) \ \psi(\|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)} \|_{\mathcal{B}}) ds \right]. \end{aligned}$$

Using Proposition (5.2.1), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_n |z(s)| + (M_n + \mathcal{L}^{\phi}) \|z_0\|_{\mathcal{B}} + K_n |x(s)| + (M_n + \mathcal{L}^{\phi}) \|x_0\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + K_n \|U(s,0)\|_{B(E)} |\phi(0)| + (M_n + \mathcal{L}^{\phi}) \|\phi\|_{\mathcal{B}} \\ &\leq K_n |z(s)| + (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}}. \end{aligned}$$

Set $c_n := (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}}$, then we obtain

$$\|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \le K_n |z(s)| + c_n.$$
(5.16)

Since $z \in B_d$, then we have for $\delta_n := K_n d + c_n$

$$\|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \le K_n |z(s)| + c_n \le \delta_n.$$
(5.17)

Using the nondecreasing character of ψ , we get for each $t \in [0, n]$

$$|F(z)(t)| \leq \widehat{M}\widetilde{M}\widetilde{M}_{1}n\left[|\widehat{y}| + \widehat{M}H||\phi||_{\mathcal{B}} + \widehat{M}\psi(\delta_{n})||p||_{L^{1}}\right] := \varrho.$$

Thus there exists a positive number ρ such that $||F(z)||_n \leq \rho$. Hence $F(B_d) \subset B_{\rho}$. Step 3 : F maps bounded sets into equi-continuous sets of $B^0_{+\infty}$. We consider B_d as in Step 2 and we show that $F(B_d)$ is equi-continuous. Let $\tau_1, \tau_2 \in J$ with $\tau_2 > \tau_1$ and $z \in B_d$. Then

$$\begin{aligned} |F(z)(\tau_2) - F(z)(\tau_1)| &\leq \int_0^{\tau_1} \|U(\tau_2, s) - U(\tau_1, s)\|_{B(E)} \|C\| \|u_{z+x}(s)\| ds \\ &+ \int_{\tau_1}^{\tau_2} \|U(\tau_2, s)\|_{B(E)} \|C\| \|u_{z+x}(s)\| ds. \end{aligned}$$

By the inequalities (5.15) and (5.16) and using the nondecreasing character of ψ , we get

$$|u_{z+x}(t)| \le \widetilde{M}_1 \left[|y^*| + \widehat{M}H \|\phi\|_{\mathcal{B}} + \widehat{M} \psi(\delta_n) \|p\|_{L^1} \right] := \omega.$$
(5.18)

Then

$$|F(z)(\tau_2) - F(z)(\tau_1)| \leq ||C||_{B(E)} \omega \int_0^{\tau_1} ||U(\tau_2, s) - U(\tau_1, s)||_{B(E)} ds + ||C||_{B(E)} \omega \int_{\tau_1}^{\tau_2} ||U(\tau_2, s)||_{B(E)} ds.$$

Noting that $|F(z)(\tau_2) - F(z)(\tau_1)|$ tends to zero as $\tau_2 - \tau_1 \to 0$ independently of $z \in B_d$. The right-hand side of the above inequality tends to zero as $\tau_2 - \tau_1 \to 0$. Since U(t,s) is a strongly continuous operator and the compactness of U(t,s) for t > s implies the continuity in the uniform operator topology (see [7, 90]). As a consequence of Steps 1 to 3 together with the Arzelá-Ascoli theorem it suffices to show that the operator F maps B_d into a precompact set in E.

Let $t \in J$ be fixed and let ϵ be such that $0 < \epsilon < t$. For $z \in B_d$ we define

$$F_{\epsilon}(z)(t) = U(t,t-\epsilon) \int_0^{t-\epsilon} U(t-\epsilon,s) C u_{z+x}(s) ds.$$

Since U(t, s) is a compact operator, the set $Z_{\epsilon}(t) = \{F_{\epsilon}(z)(t) : z \in B_d\}$ is pre-compact in E for every ϵ sufficiently small, $0 < \epsilon < t$. Moreover using (5.18), we have

$$|F(z)(t) - F_{\epsilon}(z)(t)| \leq \int_{t-\epsilon}^{t} ||U(t,s)||_{B(E)} ||C|| ||u_{z+x}(s)| ds$$

$$\leq ||C||_{B(E)} \omega \int_{t-\epsilon}^{t} ||U(t,s)||_{B(E)} ds.$$

Therefore there are precompact sets arbitrary close to the set $\{F(z)(t) : z \in B_d\}$. Hence the set $\{F(z)(t) : z \in B_d\}$ is precompact in E. So we deduce from Steps 1, 2 and 3 that F is a continuous compact operator.

Step 4: We shall show now that the operator G is a contraction. Indeed, consider $z, \overline{z} \in B^0_{+\infty}$. By (H1), (H3) and (5.17), we get for each $t \in [0, n]$ and $n \in \mathbb{N}$

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \int_0^t \widehat{M} \ l_n(s) \ \|z_{\rho(s,z_s+x_s)} - \overline{z}_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} ds \\ &\leq \int_0^t \ \widehat{M} \ K_n \ l_n(s) \ |z(s) - \overline{z}(s)| ds \\ &\leq \int_0^t \ \left[\overline{l}_n(s) \ e^{\tau L_n^*(s)}\right] \ \left[e^{-\tau L_n^*(s)} \ |z(s) - \overline{z}(s)|\right] ds \\ &\leq \int_0^t \ \left[\frac{e^{\tau} \ L_n^*(s)}{\tau}\right]' \ ds \ \|z - \overline{z}\|_n \\ &\leq \frac{1}{\tau} \ e^{\tau} \ L_n^{*(t)} \ \|z - \overline{z}\|_n. \end{aligned}$$

Therefore,

$$||G(z) - G(\overline{z})||_n \le \frac{1}{\tau} ||z - \overline{z}||_n.$$

So, the operator G is a contraction for all $n \in \mathbb{N}$. **S**tep 5 : To apply Theorem (1.10.2), we must check (C2) : i.e. it remains to show that the following set is bounded $\mathcal{E} = \left\{ z \in B^0_{+\infty} : z = \lambda F(z) + \lambda G\left(\frac{z}{\lambda}\right) \text{ for some } 0 < \lambda < 1 \right\}$.

Let $z \in \mathcal{E}$. By (5.15), we have for each $t \in [0, n]$

$$\frac{|z(t)|}{\lambda} \leq \widehat{M}\widetilde{M}\widetilde{M}_{1}n\left[|\widehat{y}| + \widehat{M}H\|\phi\|_{\mathcal{B}}\right] \\
+ \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}n\int_{0}^{n}p(s) \psi\left(\|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}}\right)ds \\
+ \widehat{M}\int_{0}^{t}p(s) \psi\left(\left\|\frac{z_{\rho(s,\frac{z_{s}}{\lambda}+x_{s})}}{\lambda} + x_{\rho(s,\frac{z_{s}}{\lambda}+x_{s})}\right\|_{\mathcal{B}}\right)ds.$$

Using the first inequality in (5.16), we get

$$\begin{aligned} \left\| \frac{z_{\rho(s,\frac{z_s}{\lambda}+x_s)}}{\lambda} + x_{\rho(s,\frac{z_s}{\lambda}+x_s)} \right\|_{\mathcal{B}} &\leq \frac{K_n |z(s)|}{\lambda} + \frac{M_n + \mathcal{L}^{\phi}}{\lambda} \|z_0\|_{\mathcal{B}} \\ &+ K_n |x(s)| + \left(M_n + \mathcal{L}^{\phi}\right) \|x_0\|_{\mathcal{B}} \\ &\leq \frac{K_n |z(s)|}{\lambda} + K_n \|U(s,0)\|_{\mathcal{B}(E)} |\phi(0)| + \left(M_n + \mathcal{L}^{\phi}\right) \|\phi\|_{\mathcal{B}} \\ &\leq \frac{K_n |z(s)|}{\lambda} + \left(K_n \widehat{M} H + M_n + \mathcal{L}^{\phi}\right) \|\phi\|_{\mathcal{B}}. \end{aligned}$$

Then, we get

$$\left\|\frac{z_{\rho(s,\frac{z_s}{\lambda}+x_s)}}{\lambda} + x_{\rho(s,\frac{z_s}{\lambda}+x_s)}\right\|_{\mathcal{B}} \leq \frac{K_n|z(s)|}{\lambda} + c_n.$$
(5.19)

By the previous inequality and the nondecreasing character of ψ , we obtain

$$\frac{|z(t)|}{\lambda} \leq \widehat{M}\widetilde{M}\widetilde{M}_{1}n\left[|\widehat{y}| + \widehat{M}H||\phi||_{\mathcal{B}}\right] + \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}n\int_{0}^{n}p(s)\ \psi(K_{n}|z(s)| + c_{n})ds + \widehat{M}\int_{0}^{t}p(s)\ \psi\left(\frac{K_{n}|z(s)|}{\lambda} + c_{n}\right)ds.$$

Consider the function $\widetilde{u}(t) := \sup_{\theta \in [0,t]} |z(\theta)|$. Then by the nondecreasing character of ψ , we get for $\lambda < 1$ and for $t \in [0, n]$

$$\frac{\widetilde{u}(t)}{\lambda} \leq \widehat{M}\widetilde{M}\widetilde{M}_{1}n\left[|\widehat{y}| + \widehat{M}H\|\phi\|_{\mathcal{B}}\right] + \widehat{M}\left(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1\right)\int_{0}^{n}p(s)\ \psi\left(\frac{K_{n}\widetilde{u}(s)}{\lambda} + c_{n}\right)ds.$$

We consider the function μ defined by $\mu(t) = \sup_{s \in [0,t]} \frac{K_n \widetilde{u}(s)}{\lambda} + c_n$ for $t \in J$. Let $t^* \in [0,t]$ be such that $\mu(t) = \frac{K_n u(t^*)}{\lambda} + c_n$. If $t^* \in [0,n]$, by the previous inequality and the nondecreasing character of ψ , we have for $\alpha_n := c_n + K_n \widehat{M} \widetilde{M} \widetilde{M}_1 n \left[|\widehat{y}| + \widehat{M} H| |\phi||_{\mathcal{B}} \right]$

$$\mu(t) \le \alpha_n + K_n \widehat{M} \left(\widehat{M} \widetilde{M} \widetilde{M}_1 n + 1 \right) \int_0^n p(s) \ \psi(\mu(s)) \ ds$$

Consequently,

$$\frac{\|z\|_n}{\alpha_n + K_n \widehat{M}\left(\widehat{M}\widetilde{M}\widetilde{M}_1 n + 1\right)\psi(\|z\|_n)\|p\|_{L^1}} \le 1.$$

Then by the condition (5.14), there exists a constant M^n_{\star} such that $\mu(t) \leq M^n_{\star}$. Since $||z||_n \leq \mu(t)$, we have $||z||_n \leq M^n_{\star}$. This shows that the set \mathcal{E} is bounded, i.e. the statement (C2) in Theorem (1.10.2) does not hold. Then the Avramescu's nonlinear alternative ([18]) implies that (C1) holds : i.e. the operator F + G has a fixed-point z^{\star} . Then, there exists at least $y^{\star}(t) = z^{\star}(t) + x(t), t \in \mathbb{R}$ which is a fixed point of the operator N, which is a mild solution of the problem (5.1) – (5.2). Thus the evolution system (5.1) – (5.2) is controllable on \mathbb{R} . Then, the proof is complete.

5.3 Neutral Problem with Infinite Delay

Before stating and proving our second main result, we define firstly the corresponding mild solution then we define the concept of controllability for that problem.

Definition 5.3.1. We say that the function $y(\cdot) : \mathbb{R} \to E$ is a mild solution of (5.3) - (5.4) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \int_0^t U(t,s)A(s)g(s, y_{\rho(s,y_s)})ds + \int_0^t U(t,s)Cu(s)ds + \int_0^t U(t,s)f(s, y_{\rho(s,y_s)})ds \quad \text{for each } t \ge 0.$$
(5.20)

Definition 5.3.2. The neutral evolution problem (5.3) - (5.4) is said to be controllable if for every initial function $\phi \in \mathcal{B}$, $y^* \in E$ and $n \in \mathbb{N}$, there is some control $u \in L^2([0,n]; E)$ such that the mild solution $y(\cdot)$ of (5.3) - (5.4) satisfies $y(n) = y^*$.

We consider the function $\rho : J \times \mathcal{B} \longrightarrow \mathbb{R}$ satisfies the hypothesis (H_{ϕ}) and the Lemma (5.2.1). We assume here that the hypotheses (H0) - (H4) hold and we need the following assumptions :

(H5) There exists a constant $\overline{M}_0 > 0$ such that $||A^{-1}(t)||_{B(E)} \leq \overline{M}_0$ for all $t \in J$.

(H6) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_n}$, such that

$$|A(t) g(t,\phi)| \le L (||\phi||_{\mathcal{B}} + 1)$$
 for all $t \in J$ and $\phi \in \mathcal{B}$.

(H7) There exists a constant $L_* > 0$ such that

$$|A(s) \ g(s,\phi) - A(\overline{s}) \ g(\overline{s},\overline{\phi})| \le L_* \ (|s-\overline{s}| + \|\phi - \overline{\phi}\|_{\mathcal{B}})$$

for all $s, \overline{s} \in J$ and $\phi, \overline{\phi} \in \mathcal{B}$.

(H8) The function g is completely continuous and for each bounded sub-set $Q \subset \mathcal{B}$, the mapping $\{t \longrightarrow g(t, x_{\rho(s, y_s)})\}$ is equicontinous in C(J, E).

Theorem 5.3.1. Suppose that the hypotheses (H0) - (H8) are satisfied and moreover

$$\frac{M^{\star\star}}{\gamma_n + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \left(\widehat{M} \widetilde{M} \widetilde{M}_1 n + 1\right) \left[M^{\star\star} + \psi(M^{\star\star})\right] \psi(\|z\|_n) \|\zeta\|_{L^1}} > 1, \quad (5.21)$$

where $\zeta(t) = \max(L; p(t))$ and $\gamma_n = (M_n + \mathcal{L}^{\phi} + K_n \widehat{M} H) \|\phi\|_{\mathcal{B}} + \frac{K_n \beta_n}{1 - \overline{M}_0 L K_n}$ with

$$\beta_{n} = \left[(\widehat{M} + 1)\overline{M}_{0}L + \widehat{M}Ln \right] \left[\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right] + \widehat{M}\widetilde{M}\widetilde{M}_{1}n \left(1 + K_{n}\overline{M}_{0}L \right) |\widehat{y}| \\ + \left[\left(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right) \overline{M}_{0}L \left[\widehat{M} + M_{n} + \mathcal{L}^{\phi} \right] + \widehat{M}H \left(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + \overline{M}_{0}LK_{n} \right) \right] \|\phi\|_{\mathcal{B}}.$$

Then the neutral evolution problem (5.3) - (5.4) is controllable on \mathbb{R} .

Proof. Consider the operator $\widetilde{N}: B_{+\infty} \to B_{+\infty}$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \int_0^t U(t,s)A(s)g(s, y_{\rho(s,y_s)})ds \\ + \int_0^t U(t,s)Cu(s)ds + \int_0^t U(t,s)f(s, y_{\rho(s,y_s)})ds, & \text{if } t \in J. \end{cases}$$

Then, fixed points of the operator \tilde{N} are mild solutions of the problem (5.3) – (5.4). Using assumption (H4), for arbitrary function $y(\cdot)$, we define the control

$$u_{y}(t) = \tilde{W}^{-1} \left[y^{*} - U(n,0) \left(\phi(0) - g(0,\phi) \right) - g(n,y_{\rho(n,y_{n})}) - \int_{0}^{n} U(n,\tau) A(\tau) g(\tau,y_{\rho(\tau,y_{\tau})}) d\tau - \int_{0}^{n} U(n,\tau) f(\tau,y_{\rho(\tau,y_{\tau})}) d\tau \right] (t).$$

Noting that by (H1), (H2), (H4), (H5) and (H7) we get

$$|u_{y}(t)| \leq \widetilde{M}_{1} \left[|y^{*}| + \widehat{M} \left(H + \overline{M}_{0}L \right) \|\phi\|_{\mathcal{B}} + \left(\widehat{M} + 1 \right) \overline{M}_{0}L + \widehat{M}Ln \right] + \widetilde{M}_{1}\overline{M}_{0}L \|y_{\rho(n,y_{n})}\|_{\mathcal{B}} + \widetilde{M}_{1}\widehat{M}L \int_{0}^{n} \|y_{\rho(\tau,y_{\tau})}\|_{\mathcal{B}}d\tau + \widetilde{M}_{1}\widehat{M} \int_{0}^{n} p(\tau)\psi(\|y_{\rho(\tau,y_{\tau})}\|_{\mathcal{B}})d\tau.$$
(5.22)

Using this control the operator \widetilde{N} has a fixed point $y(\cdot)$. Then $y(\cdot)$ is a mild solution of the neutral evolution system (5.3) - (5.4).

For $\phi \in \mathcal{B}$, we will define the function $x(.) : \mathbb{R} \to E$ by $x(t) = \phi(t)$ for $t \leq 0$ and $x(t) = U(t,0) \phi(0)$ for $t \in J$. Then $x_0 = \phi$. For each function $z \in B_{+\infty}$ with z(0) = 0, we denote by \overline{z} the function defined by $\overline{z}(t) = 0$ for t < 0 and $\overline{z}(t) = z(t)$ for $t \in J$.

If $y(\cdot)$ satisfies (5.20), we decompose it as $y(t) = z(t) + x(t), t \ge 0$, which implies $y_t = z_t + x_t$, for every $t \in J$ and the function $z(\cdot)$ satisfies $z_0 = 0$ and for $t \in J$, we get

$$\begin{aligned} z(t) &= g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)}) - U(t, 0)g(0, \phi) \\ &+ \int_0^t U(t, s)A(s)g(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})ds + \int_0^t U(t, s)Cu_{z+x}(s)ds \\ &+ \int_0^t U(t, s)f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})ds. \end{aligned}$$

Let us define the operators $\widetilde{F}, G: B^0_{+\infty} \longrightarrow B^0_{+\infty}$ by

$$\widetilde{F}(z)(t) = g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)}) - U(t, 0)g(0, \phi) + \int_0^t U(t, s)A(s)g(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})ds + \int_0^t U(t, s)Cu_{z+x}(s)ds$$

and

 $G(z)(t) = \int_0^t U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds.$ Obviously the operator \widetilde{N} has a fixed point is equivalent to $\widetilde{F} + G$ has one, so it turns to prove that $\widetilde{F} + G$ has a fixed point.

We can show as in Section 3 that the operator \widetilde{F} is continuous and compact and the operator G is a contraction. For applying Avramescu's nonlinear alternative, we must check (C2) in Theorem (1.10.2): i.e. it remains to show that the following set

$$\widetilde{\mathcal{E}} = \left\{ z \in B^0_{+\infty} : \ z = \lambda \widetilde{F}(z) + \lambda G\left(\frac{z}{\lambda}\right) \text{ for some } 0 < \lambda < 1 \right\}$$

is bounded.

Let $z \in \widetilde{\mathcal{E}}$. Then, using (H1) - (H6) and (5.22), we have for each $t \in [0, n]$

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq \left[(\widehat{M}+1)\overline{M}_{0}L + \widehat{M}Ln \right] \left[\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right] + \widehat{M}\widetilde{M}\widetilde{M}_{1}n|\widehat{y}| \\ &+ \widehat{M} \left[\overline{M}_{0}L \left(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right) + \widehat{M}\widetilde{M}\widetilde{M}_{1}nH \right] \|\phi\|_{\mathcal{B}} \\ &+ \widehat{M}\widetilde{M}\widetilde{M}_{1}\overline{M}_{0}Ln \|z_{\rho(n,z_{n}+x_{n})} + x_{\rho(n,z_{n}+x_{n})}\|_{\mathcal{B}} \\ &+ \overline{M}_{0}L \|z_{\rho(t,z_{t}+x_{t})} + x_{\rho(t,z_{t}+x_{t})}\|_{\mathcal{B}} + \widehat{M}L \int_{0}^{t} \|z_{\rho(s,z_{s}+x_{s})} + x_{\rho(s,z_{s}+x_{s})}\|_{\mathcal{B}} \, ds \\ &+ \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}Ln \int_{0}^{n} \|\left(z_{\rho(\tau,z_{\tau}+x_{\tau})}\right) + x_{\rho(\tau,z_{\tau}+x_{\tau})}\|_{\mathcal{B}} \, d\tau \\ &+ \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}n \int_{0}^{n} p(\tau) \, \psi(\|\left(z_{\rho(\tau,z_{\tau}+x_{s})}\right) + x_{\rho(\tau,z_{\tau}+x_{\tau})}\|_{\mathcal{B}}) \, d\tau \\ &+ \widehat{M} \int_{0}^{t} p(s) \, \psi\left(\left\|\frac{z_{\rho(s,\frac{z_{\tau}}{\lambda}+x_{s})}}{\lambda} + x_{\rho(s,\frac{z_{s}}{\lambda}+x_{s})}\right\|_{\mathcal{B}}\right) ds. \end{aligned}$$

By Proposition (5.2.1), we obtain $||z_{\rho(n,z_n+x_n)} + x_{\rho(n,z_n+x_n)}||_{\mathcal{B}} \leq K_n |\hat{y}| + (M_n + \mathcal{L}^{\phi}) ||\phi||_{\mathcal{B}}$. Using the inequalities (5.16) and (5.19), we have

$$\begin{aligned} \frac{|z(t)|}{\lambda} &\leq \left[(\widehat{M}+1)\overline{M}_{0}L + \widehat{M}Ln \right] \left[\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right] + \widehat{M}\widetilde{M}\widetilde{M}_{1}n|\widehat{y}| \\ &+ \widehat{M} \left[\overline{M}_{0}L \left(\widehat{M}\widetilde{M}\widetilde{M}_{1}n + 1 \right) + \widehat{M}\widetilde{M}\widetilde{M}_{1}nH \right] \|\phi\|_{\mathcal{B}} \\ &+ \widehat{M}\widetilde{M}\widetilde{M}_{1}\overline{M}_{0}Ln \left(K_{n}|\widehat{y}| + (M_{n} + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \right) \\ &+ \overline{M}_{0}L \left(K_{n}|z(t)| + (M_{n} + \mathcal{L}^{\phi} + K_{n}\widehat{M}H)\|\phi\|_{\mathcal{B}} \right) \\ &+ \widehat{M}L \int_{0}^{t} \left(K_{n}|z(s)| + c_{n} \right) ds + \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}Ln \int_{0}^{n} \left(K_{n}|z(\tau)| + c_{n} \right) d\tau \\ &+ \widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}n \int_{0}^{n} p(\tau) \psi \left(K_{n}|z(\tau)| + c_{n} \right) d\tau \\ &+ \widehat{M} \int_{0}^{t} p(s) \psi \left(\frac{K_{n}|z(s)|}{\lambda} + c_{n} \right) ds. \end{aligned}$$

We consider the function $\widetilde{u}(t) := \sup_{\theta \in [0,t]} |z(\theta)|$ then by the nondecreasing character of ψ , we obtain for $\beta_n := \left[(\widehat{M} + 1)\overline{M}_0 L + \widehat{M}Ln \right] \left[\widehat{M}\widetilde{M}\widetilde{M}_1 n + 1 \right] + \widehat{M}\widetilde{M}\widetilde{M}_1 n \left(1 + K_n \overline{M}_0 L \right) |\widehat{y}| + \left[\left(\widehat{M}\widetilde{M}\widetilde{M}_1 n + 1 \right) \overline{M}_0 L \left[\widehat{M} + M_n + \mathcal{L}^{\phi} \right] + \widehat{M}H \left(\widehat{M}\widetilde{M}\widetilde{M}_1 n \overline{M}_0 L K_n \right) \right] \|\phi\|_{\mathcal{B}} \text{ and for } \lambda < 1,$ $\frac{\widetilde{u}(t)}{\lambda} \left(1 - \overline{M}_0 L K_n \right) \leq \beta_n + \widehat{M}L \left(\widehat{M}\widetilde{M}\widetilde{M}_1 n + 1 \right) \int_0^n \left(\frac{K_n \widetilde{u}(s)}{\lambda} + c_n \right) ds \\ + \widehat{M} \left(\widehat{M}\widetilde{M}\widetilde{M}_1 n + 1 \right) \int_0^n p(s) \psi \left(\frac{K_n \widetilde{u}(s)}{\lambda} + c_n \right) ds.$

We consider the function μ defined by $\mu(t) = \sup_{s \in [0,t]} \frac{K_n \widetilde{u}(s)}{\lambda} + c_n$ for $t \in J$. Let $t^* \in [0,t]$ be such that $\mu(t) = \frac{K_n u(t^*)}{\lambda} + c_n$. By the previous inequality, we have for $\gamma_n :=$ $c_n + \frac{K_n \beta_n}{1 - \overline{M}_0 L K_n}$, then we obtain for $t \in [0, n]$

$$\mu(t) \le \gamma_n + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \left(\widehat{M} \widetilde{M} \widetilde{M}_1 n + 1 \right) \int_0^n \left[L \mu(s) + p(s) \ \psi(\mu(s)) \right] ds$$

Set $\zeta(t) := \max(L; p(t))$ for $t \in [0, n]$. Consequently, we get

$$\frac{\|z\|_n}{\gamma_n + \frac{K_n \widehat{M}}{1 - \overline{M}_0 L K_n} \left(\widehat{M} \widetilde{M} \widetilde{M}_1 n + 1\right) \left[\|z\|_n + \psi(\|z\|_n)\right] \psi(\|z\|_n) \|\zeta\|_{L^1}} \le 1.$$

Then by the condition (5.21), there exists a constant $M_{\star\star}^n$ such that $\mu(t) \leq M_{\star\star}^n$. Since $||z||_n \leq \mu(t)$, we have $||z||_n \leq M_{\star\star}^n$. This shows that the set $\widetilde{\mathcal{E}}$ is bounded, i.e. the statement (C2) in Theorem (1.10.2) does not hold. Then the nonlinear alternative due to Avramescu ([18]) implies that (C1) holds : i.e. the operator $\widetilde{F} + G$ has a fixed-point $z^{\star\star}$. Then, there exists at least $y^{\star\star}(t) = z^{\star\star}(t) + x(t), t \in \mathbb{R}$ which is a fixed point of the operator \widetilde{N} , which is a mild solution of the problem (5.3) – (5.4). Thus the neutral evolution system (5.3) – (5.4) is controllable on \mathbb{R} . Then, the proof is complete.

5.4 Fractional Partial Problem with Finite Delay

Before stating and proving the main result, we give first the definition of mild solution of the semilinear evolution problem (5.5) - (5.6).

Lemma 5.4.1. The system (5.5) - (5.6) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} C u(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) \, ds.$$
(5.23)

In other words, every solution of the integral equation (5.33) is also solution of the system (5.5) - (5.6) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (5.5) - (5.6), and using some classical results from fractional calculus to get (5.33).

Definition 5.4.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (5.5) - (5.6)if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0) \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) Cu(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, y_{\rho(s,y_s)}) ds \ t \in J.$$
(5.24)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(H;E), \rho(s,\varphi) \le 0\}.$$

We always assume that $\rho: J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter :

(H'1) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H'2) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in C(H; E)$.

(H'3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in C(H; E)$ with $||u|| \le R$ and $||v|| \le R$.

(H'4) The linear operator $W: L^2(J; E) \to E$ is defined by

$$Wu = \int_0^b U(b,s)Cu(s)ds,$$

has an induced invertible operator \tilde{W}^{-1} which takes values in $L^2(J; E) / \ker W$ and there exists positive constants \widetilde{M} and \widetilde{M}_1 such that

$$||C|| \le \widetilde{M}$$
 and $||\widetilde{W}^{-1}|| \le \widetilde{M}_1$.

Theorem 5.4.1. Suppose that the hypotheses (H'1) - (H'4) are satisfied and moreover there exists a constant $M_{\star} > 0$ such that

$$\frac{M_{\star}}{\delta_{b} + \frac{\widehat{M}q^{*}[\widehat{M}\widetilde{M}\widetilde{M}_{1} + \Gamma(\alpha)]}{\alpha\Gamma(\alpha)^{2}}M_{\star} b^{\alpha}} > 1$$
(5.25)

with

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M} H \|\phi\|_{\mathcal{B}} \right] + \frac{K_b \widehat{M} p^* b^{\alpha} [\alpha \widehat{M} \widetilde{M} \widetilde{M}_1 b + \Gamma(\alpha)]}{\alpha \Gamma(\alpha)^2}$$

Then the problem (5.5) - (5.6) has a unique controllable mild solution on [-r, b].

Proof. Transform the problem (5.5) - (5.6) into a fixed-point problem. Consider the operator $N: B_b \to B_b$ defined by :

$$N(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in [-r, 0]; \\ U(t, 0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ C \ u_y(s) \ ds & (5.26) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, y_{\rho(s, y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$

Using assumption (H'3), for arbitrary function $y(\cdot)$, we define the control

$$u_y(t) = \tilde{W}^{-1} \left[y_1 - U(b,0) \ \varphi(0) - \frac{1}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \ U(b,s) \ f(s,y_{\rho(s,y_s)}) \ ds \right] (t).$$

Noting that, we have

$$|u_{y}(t)| \leq \|\tilde{W}^{-1}\| \left[|y_{1}| + \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-\tau)^{\alpha-1} \|U(b,\tau)\|_{B(E)} |f(\tau,y_{\rho(\tau,y_{\tau})})| d\tau \right]$$

From (H'2), we get

$$|u_y(t)| \leq \widetilde{M}_1 \left[|y_1| + \widehat{M}H \|\varphi\|_{\mathcal{B}} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-\tau)^{\alpha-1} \left[p(\tau) + q(\tau) \|y_{\rho(\tau,y_\tau)}\|_{\mathcal{B}} \right] ds \right]$$
(5.27)

Clearly, fixed points of the operator N are mild solutions of the problem (5.5) - (5.6).

Let y be a possible solution of the problem (5.5) – (5.6). Given $t \leq b$, then from (H'1), (H'2), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} |f(s,y_{\rho(s,y_s)})| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|C\| \|u_{y}(s)\| \, ds \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \left[p(s) + q(s)\|y_{\rho(s,y_s)}\|\right] \, ds \\ &+ \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} |u_{y}(s)| \, ds \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \left[p(s) + q(s)\|y_{\rho(s,y_s)}\|\right] \, ds \\ &+ \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_{0}^{t} \widetilde{M}_{1} \left[|y_{1}| + \widehat{M}H\|\phi\|_{\mathcal{B}} \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \left[p(\tau) + q(\tau)\|y_{\rho(\tau,y_{\tau})}\|d\tau\right] \right] \, ds \end{aligned}$$

It follows that

$$\begin{aligned} |y(t)| + \mathcal{L}^{\varphi} \|\varphi\| &\leq \widehat{M} \|\varphi\| + \mathcal{L}^{\varphi} \|\varphi\| + \frac{\widehat{M}MM_{1}b}{\Gamma(\alpha)} \left[|y_{1}| + \widehat{M}H\|\phi\|_{\mathcal{B}} \right] \\ &+ \frac{\widehat{M}b^{\alpha}p^{*}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}b)}{\alpha\Gamma^{2}(\alpha)} \\ &+ \frac{\widehat{M}q^{*}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}b)}{\Gamma^{2}(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \left[|y(s) + \mathcal{L}^{\varphi}\|\varphi\| \right] ds \end{aligned}$$

Set

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}\widetilde{M}\widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M}H \|\phi\|_{\mathcal{B}} \right] + \frac{\widehat{M}b^{\alpha} p^*(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1 b)}{\alpha \Gamma^2(\alpha)}$$

We consider the function μ defined by

 $\mu(t) := \sup \{ \|y(s)\| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$

Let $t^* \in [-r, b]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}q^*(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1b)}{\Gamma^2(\alpha)} \int_0^b (t-s)^{\alpha-1} \mu(s)ds \quad \text{for } t \in [0,b].$$

If $t^{\star} \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. Consequently,

$$\frac{\|z\|_{n}}{\delta_{b} + \frac{\widehat{M}q^{*}[\widehat{M}\widetilde{M}\widetilde{M}_{1} + \Gamma(\alpha)]}{\alpha\Gamma(\alpha)^{2}} \|z\|_{n} b^{\alpha}} \leq 1$$

Then by (5.25), there exists a constant M^* such that $\mu(t) \leq M^*$ Since $||y|| \leq \mu(t)$, Since for every $t \in [0, b]$ we have $||y|| \leq \max\{||\varphi||, M^*\} := \Theta_b$.

Set

$$Y = \{ y \in C([-r, b]; E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1 \}$$

Clearly, Y is a closed subset of C([-r, b]; E).

We shall show that $N: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H'1) and (H'3) for each $t \in [0, b]$

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} \left| f(s, y_{\rho(s,y_s)}) - f(s, \overline{y}_{\rho(s,y_s)}) \right| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} \, ||C|| \, |u_y(s) - u_{\overline{y}}| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \, l_b(s) \, ||y_{\rho(s,y_s)} - \overline{y}_{\rho(s,y_s)}|| \, ds \\ &+ \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \widehat{M}_1 \widehat{M} \int_0^b |f(\tau, y_{\rho(\tau,y_\tau)})| \\ &- f(\tau, \overline{y}_{\rho(\tau,y_\tau)})| \, d\tau \, ds \end{aligned}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{split} |N(y)(t) - N(\overline{y})(t)| &\leq \frac{\widehat{M}l_b^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &+ \frac{\widehat{M}^2 \widetilde{M} \widetilde{M}_1 l_b^* b^\alpha}{\Gamma(\alpha+1) \Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &\leq \frac{\widehat{M}l_b^* b^\alpha}{\Gamma(\alpha+1)} \|y - \overline{y}\| + \frac{\widehat{M}^2 \widetilde{M} \widetilde{M}_1 l_b^* b^\alpha}{\Gamma(\alpha+1) \Gamma(\alpha)} \|y - \overline{y}\| \end{split}$$

Therefore,

$$\|N(y) - N(\overline{y})\|_{n} \leq \frac{\widehat{M}l_{b}^{*}b^{\alpha}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1})}{\Gamma(\alpha+1)\Gamma(\alpha)}\|y - \overline{y}\|.$$

So, for $\frac{\widehat{M}l_b^*b^{\alpha}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1)}{\Gamma(\alpha + 1)\Gamma(\alpha)} < 1$ the operator N is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y), \ \lambda \in (0,1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that (C1) holds. Thus, we deduce that the operator N has a unique fixed-point y^* which is the unique controllable mild solution of the problem (5.5) - (5.6).

5.5 Fractional Neutral Problem with Finite Delay

In this section, we give controllability result for the neutral functional fractional differential evolution problem with infinite delay (5.7) - (5.8). Firstly, we define its mild solution.

Lemma 5.5.1. The system (5.7) - (5.8) is equivalent to the nonlinear integral equation

$$y(t) = [\varphi(0) - g(0, \varphi)] + g(t, y_{\rho(t, y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} A(s) y(s) \, ds \qquad (5.28)$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t U(t,s) Cu(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s, y_{\rho(s,y_s)}) ds.$$

In other words, every solution of the integral equation (5.28) is also solution of the system (5.7) - (5.8) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (5.7) - (5.8), and using some classical results from fractional calculus to get (5.28).

Definition 5.5.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (5.7) - (5.8)if $y(t) = \varphi(t)$ for all $t \in H$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s,y_{\rho(s,y_s)}) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) Cu(s) ds \quad \text{for each } t \in J.$$
(5.29)

Definition 5.5.2. The neutral evolution problem (5.7) - (5.8) is said to be controllable on [-r, b] if for every initial function $\varphi \in C(H; E)$ and $y_1 \in E$ there exists a control $u \in L^2(J; E)$ such that the mild solution $y(\cdot)$ of (5.7) - (5.8) satisfies $y(b) = y_1$.

We consider the hypotheses (H_{φ}) , (H1) - (H3) and we need the following assumptions (H'5) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} \left(|s - \overline{s}| + \|\varphi - \overline{\varphi}\|\right)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

(H'6) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H'7) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

$$|A(t) g(t, \varphi)| \le L (||\varphi|| + 1)$$
 for all $t \in J$ and $\varphi \in C(H; E)$.

(H'8) The function g is completely continuous and for any bounded set $Q \subseteq \mathcal{B}$ the set $\{t \to g(t, x_t) : x \in Q\}$ is equi-continuous in C(J; E).

Theorem 5.5.1. Suppose that the hypotheses (H'1) - (H'8) are satisfied and moreover there exists a constant $M^{\star\star} > 0$ with

$$\frac{M^{\star\star}}{\beta_b + \frac{\widehat{M}q^*[\widehat{M}\widetilde{M}\widetilde{M}_1b + \Gamma(\alpha)]}{\alpha\Gamma(\alpha)^2}}M^{\star\star}b^{\alpha} > 1,$$
(5.30)

where
$$\beta_b = \mathcal{L}^{\varphi} \|\varphi\| + \frac{\delta_b}{(1 - \overline{M}_0 L)}$$
 and

$$\delta_b := \widetilde{M}_1 \overline{M}_0 L |y_1| + \widehat{M} \|\varphi\| + \overline{M}_0 L \left(\mathcal{L}^{\varphi} \|\varphi\| + 1\right) + \widehat{M} \overline{M}_0 L \left(\|\varphi\| + 1\right) + \frac{\widetilde{M} \widetilde{M} \widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M} H \|\varphi\|_{\mathcal{B}} \right] + \frac{\widetilde{M} b^{\alpha} p^*(\Gamma(\alpha) + \widetilde{M} \widetilde{M} \widetilde{M}_1 b)}{\alpha \Gamma^2(\alpha)}.$$

Then the neutral evolution problem (5.7) - (5.8) has a unique controllable mild solution on [-r, b].

Proof. Consider the operator $\widetilde{N}: B_b \to B_b$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \varphi(t), & \text{if } t \in [-r, 0]; \\ U(t, 0) \ [\varphi(0) - g(0, \varphi)] + g(t, y_{\rho(t, y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ C \ u_y(s) \ ds \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, y_{\rho(s, y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(5.31)

Using assumption (H'4), for arbitrary function $y(\cdot)$, we define the control

$$u_{y}(t) = \tilde{W}^{-1} \left[y_{1} - U(b,0) \left(\varphi(0) - g(0,\varphi) \right) - g(b, y_{\rho(b,y_{b})}) - \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} U(b,s) f(s, y_{\rho(s,y_{s})}) ds \right] (t).$$

Noting that

$$\begin{aligned} |u_{y}(t)| &\leq \|\widetilde{W}^{-1}\| \left[\|y_{1}\| + \|U(t,0)\|_{B(E)} \left(|\varphi(0)| + \|A^{-1}(0)\| |A(0)g(0,\varphi)| \right) \\ &+ \|A^{-1}(b)\| |A(b)g(b,y_{\rho(b,y_{b})})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} U(b,\tau) f(\tau,y_{\rho(\tau,y_{\tau})}) d\tau \right] \\ &\leq \widetilde{M}_{1} \left[\|y_{1}\| + \widehat{M}H\| \varphi \| + \widehat{M}\overline{M}_{0}L(\|\varphi\| + 1) + \overline{M}_{0}L(\|y_{\rho(b,y_{b})}\| + 1) \right] \\ &+ \frac{\widetilde{M}_{1}\widehat{M}}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \left[p(\tau) + q(\tau) \|y_{\rho(\tau,y_{\tau})}\| \right]. \end{aligned}$$

we get

$$|u_{y}(t)| \leq \widetilde{M}_{1} \left[|y_{1}| + \widehat{M} \left(H + \overline{M}_{0}L \right) \|\varphi\| + \left(\widehat{M} + 1 \right) \overline{M}_{0}L \right] + \widetilde{M}_{1}\overline{M}_{0}L \|y_{\rho(b,y_{b})}\| + \frac{\widehat{M}\widetilde{M}_{1}p^{*}b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}\widetilde{M}_{1}q^{*}}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \|y_{\rho(s,y_{s})}\| ds$$

$$(5.32)$$

We shall show that using this control the operator \tilde{N} has a fixed point $y(\cdot)$, which is a mild solution of the neutral evolution system (5.7) - (5.8).

Let y be a possible solution of the problem (5.7)–(5.8). Given $t \leq b$, then from (H'1)-(H'2), (H'4) - (H'7), (H_{φ}) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq |g(t, y_{\rho(t,y_t)})| + |U(t, 0)[\varphi(0) - g(0, \varphi)]| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \|f(s, y_{\rho(s,y_s)})\| \, ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \|C\| \|u_y(s)\| \, ds \\ &\leq \|A^{-1}(t)\| \|A(t)g(t, y_{\rho(t,y_s)})\| + \|U(t, 0)\|_{B(E)} \|A^{-1}(0)\| \|A(0)g(0, \varphi)\| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s)\|y_{\rho(s,y_s)}\|\right] \, ds \\ &+ \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} |u_y(s)| \, ds \end{aligned}$$

$$\begin{aligned} |y(t)| &\leq \widehat{M} \|\varphi\| + \overline{M}_0 L \left(\|y_{\rho(t,y_t)}\| + 1 \right) + \widehat{M} \overline{M}_0 L \left(\|\varphi\| + 1 \right) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|y_{\rho(s,y_s)}\| \right] ds \\ &+ \frac{\widehat{M} \widetilde{M}}{\Gamma(\alpha)} \int_0^t \widetilde{M}_1 \left[|y_1| + \widehat{M} \left(H + \overline{M}_0 L \right) \|\varphi\| + \left(\widehat{M} + 1 \right) \overline{M}_0 L \right. \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \left[p(\tau) + q(\tau) \|y_{\rho(\tau,y_\tau)}\| d\tau \right] \right] ds \\ &+ \widetilde{M}_1 \overline{M}_0 L \|y_{\rho(b,y_b)}\| \end{aligned}$$

Since $||y_{\rho(t,y_t)}|| \leq |y(t)| + \mathcal{L}^{\varphi}||\varphi||$, we have

$$\begin{aligned} (1 - \overline{M}_{0}L)|y(t)| &\leq \widetilde{M}_{1}\overline{M}_{0}L|y_{1}| + \widehat{M}||\varphi|| + \overline{M}_{0}L\left(\mathcal{L}^{\varphi}||\varphi|| + 1\right) + \widehat{M}\overline{M}_{0}L\left(||\varphi|| + 1\right) \\ &+ \frac{\widehat{M}\widetilde{M}\widetilde{M}_{1}b}{\Gamma(\alpha)}\left[|y_{1}| + \widehat{M}H||\varphi||\right] + \frac{\widehat{M}b^{\alpha}p^{*}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}b)}{\alpha\Gamma^{2}(\alpha)} \\ &+ \frac{\widehat{M}q^{*}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}b)}{\Gamma^{2}(\alpha)}\int_{0}^{b}(t - s)^{\alpha - 1}\left[|y(s) + \mathcal{L}^{\varphi}||\varphi||\right]ds \end{aligned}$$

 Set

$$\delta_{b}: = \widetilde{M}_{1}\overline{M}_{0}L|y_{1}| + \widehat{M}\|\varphi\| + \overline{M}_{0}L\left(\mathcal{L}^{\varphi}\|\varphi\| + 1\right) + \widehat{M}\overline{M}_{0}L\left(\|\varphi\| + 1\right) \\ + \frac{\widetilde{M}\widetilde{M}\widetilde{M}_{1}b}{\Gamma(\alpha)}\left[|y_{1}| + \widehat{M}H\|\varphi\|\right] + \frac{\widehat{M}b^{\alpha}p^{*}(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}b)}{\alpha\Gamma^{2}(\alpha)}.$$

Then

$$(1 - \overline{M}_0 L)|y(t)| \leq \delta_b + \frac{\widehat{M}q^*(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1 b)}{\Gamma^2(\alpha)} \int_0^b (t - s)^{\alpha - 1} \left[|y(s) + \mathcal{L}^{\varphi} \|\varphi\| \right] ds,$$

with $\beta_b = \mathcal{L}^{\varphi} \|\varphi\| + \frac{\delta_b}{(1 - \overline{M}_0 L)}.$ Then

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \leq \beta_b + \frac{\widehat{M}q^* \left(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1b\right)}{(1 - \overline{M}_0 L)\Gamma^2(\alpha)} \int_0^b (t - s)^{\alpha - 1} \left[|y(s) + \mathcal{L}^{\varphi} \|\varphi\|\right] ds$$

We consider the function μ defined by

$$\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, b]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \beta_b + \frac{\widehat{M}q^*(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_1b)}{(1 - \overline{M}_0L)\Gamma^2(\alpha)} \int_0^b (t - s)^{\alpha - 1} \mu(s)ds \quad \text{for } t \in [0, b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. Consequently,

$$\frac{\|y\|}{\beta_b + \frac{\widehat{M}q^*[\widehat{M}\widetilde{M}\widetilde{M}_1b + \Gamma(\alpha)]}{\alpha\Gamma(\alpha)^2} \|y\| b^{\alpha}} \le 1.$$

Then by (5.30), there exists a constant $M^{\star\star}$ such that $\mu(t) \leq M^{\star\star}$, thus $||y|| \leq \mu(t)$. Since for every $t \in [0, b]$, we have $||y|| \leq \max\{||\varphi||, M^{\star\star}\} := \Theta_b$.

 Set

 $Y = \{ y \in C([-r, b]; E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1\}.$

Clearly, Y is a closed subset of C([-r,b];E).

We shall show that $N: Y \to C([-r, b]; E)$ is a contraction operator. Indeed, consider $y, \overline{y} \in Y$, thus using (H'1) and (H'3) for each $t \in [0, b]$, we get

$$\begin{aligned} |N(y)(t) - N(\overline{y})(t)| &\leq |g(t, y_{\rho(t, y_t)}) - g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} ||U(t, s)||_{B(E)} ||f(s, y_{\rho(s, y_s)}) - f(s, \overline{y}_{\rho(s, y_s)})| ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} ||U(t, s)||_{B(E)} ||C|| |u_y(s) - u_{\overline{y}}| ds \\ &\leq |g(t, y_{\rho(t, y_t)}) - g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} l_b(s) ||y_{\rho(s, y_s)} - \overline{y}_{\rho(s, y_s)}|| ds \\ &+ \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \widetilde{M}_1 \widehat{M} \int_0^b |f(\tau, y_{\rho(\tau, y_\tau)}) - f(\tau, \overline{y}_{\rho(\tau, y_\tau)})| d\tau ds \end{aligned}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{split} |N(y)(t) - N(\overline{y})(t)| &\leq ||A^{-1}(t)|||A(t)g(t, y_{\rho(t,y_t)}) - A(t)g(t, \overline{y}_{\rho(t,y_t)})| \\ &+ \frac{\widehat{M}l_b^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &+ \frac{\widehat{M}^2 \widetilde{M} \widetilde{M}_1 l_b^* b^{\alpha}}{\Gamma(\alpha+1)\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |y(s) - \overline{y}(s)| \ ds \\ &\leq \overline{M}_0 L_\star |y(t) - \overline{y}(t)| + \frac{\widehat{M}l_b^* b^{\alpha}}{\Gamma(\alpha+1)} ||y - \overline{y}||_n + \frac{\widehat{M}^2 \widetilde{M} \widetilde{M}_1 l_b^* b^{\alpha}}{\Gamma(\alpha+1)\Gamma(\alpha)} ||y - \overline{y}||_n \end{split}$$

Therefore,

$$\|N(y) - N(\overline{y})\|_{n} \leq \left[\overline{M}_{0}L_{\star} + \frac{\widehat{M}l_{b}^{*}b^{\alpha}\left(\Gamma(\alpha) + \widehat{M}\widetilde{M}\widetilde{M}_{1}\right)}{\Gamma(\alpha+1)\Gamma(\alpha)}\right]\|y - \overline{y}\|$$

So, for $\left[\overline{M}_0 L_\star + \frac{\widehat{M} l_b^* b^{\alpha}(\Gamma(\alpha) + \widehat{M} \widetilde{M} \widetilde{M}_1)}{\Gamma(\alpha + 1)\Gamma(\alpha)}\right] < 1$, the operator N is a contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y), \lambda \in (0,1)$. Then the statement (C2) in Theorem 1.10.4 does not hold. The nonlinear alternative of Frigon and Granas shows that

(C1) holds. Thus, we deduce that the operator N has a unique fixed-point y^* which is the unique controllable mild solution of the problem (5.7) - (5.8).

Fractional Partial Problem with Infinite Delay 5.6

Before stating and proving the main result, we give first the definition of mild solution of the semilinear evolution problem (5.9) - (5.10).

Lemma 5.6.1. The system (5.9) - (5.10) is equivalent to the nonlinear integral equation

$$y(t) = \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} C u(s) ds$$

5.5 Frac. Evo. Equa. with Infinite S.D.D.

$$+\frac{1}{\Gamma(\alpha)}\int_{0}^{t} (t-s)^{\alpha-1}f(s,y_{\rho(s,y_s)}) \, ds.$$
(5.33)

In other words, every solution of the integral equation (5.33) is also solution of the system (5.9) - (5.10) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (5.9) - (5.10), and using some classical results from fractional calculus to get (5.33).

Definition 5.6.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (5.9) - (5.10) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0) \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) Cu(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, y_{\rho(s,y_s)}) ds \quad t \in J.$$
(5.34)

Definition 5.6.2. The neutral evolution problem (5.9) - (5.10) is said to be controllable on $[-\infty, b]$ if for every initial function $\phi \in \mathcal{B}$ and $y_1 \in E$ there exists a control $u \in L^2(J; E)$ such that the mild solution $y(\cdot)$ of (5.9) - (5.10) satisfies $y(b) = y_1$.

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}$$

We always assume that $\rho: J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter

(H01) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H02) There exist two functions $p, q \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t,u)| \le p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H03) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$|f(t, u) - f(t, v)| \le l_R(t) ||u - v||$$

for all $u, v \in \mathcal{B}$ with $||u|| \leq R$ and $||v|| \leq R$.

Consider the following space

$$\Omega = \left\{ y : (-\infty, b] \to E : y|_{(-\infty, 0]} \in B \text{ and } y|_{[0, b]} \text{ is continuous } \right\},$$

(H04) The linear operator $W: L^2(J; E) \to E$ is defined by

$$Wu = \int_0^b U(b,s)Cu(s)ds,$$

has an induced invertible operator \widetilde{W}^{-1} which takes values in $L^2(J; E) / \ker W$ and there exists positive constants \widetilde{M} and \widetilde{M}_1 such that

$$||C|| \le \widetilde{M}$$
 and $||\widetilde{W}^{-1}|| \le \widetilde{M}_1$.

Remark 5.6.1. For the construction of W and \tilde{W}^{-1} see the paper by Carmichael and Quinn [38].

Theorem 5.6.1. Suppose that the hypotheses (H01) - (H04) are satisfied and moreover there exists a constant $M_{\star} > 0$ such that

$$\frac{M_{\star}}{\sigma_b + \frac{K_b \widehat{M} q^* [\widehat{M} \widetilde{M} \widetilde{M}_1 + \Gamma(\alpha)]}{\alpha \Gamma(\alpha)^2} M_{\star} b^{\alpha}} > 1$$
(5.35)

with $\sigma_b := c_b + \frac{K_b \widehat{M} \widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M} H \| \phi \|_{\mathcal{B}} \right] + \frac{K_b \widehat{M} p^* b^{\alpha} [\alpha \widehat{M} \widetilde{M}_1 b + \Gamma(\alpha)]}{\alpha \Gamma(\alpha)^2}$. Then the problem (5.9) - (5.10) is controllable on $(-\infty, b]$.

Proof. Transform the problem (5.9) - (5.10) into a fixed-point problem. Consider the operator $N: B_b \to B_b$ defined by :

$$N(y)(t) = \begin{cases} \phi(t), & \text{if } t \in (-\infty, 0]; \\ U(t, 0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ C \ u_y(s) \ ds & (5.36) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, y_{\rho(s, y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$

Using assumption (H03), for arbitrary function $y(\cdot)$, we define the control

$$u_y(t) = \tilde{W}^{-1} \left[y_1 - U(b,0) \ \phi(0) - \frac{1}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \ U(b,s) \ f(s,y_{\rho(s,y_s)}) \ ds \right] (t).$$

Noting that, we have

$$|u_{y}(t)| \leq \|\tilde{W}^{-1}\| \left[|y_{1}| + \|U(t,0)\|_{B(E)} |\phi(0)| + \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-\tau)^{\alpha-1} \|U(b,\tau)\|_{B(E)} |f(\tau,y_{\rho(\tau,y_{\tau})})| d\tau \right]$$

From (H02), we get

$$|u_y(t)| \leq \widetilde{M}_1 \left[|y_1| + \widehat{M}H \|\phi\|_{\mathcal{B}} + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-\tau)^{\alpha-1} \left[p(\tau) + q(\tau) \|y_{\rho(\tau,y_\tau)}\|_{\mathcal{B}} \right] ds \right]$$
(5.37)

Clearly, fixed points of the operator N are mild solutions of the problem (5.9) - (5.10).

For $\phi \in \mathcal{B}$, we will define the function $x(.): (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \in (-\infty, 0]; \\ U(t, 0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in B_b$, set

$$y(t) = z(t) + x(t).$$
 (5.38)

It is obvious that y satisfies (5.34) if and only if z satisfies $z_0 = 0$ and for $t \in J$, we have

$$z(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) C u_{z+x}(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) ds$$

Let

$$B_b^0 = \{ z \in B_b : z_0 = 0 \}.$$

For any $z \in B_b^0$ we have

$$||z||_{b} = \sup\{ |z(t)| : t \in J \} + ||z_{0}||_{\mathcal{B}} = \sup\{ |z(t)| : t \in J \}.$$

Thus $(B_b^0, \|\cdot\|_b)$ is a Banach space.

Define the operators $F,G:B^0_b\to B^0_b$ by :

$$F(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ C \ u_{z+x}(s) \ ds$$
(5.39)

and

$$G(z)(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ f(s, z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}) \ ds \quad \text{for } t \in J.$$
(5.40)

Obviously the operator N has a fixed point is equivalent to F + G has one, so it turns to prove that F + G has a fixed point. The proof will be given in several steps.

Let us first show that the operator F is continuous and compact.

Step 1 : F is continuous.

Let $(z_n)_n$ be a sequence in B_b^0 such that $z_n \to z$ in B_b^0 . Then using (5.37), we get

$$\begin{aligned} |F(z_n)(t) - F(z)(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} ||C|| |u_{z_n}(s) - u_z(s)| ds \\ &\leq \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \widetilde{M}_1 \widehat{M} \times \\ &\quad \times \int_0^b |f(\tau, z_{n\rho(\tau, z_\tau + x_\tau)} + x_{\rho(\tau, z_\tau + x_\tau)}) - f(\tau, z_{\rho(\tau, z_\tau + x_\tau)} + x_{\rho(s, z_\tau + x_\tau)})| d\tau ds \\ &\leq \frac{\widehat{M}^2 \widetilde{M} \widetilde{M}_1 b^{\alpha+1}}{\Gamma(\alpha+1)} \times \\ &\quad \times \int_0^b |f(s, z_{n\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) - f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)})| ds \end{aligned}$$

Since f is L^1 -Carathéodory, we obtain by the Lebesgue dominated convergence theorem

$$|F(z_n)(t) - F(z)(t)| \to 0$$
 as $n \to +\infty$.

Thus F is continuous.

Step 2: *F* maps bounded sets of B_b^0 into bounded sets. For any d > 0, there exists a positive constant ℓ such that for each $z \in B_d = \{z \in B_b^0 : ||z||_n \le d\}$ one has $||F(z)||_n \le \ell$.

Let $z \in B_d$. By (5.37), we have for each $t \in J$

$$\begin{aligned} |F(z)(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|C\| \|u_{z+x}(s)\| ds \\ &\leq \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |u_{z+x}(s)| ds \\ &\leq \frac{\widehat{M}\widetilde{M}}{\Gamma(\alpha)} \int_0^t \widetilde{M}_1 \left[|y_1| + \widehat{M}H\|\phi\|_{\mathcal{B}} \\ &\quad + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \left[p(\tau) + q(\tau) \|z_{\rho(\tau,z_\tau+x_\tau)} + x_{\rho(\tau,z_\tau+x_\tau)} \|d\tau \right] \right] ds \\ &\leq \frac{\widehat{M}\widetilde{M}\widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M}H\|\phi\|_{\mathcal{B}} + \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} \times \\ &\quad \times + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} q(s) \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)} \|ds \right] \end{aligned}$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, b]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_b|x(s)| + (M_b + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\|U(s,0)\|_{B(E)}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\widehat{M}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_b |z(s)| + K_b \widehat{M}H \|\phi\|_{\mathcal{B}} + (M_b + \mathcal{L}^{\phi}) \|\phi\|_{\mathcal{B}} \\ &\leq K_b |z(s)| + (M_b + \mathcal{L}^{\phi} + K_b \widehat{M}H) \|\phi\|_{\mathcal{B}} \end{aligned}$$

Set $C_b := (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}}$ and $\delta_b := K_b d + C_b$. Then

$$\|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \le K_b|z(s)| + C_b \le \delta_b.$$
(5.41)

For each $t \in [0, b]$, it follows that

$$|F(z)(t)| \le \frac{\widehat{M}\widetilde{M}\widetilde{M}_{1}b}{\Gamma(\alpha)} \left[|y_{1}| + \widehat{M}H \|\phi\|_{\mathcal{B}} + \frac{\widehat{M}b^{\alpha}(p^{*} + q^{*}\delta_{b})}{\Gamma(\alpha + 1)} \right] := \ell.$$

Thus there exists a positive number ℓ such that

$$||F(z)||_n \le \ell.$$

Hence $F(B_d) \subset B_d$.

Step 3 : F maps bounded sets into equi-continuous sets of B_b^0 . We consider B_d as in Step 2 and we show that $F(B_d)$ is equi-continuous.

Let $\tau_1, \tau_2 \in J$ with $\tau_2 > \tau_1$ and $z \in B_d$. Then

$$|F(z)(\tau_2) - F(z)(\tau_1)| \le \frac{1}{\Gamma(\alpha)} \int_0^{\tau_1} |(\tau_2 - s)^{\alpha - 1} - (\tau_1 - s)^{\alpha - 1}| ||U(\tau_2, s) - U(\tau_1, s)|| \times ||U(\tau_2, s)|| ||U(\tau_2, s)|| ds + \frac{1}{\Gamma(\alpha)} \int_{\tau_1}^{\tau_2} |(\tau_2 - s)^{\alpha - 1}| ||U(\tau_2, s)||_{B(E)} ||C|| ||u_{z+x}(s)| ds.$$

By the inequalities (5.37) and (5.41) we get

$$|u_{z+x}(t)| \le \widetilde{M}_1 \left[|y_1| + \widehat{M}H \|\phi\|_{\mathcal{B}} + \frac{\widehat{M}b^{\alpha}(p^* + q^*\delta_b)}{\Gamma(\alpha + 1)} \right] := \omega.$$
(5.42)

Then

$$\begin{aligned} |F(z)(\tau_2) - F(z)(\tau_1)| &\leq \frac{\|C\|_{B(E)} \omega}{\Gamma(\alpha)} \int_0^{\tau_1} |(\tau_2 - s)^{\alpha - 1} - (\tau_1 - s)^{\alpha - 1}| \|U(\tau_2, s) - U(\tau_1, s)\| ds \\ &+ \frac{\|C\|_{B(E)} \omega}{\Gamma(\alpha)} \int_{\tau_1}^{\tau_2} |(\tau_2 - s)^{\alpha - 1}| \|U(\tau_2, s)\|_{B(E)} ds. \end{aligned}$$

Noting that $|F(z)(\tau_2) - F(z)(\tau_1)|$ tends to zero as $\tau_2 - \tau_1 \to 0$ independently of $z \in B_d$. The right-hand side of the above inequality tends to zero as $\tau_2 - \tau_1 \to 0$. Since U(t,s) is a strongly continuous operator and the compactness of U(t,s) for t > s implies the continuity in the uniform operator topology (see [7, 90]). As a consequence of Steps 1 to 3 together with the Arzelá-Ascoli theorem it suffices to show that the operator F maps B_d into a precompact set in E.

Let $t \in J$ be fixed and let ϵ be a real number satisfying $0 < \epsilon < t$. For $z \in B_d$ we define

$$F_{\epsilon}(z)(t) = \frac{1}{\Gamma(\alpha)} U(t, t-\epsilon) \int_0^{t-\epsilon} (t-s)^{\alpha-1} U(t-\epsilon, s) C u_{z+x}(s) ds$$

Since U(t,s) is a compact operator, the set $Z_{\epsilon}(t) = \{F_{\epsilon}(z)(t) : z \in B_d\}$ is pre-compact in E for every ϵ sufficiently small, $0 < \epsilon < t$. Moreover using (5.42), we have

$$|F(z)(t) - F_{\epsilon}(z)(t)| \leq \frac{1}{\Gamma(\alpha)} \int_{t-\epsilon}^{t} (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} ||C|| ||u_{z+x}(s)|| ds$$

$$\leq \frac{||C||_{B(E)} \omega}{\Gamma(\alpha)} \int_{t-\epsilon}^{t} (t-s)^{\alpha-1} ||U(t,s)||_{B(E)} ds.$$

Therefore there are precompact sets arbitrary close to the set $\{F(z)(t) : z \in B_d\}$. Hence the set $\{F(z)(t) : z \in B_d\}$ is precompact in E. So we deduce from Steps 1, 2 and 3 that F is a

compact operator.

Step 4 : G is a contraction. Let $z, \overline{z} \in \Omega^0$. By the hypotheses (H01) and (H03), we get for each $t \in [0, b]$

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \times \\ &\times |f(s, z_{\rho(s, z_{s}+x_{s})} + x_{\rho(s, z_{s}+x_{s})}) - f(s, \overline{z}_{\rho(s, z_{s}+x_{s})} + x_{\rho(s, z_{s}+x_{s})})| ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} l_{b}(s) \|z_{\rho(s, z_{s}+x_{s})} + x_{\rho(s, z_{s}+x_{s})} - \overline{z}_{\rho(s, z_{s}+x_{s})} - x_{\rho(s, z_{s}+x_{s})}\|_{\mathcal{B}} ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t-s)^{\alpha-1} l_{b}(s) \|z_{\rho(s, z_{s}+x_{s})} - \overline{z}_{\rho(s, z_{s}+x_{s})}\|_{\mathcal{B}} ds. \end{aligned}$$

Use the inequality (5.41), for get

$$\begin{aligned} |G(z)(t) - G(\overline{z})(t)| &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} K_b l_b(s) |z(s) - \overline{z}(s)| \ ds \\ &\leq \frac{\widehat{M} K_b l_b^* b^\alpha}{\Gamma(\alpha+1)} ||z(t) - \overline{z}(t)|| \end{aligned}$$

Therefore

$$\|G(z) - G(\overline{z})\|_n \le \frac{\widehat{M}K_b l_b^* b^{\alpha}}{\Gamma(\alpha + 1)} \|z(t) - \overline{z}(t)\|.$$

Then the operator G is a contraction.

Step 5: For applying Theorem (1.10.2), we must check (Av2): i.e. it remains to show that the set

$$\Sigma = \left\{ z \in \Omega^0 : z = \lambda F(z) + \lambda G\left(\frac{z}{\lambda}\right) \quad \text{for some } \lambda \in]0, 1[\right\}.$$

is bounded.

Let $z \in \Sigma$. By (H01) - (H02) and (H04), we have for each $t \in [0, b]$

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|C\| \|u_{z+x}(s)\| ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s, z_{\rho\left(s, \frac{z_s}{\lambda} + x_s\right)} + x_{\rho\left(s, \frac{z_s}{\lambda} + x_s\right)}) \right| ds \\ &\leq \frac{\widehat{M}\widetilde{M}\widetilde{M}_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M}H\|\phi\|_{\mathcal{B}} \\ &\quad \times + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \left[p(\tau) + q(\tau) \|z_{\rho(\tau,\cdot)} + x_{\rho(\tau,\cdot)} \|d\tau \right] ds \right] \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho\left(s, \frac{z_s}{\lambda} + x_s\right)} + x_{\rho\left(s, \frac{z_s}{\lambda} + x_s\right)} \|B\right] ds \end{aligned}$$

Lemma 1.7.2 and inequality (5.41)

$$\begin{aligned} \frac{1}{\lambda} |z(t)| &\leq \frac{\widehat{M}MM_1b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M}H \|\phi\|_{\mathcal{B}} \right] \\ &+ \frac{\widehat{M}^2 \widetilde{M}\widetilde{M}_1 b}{\Gamma(\alpha)^2} \int_0^b (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)} \|_{\mathcal{B}} \right] ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,\frac{z_s}{\lambda}+x_s)} + x_{\rho(s,\frac{z_s}{\lambda}+x_s)} \|_{\mathcal{B}} \right] ds \end{aligned}$$

We consider the function $u(t) := \sup_{\theta \in [0,b]} |z(\theta)|$ and use the fact that $0 < \lambda < 1$

$$\frac{K_{b}}{\lambda}u(t) + C_{b} \leq C_{b} + \frac{K_{b}\widehat{M}\widetilde{M}\widetilde{M}_{1}b}{\Gamma(\alpha)} \left[|y_{1}| + \widehat{M}H||\phi||_{\mathcal{B}} \right] + \frac{K_{b}\widehat{M}p^{*}b^{\alpha}[\alpha\widehat{M}\widetilde{M}\widetilde{M}_{1}b + \Gamma(\alpha)]}{\alpha\Gamma(\alpha)^{2}} \\
+ \frac{K_{b}\widehat{M}^{2}\widetilde{M}\widetilde{M}_{1}}{\Gamma(\alpha)^{2}} \int_{0}^{b} (t - s)^{\alpha - 1}q(s) \left(\frac{K_{b}}{\lambda}|z(s)| + C_{b} \right) ds \\
+ \frac{K_{b}\widehat{M}}{\Gamma(\alpha)} \int_{0}^{t} (t - s)^{\alpha - 1}q(s) \left(\frac{K_{b}}{\lambda}|z(s)| + C_{b} \right) ds$$

$$= C + \frac{K_{b}\widehat{M}\widetilde{M}\widetilde{M}_{1}b}{\Gamma(\alpha)} \left[|y_{1}| + \widehat{M}H||\phi||_{c} \right] + \frac{K_{b}\widehat{M}p^{*}b^{\alpha}[\alpha\widehat{M}\widetilde{M}_{1}b + \Gamma(\alpha)]}{\alpha\widetilde{M}\widetilde{M}_{1}b + \Gamma(\alpha)} \quad \text{Then, we have}$$

Set $\sigma_b := C_b + \frac{K_b M M M_1 b}{\Gamma(\alpha)} \left[|y_1| + \widehat{M} H \| \phi \|_{\mathcal{B}} \right] + \frac{K_b M p^* b^{\alpha} [\alpha M M M_1 b + \Gamma(\alpha)]}{\alpha \Gamma(\alpha)^2}$. Then, we have

$$\frac{K_b}{\lambda}u(t) + C_b \leq \sigma_b + \frac{K_b\widehat{M^2}\widetilde{M}\widetilde{M_1}}{\Gamma(\alpha)^2} \int_0^b (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)| + c_b\right) ds + \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)| + C_b\right) ds.$$

We consider the function μ defined by

$$\mu(t) = \sup \{ \frac{K_b}{\lambda} u(s) + C_b : 0 \le s \le t \}, \ 0 \le t \le b$$

Let $t^* \in [0, t]$ such that $\mu(t) = \frac{K_b}{\lambda}u(t^*) + C_b$. From the previous inequality, we have for all $t \in [0, b]$

$$\mu(t) \le \sigma_b + \frac{K_b \widehat{M^2} \widetilde{M} \widetilde{M}_1 q^*}{\Gamma(\alpha)^2} \int_0^b (t-s)^{\alpha-1} \mu(s) ds + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^b (t-s)^{\alpha-1} \mu(s) ds$$

Then, we have

$$\mu(t) \le \sigma_b + \frac{K_b \widehat{M} q^* [\widehat{M} \widetilde{M} \widetilde{M}_1 + \Gamma(\alpha)]}{\Gamma(\alpha)^2} \int_0^b (t-s)^{\alpha-1} \mu(s) ds.$$

Consequently,

$$\frac{\|z\|_n}{\sigma_b + \frac{K_b \widehat{M} q^* [\widehat{M} \widetilde{M} \widetilde{M}_1 + \Gamma(\alpha)]}{\alpha \Gamma(\alpha)^2} \|z\|_n \ b^{\alpha}} \le 1.$$

Then by (5.35), there exists a constant M^* such that $||z||_n \neq M^*$. Set

$$\mathcal{U} = \{ z \in B_b^0 : ||z||_n \le M^* + 1 \}.$$

Clearly, \mathcal{U} is a closed subset of B_b^0 . From the choice of \mathcal{U} there is no $z \in \partial \mathcal{U}$ such that $z = \lambda F(z) + \lambda G\left(\frac{z}{\lambda}\right)$ some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.5 does not hold. As a consequence of the nonlinear alternative of Avramescu ([18]), we deduce that (C1) holds : i.e. the operator F + G as a fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), t \in (-\infty, b]$ is a fixed point of the operator N, which is a mild solution of the problem (5.9) – (5.10). Thus the evolution system (5.9) – (5.10) is controllable on $(-\infty, b]$.

5.7 Fractional Neutral Problem with Infinite Delay

In this section, we give controllability result for the neutral functional fractional differential evolution problem with infinite state-dependent delay (5.11) - (5.12). Firstly, we define its mild solution.

Lemma 5.7.1. The system (5.11) - (5.12) is equivalent to the nonlinear integral equation

$$y(t) = [\phi(0) - g(0, \phi)] + g(t, y_{\rho(t, y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} A(s) y(s) \, ds \qquad (5.43)$$
$$+ \frac{1}{\Gamma(\alpha)} \int_0^t U(t, s) C u(s) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} f(s, y_{\rho(s, y_s)}) \, ds.$$

In other words, every solution of the integral equation (5.43) is also solution of the system
$$(5.11) - (5.12)$$
 and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (5.11)–(5.12), and using some classical results from fractional calculus to get (5.43).

Definition 5.7.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (5.11) - (5.12) if $y(t) = \phi(t)$ for all $t \leq 0$ and y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s,y_{\rho(s,y_s)}) ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) Cu(s) ds \quad for \ each \ t \in J.$$
(5.44)

Definition 5.7.2. The neutral evolution problem (5.11) - (5.12) is said to be controllable on the $(-\infty, b]$ if for every initial function $\phi \in \mathcal{B}$ and $y_1 \in E$ there exists a control $u \in L^2(J; E)$ such that the mild solution $y(\cdot)$ of (5.11) - (5.12) satisfies $y(b) = y_1$.

We consider the hypotheses (H_{ϕ}) , (H01) - (H03) and we need the following assumptions (H05) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\phi) - A(\overline{s})g(\overline{s},\overline{\phi})| \le L_{\star} (|s - \overline{s}| + ||\phi - \overline{\phi}||)$$

for all $\phi, \overline{\phi} \in \mathcal{B}$.

(H06) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H07) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

$$|A(t) g(t,\phi)| \leq L (||\phi||_{\mathcal{B}} + 1)$$
 for all $t \in J$ and $\phi \in \mathcal{B}$.

(H08) The function g is completely continuous and for any bounded set $Q \subseteq \mathcal{B}$ the set $\{t \rightarrow g(t, x_t) : x \in Q\}$ is equi-continuous in C(J; E).

Theorem 5.7.1. Suppose that the hypotheses (H01) - (H08) are satisfied and moreover there exists a constant $M^{\star\star} > 0$ with

$$\frac{M^{\star\star}}{\delta_b + \frac{K_b \widehat{M}(\widehat{M}\widetilde{M}\widetilde{M}_1 b^{\alpha} + \Gamma(\alpha + 1))}{(1 - \overline{M}_0 L K_b) \Gamma^2(\alpha + 1)} \quad [q^* b^{\alpha} M^{\star\star}]} > 1,$$
(5.45)

where $\delta_b := C_b + \frac{K_b \sigma_b}{(1 - \overline{M}_0 L K_b)}$. and

$$\begin{split} \sigma_b &= K_b \overline{M}_0 L (C_b + 1 + \widehat{M} (\|\phi\|_{\mathcal{B}} + 1)) \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha + 1)} \left[|y_1| + \widehat{M} \left(H + \overline{M}_0 L \right) \|\phi\|_{\mathcal{B}} + \left(\widehat{M} + 1 \right) \overline{M}_0 L \right] \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha} \overline{M}_0 L}{\Gamma(\alpha + 1)} (K_b |y_1| + (M_b + L^{\phi}) \|\phi\|_{\mathcal{B}}) \\ &+ \frac{K_b \widehat{M} p^* b^{\alpha} [\Gamma(\alpha + 1) + \widehat{M} \widetilde{M} \widetilde{M}_1 \widetilde{M}_1 b^{\alpha}]}{\Gamma^2(\alpha + 1)}. \end{split}$$

Then the neutral evolution problem (5.11) - (5.12) is controllable on $(-\infty, b]$.

Proof. Consider the operator $\widetilde{N}: B_b \to B_b$ defined by :

$$\widetilde{N}(y)(t) = \begin{cases} \phi(t), & \text{if } t \in (-\infty, 0]; \\ U(t, 0) \ [\phi(0) - g(0, \phi)] + g(t, y_{\rho(t, y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ C \ u_y(s) \ ds \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, y_{\rho(s, y_s)}) \ ds, & \text{if } t \in J. \end{cases}$$
(5.46)

Using assumption (H3), for arbitrary function $y(\cdot)$, we define the control

$$u_{y}(t) = \tilde{W}^{-1} \left[y_{1} - U(b,0) \left(\phi(0) - g(0,\phi) \right) - g(b,y_{\rho(b,y_{b})}) - \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} U(b,s) f(s,y_{\rho(s,y_{s})}) ds \right] (t).$$

Noting that

$$\begin{aligned} |u_{y}(t)| &\leq \|\widetilde{W}^{-1}\| \left[\|y_{1}\| + \|U(t,0)\|_{B(E)} \left(|\phi(0)| + \|A^{-1}(0)\| |A(0)g(0,\phi)| \right) \\ &+ \|A^{-1}(b)\| |A(b)g(b,y_{\rho(b,y_{b})})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} U(b,\tau) f(\tau,y_{\rho(\tau,y_{\tau})}) d\tau \right] \\ &\leq \widetilde{M}_{1} \left[\|y_{1}\| + \widehat{M}H\| \phi \|_{\mathcal{B}} + \widehat{M}M_{0}L(\|\phi\|_{\mathcal{B}} + 1) + \overline{M}_{0}L(\|y_{\rho(b,y_{b})}\|_{\mathcal{B}} + 1) \right] \\ &+ \frac{\widetilde{M}_{1}\widehat{M}}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \left[p(\tau) + q(\tau) \|y_{\rho(\tau,y_{\tau})}\|_{\mathcal{B}} \right]. \end{aligned}$$

we get

$$\begin{aligned} |u_{y}(t)| &\leq \widetilde{M}_{1} \left[|y_{1}| + \widehat{M} \left(H + \overline{M}_{0}L \right) \|\phi\|_{\mathcal{B}} + \left(\widehat{M} + 1 \right) \overline{M}_{0}L \right] \\ &+ \widetilde{M}_{1} \overline{M}_{0}L \|y_{\rho(b,y_{b})}\|_{\mathcal{B}} + \frac{\widehat{M} \widetilde{M}_{1} p^{*} b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M} \widetilde{M}_{1} q^{*}}{\Gamma(\alpha)} \int_{0}^{b} (t-s)^{\alpha-1} \|y_{\rho(s,y_{s})}\|_{\mathcal{B}} ds \end{aligned}$$

$$(5.47)$$

We shall show that using this control the operator \widetilde{N} has a fixed point $y(\cdot)$, which is a mild solution of the neutral evolution system (5.11) - (5.12).

For $\phi \in \mathcal{B}$, we will define the function $x(.): (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \in (-\infty, 0]; \\ U(t, 0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in B_b$, set

$$y(t) = z(t) + x(t).$$
 (5.48)

It is obvious that y satisfies (5.44) if and only if z satisfies $z_0 = 0$ and for $t \in J$, we get

$$z(t) = g(t, z_t + x_t) - U(t, 0)g(0, \phi) + \int_0^t U(t, s)A(s)g(s, z_s + x_s)ds + \int_0^t U(t, s)Cu_z(s)ds + \int_0^t U(t, s)f(s, z_s + x_s)ds.$$

Define the operators $\widetilde{F}, \widetilde{G}: B^0_b \to B^0_b$ by :

$$\begin{split} \widetilde{F}(z)(t) &= g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)}) - U(t, 0)g(0, \phi) \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ C \ u_{z + x}(s) \ ds \\ \widetilde{G}(z)(t) &= \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) \ f(s, z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}) \ ds. \end{split}$$

Obviously the operator \widetilde{N} has a fixed point is equivalent to $\widetilde{F} + \widetilde{G}$ has one, so it turns to prove that $\widetilde{F} + \widetilde{G}$ has a fixed point. We can show as in Section 3 that the operator \widetilde{F} is

continuous and compact and \tilde{G} is a contraction. To apply Theorem 1.10.5, we must check (C2), i.e., it remains to show that the set

$$\Sigma = \left\{ z \in \Omega^0 : z = \lambda \widetilde{F}(z) + \lambda \widetilde{G}\left(\frac{z}{\lambda}\right) \quad \text{for some } \lambda \in]0,1[\right\}.$$

is bounded.

Let $z \in \Sigma$. By (H01) to (H05) and (5.47), we have for each $t \in J$

$$\begin{split} \frac{1}{\lambda} |z(t)| &\leq \|A^{-1}(t)\| \|A(t)g(t, z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)})| + \|U(t, 0)\|_{B(E)} \|A^{-1}(0)\| \|A(0)g(0, \phi)| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \|C\| \|u_{z + x}(s)\| ds \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \left| f(s, z_{\rho(s, \frac{z_s}{\lambda} + x_s)} + x_{\rho(s, \frac{z_s}{\lambda} + x_s)}) \right| ds \\ &\leq \overline{M}_0 L \left(\|z_{\rho(t, z_t + x_t)} + x_{\rho(t, z_t + x_t)} \|B + 1 \right) + \widehat{MM}_0 L \left(\|\phi\|_B + 1 \right) \\ &+ \frac{\widehat{M}\widetilde{M}\widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha + 1)} \left[\|y_1\| + \widehat{M} \left(H + \overline{M}_0 L \right) \|\phi\|_B + \left(\widehat{M} + 1 \right) \overline{M}_0 L \right] \\ &+ \frac{\widehat{M}\widetilde{M}\widetilde{M}_1 b^{\alpha} \overline{M}_0 L}{\Gamma(\alpha + 1)} \|z_{\rho(b, y_b)} + x_{\rho(b, y_b)}\|_B + \frac{\widehat{M}p^* b^{\alpha}[\Gamma(\alpha + 1) + \widehat{M}\widetilde{M}\widetilde{M}_1 \widetilde{M}_1 b^{\alpha}]}{\Gamma^2(\alpha + 1)} \\ &+ \frac{\widehat{M}^2 \widetilde{M}\widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha) \Gamma(\alpha + 1)} \int_0^b (t - s)^{\alpha - 1} q(s) \|z_{\rho(s, z_s + x_s)} + x_{\rho(s, z_s + x_s)}\|_B ds \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) \|z_{\rho(s, \frac{z_s}{\lambda} + x_s)} + x_{\rho(s, \frac{z_s}{\lambda} + x_s)} \|g ds \end{split}$$

Noting that we have $||z_{\rho(b,y_b)} + x_{\rho(b,y_b)}||_{\mathcal{B}} \leq K_b |y_1| + (M_b + L^{\phi}) ||\phi||_{\mathcal{B}}$ and using the first inequality $|||z_{\rho(t,y_t)} + x_{\rho(t,y_t)}||_{\mathcal{B}} \leq K_b |z(t)| + C_b$ we obtain

$$\begin{aligned} \frac{K_b}{\lambda} |z(t)| &\leq K_b \overline{M}_0 L(K_b | z(t)| + C_b + 1) + K_b \widehat{M} \overline{M}_0 L(\|\phi\|_{\mathcal{B}} + 1) \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha + 1)} \left[|y_1| + \widehat{M} \left(H + \overline{M}_0 L \right) \|\phi\|_{\mathcal{B}} + \left(\widehat{M} + 1 \right) \overline{M}_0 L \right] \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha} \overline{M}_0 L}{\Gamma(\alpha + 1)} (K_b | y_1| + (M_b + L^{\phi}) \|\phi\|_{\mathcal{B}}) \\ &+ \frac{K_b \widehat{M} p^* b^{\alpha} [\Gamma(\alpha + 1) + \widehat{M} \widetilde{M} \widetilde{M}_1 \widetilde{M}_1 b^{\alpha}]}{\Gamma^2(\alpha + 1)} \\ &+ \frac{K_b \widehat{M}^2 \widetilde{M} \widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha) \Gamma(\alpha + 1)} \int_0^b (t - s)^{\alpha - 1} q(s) (K_b | z(s)| + C_b) ds \\ &+ \frac{K_b \widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) (\frac{K_b}{\lambda} | z(s)| + C_b) ds \end{aligned}$$

We consider the function $u(t) := \sup_{\theta \in [0,b]} |z(\theta)|$ and use the fact that $0 < \lambda < 1$ Then, we have

$$\frac{K_b}{\lambda}u(t) \leq \sigma_b + K_b\overline{M}_0L\frac{|u(t)|}{\lambda} + \frac{K_b\widehat{M}^2\widetilde{M}\widetilde{M}_1b^{\alpha}}{\Gamma(\alpha)\Gamma(\alpha+1)} \int_0^b (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)|+c_b\right) ds + \frac{K_b\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1}q(s)\left(\frac{K_b}{\lambda}|z(s)|+C_b\right) ds.$$

Set:

$$\begin{split} \sigma_b &= K_b \overline{M}_0 L(C_b + 1 + \widehat{M}(\|\phi\|_{\mathcal{B}} + 1)) \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha + 1)} \left[\|y_1\| + \widehat{M} \left(H + \overline{M}_0 L \right) \|\phi\|_{\mathcal{B}} + \left(\widehat{M} + 1 \right) \overline{M}_0 L \right] \\ &+ \frac{K_b \widehat{M} \widetilde{M} \widetilde{M}_1 b^{\alpha} \overline{M}_0 L}{\Gamma(\alpha + 1)} (K_b |y_1| + (M_b + L^{\phi}) \|\phi\|_{\mathcal{B}}) + \frac{K_b \widehat{M} p^* b^{\alpha} [\Gamma(\alpha + 1) + \widehat{M} \widetilde{M} \widetilde{M}_1 \widetilde{M}_1 b^{\alpha}]}{\Gamma^2(\alpha + 1)} \end{split}$$

Then

$$\frac{K_b}{\lambda} (1 - \overline{M}_0 L) u(t) \leq K_b \sigma_b + \frac{K_b \widehat{M}^2 \widetilde{M} \widetilde{M}_1 b^{\alpha}}{\Gamma(\alpha) \Gamma(\alpha + 1)} \int_0^b (t - s)^{\alpha - 1} q(s) (K_b u(s) + C_b) ds \\
+ \frac{K_b \widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s) (\frac{K_b}{\lambda} u(s) + C_b) ds$$

with $\delta_b := C_b + \frac{K_b \sigma_b}{(1 - \overline{M}_0 L)}$, thus

$$\frac{K_b}{\lambda}u(t) + C_b \leq \delta_b + \frac{K_b\widehat{M}^2\widetilde{M}\widetilde{M}_1b^{\alpha}}{(1 - \overline{M}_0L)\Gamma(\alpha)\Gamma(\alpha + 1)} \int_0^b (t - s)^{\alpha - 1} q(s)(K_bu(s) + C_b)ds \\
+ \frac{K_b\widehat{M}}{(1 - \overline{M}_0L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} q(s)(\frac{K_b}{\lambda}u(s) + C_b)ds$$

We consider the function μ defined by

$$\mu(t) := \sup \{ \frac{K_b}{\lambda} | u(s) | + C_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [0, t]$ be such that $\mu(t) = \frac{K_b}{\lambda} |u(t^*)| + C_b$. If $t \in J$, by the previous inequality, we have for $t \in J$

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M}(\widehat{M}\widetilde{M}\widetilde{M}_1 b^{\alpha} + \Gamma(\alpha + 1))}{(1 - \overline{M}_0 L)\Gamma(\alpha)\Gamma(\alpha + 1)} \int_0^b (t - s)^{\alpha - 1} q(s)\mu(s)ds$$

Consequently,

$$\frac{\|z\|_b}{\delta_b + \frac{K_b \widehat{M}(\widehat{M}\widetilde{M}\widetilde{M}_1 b^{\alpha} + \Gamma(\alpha + 1))}{(1 - \overline{M}_0 L)\Gamma^2(\alpha + 1)} [q^* b^{\alpha} \|z\|_b]} \leq 1.$$

Then by (5.45), there exists a constant $M^{\star\star}$ such that $||z||_b \neq M^{\star\star}$. Set

$$\tilde{\mathcal{U}} = \{ z \in B_b^0 : ||z||_b \le M^{\star \star} + 1 \}.$$

Clearly, $\widetilde{\mathcal{U}}$ is a closed subset of B_b^0 . From the choice of $\widetilde{\mathcal{U}}$ there is no $z \in \partial \widetilde{\mathcal{U}}$ such that $z = \lambda \ \widetilde{F}(z)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.5 does not hold. As a consequence of the nonlinear alternative of Avramescu ([18]), we deduce that (C1) holds : i.e. the operator \widetilde{F} has a fixed-point z^* . Then $y^*(t) = z^*(t) + x(t), t \in (-\infty, b]$ is a fixed point of the operator \widetilde{N} , which is a mild solution of the problem (5.11) – (5.12). Thus the fractional neutral evolution system (5.11) – (5.12) is controllable on $(-\infty, b]$.

5.8 Examples

To illustrate the previous results, we give in this section six examples.

Example 1. Consider the partial differential equation

$$\begin{cases} \frac{\partial z}{\partial t}(t,\xi) = \frac{\partial^2 z(t,\xi)}{\partial \xi^2} + d(\xi)u(t) + a_0(t,\xi)z(t,\xi) \\ + \int_{-\infty}^0 a_1(s-t)z \left[s - \rho_1(t)\rho_2 \left(\int_0^{\pi} a_2(\theta)|z(t,\theta)|^2 d\theta \right), \xi \right] ds, \\ for \ t \ge 0, \ \xi \in [0,\pi], \end{cases}$$
(5.49)
$$z(t,0) = \ z(t,\pi) = 0, \qquad \text{for } t \ge 0, \\ z(\theta,\xi) = \ z_0(\theta,\xi), \qquad \text{for } -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$

where $a : \mathbb{R}^+ \times [0,\pi] \to \mathbb{R}$ is a continuous function and is uniformly Hölder continuous in $t ; a_0 : \mathbb{R}^+ \times [0,\pi] \to \mathbb{R} ; a_1 : \mathbb{R}^- \to \mathbb{R} ; a_2 : [0,\pi] \to \mathbb{R} ; \rho_i : \mathbb{R}^+ \to \mathbb{R}$ for i = 1,2; $z_0 : \mathbb{R}_- \times [0,\pi] \to \mathbb{R}$ and $d : \mathbb{R}_+ \to E$ are continuous functions. $u(\cdot) : \mathbb{R}_+ \to E$ is a given control.

To study this system, we consider the space $E = L^2([0, \pi], \mathbb{R})$ and the operator $A : D(A) \subset E \to E$ given by Aw = w'' with $D(A) := \{ w \in E : w'' \in E, w(0) = w(\pi) = 0 \}$. It is well known that A is the infinitesimal generator of an analytic semigroup $\{T(t)\}_{t\geq 0}$ on E. Furthermore, A has discrete spectrum with eigenvalues $-n^2$, $n \in \mathbb{N}$, and corresponding normalized eigenfunctions given by $z_n(\xi) = \frac{\sin(n\xi)\sqrt{2}}{\sqrt{\pi}}$. In addition, $\{z_n : n \in \mathbb{N}\}$ is an orthonormal basis of E and $T(t)x = \sum_{n=1}^{+\infty} e^{-n^2t}(x, z_n)z_n$ for $x \in E$ and $t \geq 0$. It follows from this representation that T(t) is compact for every t > 0 and that $||T(t)|| \leq e^{-t}$ for every $t \geq 0$. On the domain D(A), we define the operators $A(t) : D(A) \subset E \to E$ by

$$A(t)x(\xi) = Ax(\xi) + a_0(t,\xi)x(\xi)$$

By assuming that $a_0(.,.)$ is continuous and that $a_0(t,\xi) \leq -\delta_0$ ($\delta_0 > 0$) for every $t \in \mathbb{R}, \xi \in [0,\pi]$, it follows that the system

$$u'(t) = A(t)u(t) \quad t \ge s; \quad u(s) = x \in E,$$

has an associated evolution family given by $U(t,s)x(\xi) = \left[T(t-s)exp\left(\int_s^t a_0(\tau,\xi)d\tau\right)x\right](\xi).$

From this expression, it follows that U(t,s) is a compact linear operator and that

$$||U(t,s)|| \le e^{-(1+\delta_0)(t-s)} \text{ for every } (t,s) \in \Delta.$$

Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ the space of bounded uniformly continuous functions defined from \mathbb{R}_{-} to E endowed with the uniform norm $\|\phi\| = \sup_{\theta \in \mathbb{R}_{-}} |\phi(\theta)|$.

Theorem 5.8.1. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds and the functions $d : \mathbb{R}_+ \to E$, $\rho_i : \mathbb{R}_+ \to \mathbb{R}_+, i = 1, 2, a_1 : \mathbb{R}^- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then the evolution system (5.49) is controllable on $(-\infty, +\infty)$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions and let $C \in L(\mathbb{R}; E)$ be defined as : $Cu(t)(\xi) = d(\xi)u(t), t \ge 0, \xi \ge 0, u \in \mathbb{R}, d(\xi) \in E,$

which permit to transform system (5.49) into the abstract system (5.1) - (5.2). Moreover, the function f is a bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem (1.10.2). Thus, the conclusion of our theorem hold.

From Remark (1.7.2), we have the following result.

Corollary 5.8.1. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then the evolution problem (5.49) is controllable on \mathbb{R} .

Example 2. Consider the partial differential equation

$$\begin{cases} \frac{\partial}{\partial t} \left[u(t,\xi) - \int_{-\infty}^{0} a_3(s-t)u\left(s - \rho_1(t)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|u(t,\theta)|^2 d\theta\right), \xi\right) ds \right] \\ = \frac{\partial^2 u(t,\xi)}{\partial \xi^2} + a_0(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0} a_1(s-t)u\left(s - \rho_1(t)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|u(t,\theta)|^2 d\theta\right), \xi\right) ds, \\ for \ t \ge 0, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad \text{for } t \ge 0, \\ v(\theta,\xi) = v_0(\theta,\xi), \qquad \text{for } -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(5.50)

where $a_3 : \mathbb{R}^- \to \mathbb{R}$ is a continuous function and a, a_i for $i = 0, 1, 2, \rho_i$ for $i = 1, 2, z_0, d$ and $u(\cdot)$ are defined as in (5.49).

Theorem 5.8.2. Let $\mathcal{B} = BUC(\mathbb{R}_{-}; E)$ and $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds and the functions $d : \mathbb{R}_{+} \to E$, $\rho_{i} : \mathbb{R}_{+} \to \mathbb{R}_{+}, i = 1, 2, a_{1}, a_{3} : \mathbb{R}^{-} \to \mathbb{R}$ and $a_{2} : [0, \pi] \to \mathbb{R}$ are continuous. Then the evolution system (5.50) is controllable on $(-\infty, +\infty)$. **Proof.** From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$
$$Cu(t)(\xi) = d(\xi)u(t), \ t \in [0,b], \ \xi \in [0,\pi], \ u \in \mathbb{R}, \ d(\xi) \in E$$

which permit to transform system (5.50) into the abstract system (5.3) - (5.4). Moreover, the function f is a bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem (5.3.1). Thus, the conclusion of our theorem hold.

From Remark (1.7.2), we have the following result.

Corollary 5.8.2. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a unique mild solution of (5.50) on \mathbb{R} .

Example 3. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + d(\xi)u(t) + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0}a_{1}(s-t)u\left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(5.51)

where $a_1: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 5.8.3. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, $\rho_i : [0, b] \to \mathbb{R}$ for i = 1, 2 are continuous and the functions $a_1 : [-r, 0] \to \mathbb{R}$; $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then the problem (5.51) is controllable on [-r, b].

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

Finally let $C \in B(\mathbb{R}; E)$ be defined as

$$Cu(t)(\xi) = d(\xi)u(t), \ t \in [0, b], \ \xi \in [0, \pi], \ u \in \mathbb{R}, \ d(\xi) \in E.$$

are well defined functions, which permit to transform system (5.51) into the abstract system (5.5) - (5.6). Moreover, the function f is bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem 5.4.1. From Remark 1.7.1, we have the following result.

Corollary 5.8.3. Let $\varphi \in C(H; E)$ be continuous and bounded. Then the problem (5.51) is controllable on [-r, b].

Example 4. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-r}^{0} a_{3}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + d(\xi)u(t) + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0} a_{1}(s-t)u \left(s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta \right), \xi \right) ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -r < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(5.52)

where $a_3: [-r, 0] \to \mathbb{R}$ is a continuous function

Theorem 5.8.4. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, $\rho_i : [0, b] \to \mathbb{R}$ for i = 1, 2 are continuous and the functions $a_1, a_3 : [-r, 0] \to \mathbb{R}$; $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then the problem (5.52) is controllable on [-r, b].

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$

$$g(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

$$Cu(t)(\xi) = d(\xi)u(t), \ t \in [0,b], \ \xi \in [0,\pi], \ u \in \mathbb{R}, \ d(\xi) \in E.$$

are well defined functions, which permit to transform system (5.52) into the abstract system (5.7) - (5.8). Moreover, the function f is bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem 5.5.1. From Remark 1.7.1, we have the following result.

Corollary 5.8.4. Let $\varphi \in C(H; E)$ be continuous and bounded. Then the problem (5.52) is conntrollable on [-r, b].

Example 5. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + d(\xi)u(t) + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0} a_{1}(s-t)u \left[s - \rho_{1}(t)\rho_{2} \left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right), \xi\right] ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(5.53)

where $a_1: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 5.8.5. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_i : [0,b] \to \mathbb{R}$ for i = 1, 2 are continuous and the functions $a_1 : (-\infty, 0] \to \mathbb{R}$; $a_2 : [0,\pi] \to \mathbb{R}$ are continuous. Then the problem (5.53) is controllable on [-r, b].

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

Finally let $C \in B(\mathbb{R}; E)$ be defined as

$$Cu(t)(\xi) = d(\xi)u(t), \ t \in [0, b], \ \xi \in [0, \pi], \ u \in \mathbb{R}, \ d(\xi) \in E.$$

are well defined functions, which permit to transform system (5.53) into the abstract system (5.9) - (5.10). Moreover, the function f is bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem 5.6.1. From Remark 1.7.2, we have the following result.

Corollary 5.8.5. Let $\phi \in \mathcal{B}$ be continuous and bounded. The problem (5.53) is controllable on $(-\infty, b]$.

Example 6. Consider the partial differential equation

$$\begin{cases} {}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-\infty}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\ = \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + d(\xi)u(t) + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(5.54)

where $a_3: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 5.8.6. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_i : [0, b] \to \mathbb{R}$ for i = 1, 2 are continuous and the functions $a_1, a_3 : \mathbb{R}_- \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then the problem (5.54) is controllable on $(-\infty, b]$.

Proof. From the assumptions, we have that

$$f(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds$$

$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

$$Cu(t)(\xi) = d(\xi)u(t), \ t \in [0,b], \ \xi \in [0,\pi], \ u \in \mathbb{R}, \ d(\xi) \in E$$

are well defined functions, which permit to transform system (5.54) into the abstract system (5.11) - (5.12). Moreover, the function f is bounded linear operator. Now, the controllability of mild solutions can be deduced from a direct application of Theorem 5.7.1. From Remark 1.7.2, we have the following result.

Corollary 5.8.6. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then the problem (5.54) is controllable on $(-\infty, b]$.

Chapter 6

Fractional Evolution Inclusions with State-Dependent Delay

6.1 Introduction

In this chapter, we will give sufficient conditions for the existence of mild solutions for some classes of Caputo's fractional derivative order of partial functional and neutral functional differential evolution inclusions with finite and infinite state-dependent delay¹.

Using the alternative of Frigon for multivalued contraction maps in Banach space (see [50]), combined with the semi-group theory.

The existence of the mild solutions is demonstrated in section 6.2 for the following class of fractional evolution inclusion with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) \in A(t)y(t) + F(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \ t \in J = [0, b]$$
(6.1)

$$y(t) = \varphi(t), \quad t \in H = [-r, 0],$$
 (6.2)

where $0 < r < +\infty$, ${}^{c}D_{0}^{\alpha}$ is the standard Caputo's fractional derivative of order $\alpha \in (0, 1)$, $F: J \times C(H; E) \to \mathcal{P}(E)$ is a multivalued map with nonempty compact values, $\mathcal{P}(E)$ is the family of all subsets of E, $\rho: J \times C(H; E) \to \mathbb{R}$ and $\varphi \in C(H; E)$ are given functions and $\{A(t)\}_{t\geq 0}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 6.3, we consider the following class of fractional neutral evolution inclusion with finite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t)[y(t) - g(t, y_{\rho(t, y_{t})})] \in A(t)y(t) + F(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(6.3)

$$y(t) = \varphi(t), \quad t \in H, \tag{6.4}$$

where $A(\cdot)$, F and φ are as in problem (6.1) – (6.2) and $g: J \times C(H; E) \to E$ is a given function.

In section 6.4, we study the following class of fractional evolution inclusion with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t) \in A(t)y(t) + F(t, y_{\rho(t,y_{t})}), \quad \text{a.e. } t \in J$$

(6.5)

¹[15] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Multivalued fractional partial functional and neutral functional fractional evolution equations with state-dependent delay, submitted.

$$y_0 = \phi \in \mathcal{B},\tag{6.6}$$

where \mathcal{B} is an abstract *phase space* to be specified later, $F: J \times \mathcal{B} \to E, \rho: J \times \mathcal{B} \to \mathbb{R}$ and $\phi \in \mathcal{B}$ are given functions and $\{A(t)\}_{t \in J}$ is a family of linear closed (not necessarily bounded) operators from E into E that generates an evolution system of operators $\{U(t,s)\}_{(t,s)\in J\times J}$ for $s \leq t$.

An extension of this problem is given in section 6.5, we consider the following class of fractional neutral evolution inclusion with infinite state-dependent delay

$${}^{c}D_{0}^{\alpha}y(t)[y(t) - g(t, y_{\rho(t, y_{t})})] \in A(t)y(t) + F(t, y_{\rho(t, y_{t})}), \quad \text{a.e.} \quad t \in J$$
(6.7)

$$y_0 = \phi \in \mathcal{B},\tag{6.8}$$

where $A(\cdot)$, F and φ are as in problem (6.5) – (6.6) and $g: J \times \mathcal{B} \to E$ is a given function. Finally, section 6.6 is devoted to examples illustrating the abstract theory considered in previous sections.

6.2 Partial Multivalued Problem with Finite Delay

Before stating and proving the main result, we give first the definition of mild solution of the semilinear evolution problem (6.1) - (6.2).

Lemma 6.2.1. The system (6.1) - (6.2) is equivalent to the nonlinear integral equation

$$y(t) = \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \ ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s) \ ds.$$
(6.9)

In other words, every solution of the integral equation (6.9) is also solution of the system (6.1) - (6.2) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (6.1) - (6.2), and using some classical results from fractional calculus to get (6.9).

Definition 6.2.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of (6.1) - (6.2)if $y(t) = \varphi(t)$ for all $t \in [-r, 0]$ and y satisfies the following integral equation

Definition 6.2.2. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of the evolution system (6.1) - (6.2) if $y(t) = \varphi(t)$ for all $t \in H$ and the restriction of $y(\cdot)$ to the interval J is continuous and there exists $f(\cdot) \in L^1(J; E) : f(t) \in F(t, y_{\rho(t,y_t)})$ a.e. in J such that y satisfies the following integral equation :

$$y(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds \quad \text{for each } t \in J.$$
(6.10)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\varphi) : (s,\varphi) \in J \times C(H;E), \rho(s,\varphi) \le 0\}.$$

We always assume that $\rho: J \times C(H; E) \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:
(H_{φ}) The function $t \to \varphi_t$ is continuous from $\mathcal{R}(\rho^-)$ into C(H; E) and there exists a continuous and bounded function $\mathcal{L}^{\varphi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\varphi_t\| \leq \mathcal{L}^{\varphi}(t) \|\varphi\|$$
 for every $t \in \mathcal{R}(\rho^-)$.

We need to introduce the following hypothesis

(H1) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H2) The multifunction $F : J \times C(H; E) \longrightarrow \mathcal{P}(E)$ is L^1_{loc} -Carathéodory with compact and convex values for each $u \in C(H; E)$ and there exist a function $p \in L^1_{loc}(J; \mathbb{R}_+)$ and such that

 $||F(t,u)||_{\mathcal{P}(E)} \le p(t) + q(t) ||u||_B \text{ for a.e. } t \in J \text{ and each } u \in C(H; E).$

(H3) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$H_d(F(t, u) - F(t, v)) \le l_R(t) ||u - v||$$

for each $t \in J$ and for all $u, v \in C(H; E)$ with $||u|| \leq R$ and $||v|| \leq R$ and

$$d(0, F(t, 0)) \le l_R(t) \quad \text{a.e.} \quad t \in J.$$

Theorem 6.2.1. Assume that the hypotheses (H_{φ}) and (H1) - (H3) hold and moreover

$$\frac{\widehat{M}l_b^*b^{\alpha}}{\Gamma(1+\alpha)} < 1 \tag{6.11}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (6.1) - (6.2) has at least mild solution on [-r, b].

Proof. Transform the problem (6.1) - (6.2) into a fixed-point problem. Consider the multivalued operator $N: C([-r,b]; E) \to \mathcal{P}(C([-r,b]; E))$ defined by :

$$N(y) = \begin{cases} h \in C([-r,b];E) : h(t) = \begin{cases} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ \varphi(0) \\ +\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds, & \text{if } t \in J. \end{cases} \end{cases}$$

where $f \in S_{F,y} = \{ v \in L^1(J; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in J \}.$

Clearly, fixed points of the operator N are mild solutions of the problem (6.1) - (6.2). We remark also that, for each $y \in C([-r, b]; E)$, the set $S_{F,y}$ is nonempty since, by (H2), F has a measurable selection (see [40], Theorem III.6).

Let y be a possible fixed point of the operator N. Given $t \leq b$, then y should be solution of the inclusion $y \in \lambda$ N(y) for some $\lambda \in (0, 1)$ and there exists $f \in S_{F,y} \Leftrightarrow f(t) \in F(t, y_{\rho(t,y_t)})$ such that, for each $t \in J$. So, from (H1), (H2), (H $_{\varphi}$) and Lemma 5.2.1, we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq \|U(t,0)\|_{B(E)} |\varphi(0)| + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} |f(s)| \ ds \\ &\leq \widehat{M} \|\varphi\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|y_{\rho(s,y_s)}\| \right] ds \end{aligned}$$

It follows that

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \le \widehat{M} \|\varphi\| + \mathcal{L}^{\varphi} \|\varphi\| + \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[|y(s) + \mathcal{L}^{\varphi} \|\varphi\| \right] ds$$

Set

$$\delta_b := (\widehat{M} + \mathcal{L}^{\varphi}) \|\varphi\| + \frac{\widehat{M}p^* b^{\alpha}}{\Gamma(\alpha + 1)}$$

We consider the function μ defined by

$$\mu(t) := \sup \{ \|y(s)\| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds \quad \text{for } t \in [0,b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}q^*b^{\alpha}}{\Gamma(\alpha+1)}\right]$$
$$= \Lambda_b$$

Since $||y|| \leq \mu(t)$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||y|| \leq \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

Set $Y = \{ y \in C([-r, b]; E) : \sup\{|y(t)| : 0 \le t \le b\} \le \Theta_b + 1\}$. Clearly, Y is a closed subset of C([-r, b]); E).

We shall show that $N: \overline{Y} \to \mathcal{P}(C([-r, b]; E))$ is a contraction and an admissible operator. First, we prove that N is a contraction; Let $y, \overline{y} \in C([-r, b]; E)$ and $h \in N(y)$. Then there exists $f(t) \in F(t, y_{\rho(t,y_t)})$ such that for each $t \in [0, b]$

$$h(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \ f(s) \ ds.$$

From (H3) it follows that

$$H_d(F(t, y_{\rho(t, y_t)}), F(t, \overline{y}_{\rho(t, y_t)})) \le l_b(t) \|y_{\rho(t, y_t)} - \overline{y}_{\rho(t, y_t)}\|.$$

Hence, there is $\rho \in F(t, \overline{y}_{\rho(t,y_t)})$ such that

$$|f(t) - \varrho| \le l_b(t) ||y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)}||, t \in [0, b].$$

Consider $\mathcal{U}_{\star}: [0,b] \to \mathcal{P}(E)$, given by

$$\mathcal{U}_{\star} = \{ \rho \in E : |f(t) - \rho| \le l_b(t) \| y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)} \| \}.$$

Since the multivalued operator $\mathcal{V}(t) = \mathcal{U}_{\star}(t) \cap F(t, \overline{y}_{\rho(t,y_t)})$ is measurable (in [40], see Proposition III.4), there exists a function $\overline{f}(t)$, which is a measurable selection for \mathcal{V} . So, $\overline{f}(t) \in F(t, \overline{y}_{\rho(t,y_t)})$ and we obtain for each $t \in [0, b]$

$$|f(t) - \overline{f}(t)| \le l_b(t) ||y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)}||$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{aligned} |f(t) - \overline{f}(t)| &\leq l_b(t) |y(t) - \overline{y}(t)| \\ &\leq l_b(t) ||y - \overline{y}|| \end{aligned}$$

Let us define, for each $t \in [0, b]$

$$\overline{h}(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ \overline{f}(s) \ ds.$$

Then, we can show as that we have for each $t \in [0, b]$

$$\begin{aligned} |h(t) - \overline{h}(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \left| f(s) - \overline{f}(s) \right| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \, l_b(s) \|y - \overline{y}\| \, ds \\ &\leq \frac{\widehat{M} l_b^* b^{\alpha}}{\Gamma(\alpha+1)} \|y - \overline{y}\| \end{aligned}$$

Therefore, we have

$$\|h - \overline{h}\| \le \frac{\widehat{M}l_b^* b^\alpha}{\Gamma(\alpha + 1)} \|y - \overline{y}\|.$$

By an analogous relation, obtained by interchanging the roles of y and \overline{y} , it follows that

$$H_d(N(y), N(\overline{y})) \le \frac{\widehat{M}l_b^* b^{\alpha}}{\Gamma(\alpha+1)} \|y - \overline{y}\|.$$

So, for $\frac{\widehat{M}l_b^*b^{\alpha}}{\Gamma(\alpha+1)} < 1, N$ is a contraction.

It's remains to show that N is an admissible operator. Let $y \in C([-r, b]; E)$ and consider the operator $N: C([-r, b]; E) \to \mathcal{P}(C([-r, b]; E))$, given by

$$N(y) = \left\{ h \in C([-r, b], E) : h(t) = \left\{ \begin{array}{ll} \varphi(t), & \text{if } t \in H; \\ U(t, 0) \ \varphi(0) & & \\ +\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t, s) \ f(s) \ ds, & \text{if } t \in [0, b], \end{array} \right\}$$

where $f \in S_{F,y}^n = \{ v \in L^1([0,b]; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in [0,b] \}.$

From (H1) - (H3) and since F is a multivalued map with compact values, we can prove that for every $y \in C([-r,b]; E)$, $N(y) \in \mathcal{P}_{cp}(C([-r,b]; E))$ and there exists $y_{\star} \in C([-r,b]; E)$ such that $y_{\star} \in N(y_{\star})$. Let $h \in C([-r,b]; E)$, $\overline{y} \in \overline{Y}$ and $\epsilon > 0$. Assume that $y_{\star} \in N(\overline{y})$, then we have

$$\begin{aligned} \|\overline{y}(t) - y_{\star}(t)\| &\leq \|\overline{y}(t) - h(t)\| + \|y_{\star}(t) - h(t)\| \\ &\leq \|\overline{y} - N(\overline{y})\| + \|y_{\star}(t) - h(t)\|. \end{aligned}$$

Since h is arbitrary, we may suppose that

$$h \in B(y_{\star}, \epsilon) = \{h \in C([-r, b]; E) : ||h - y_{\star}|| \le \epsilon\}.$$

Therefore,

$$\|\overline{y} - y_\star\| \le \|\overline{y} - N(\overline{y})\| + \epsilon$$

If y is not in $N(\overline{y})$, then $||y_{\star} - N(\overline{y})|| \neq 0$. Since $N(\overline{y})$ is compact, there exists $x \in N(\overline{y})$ such that $||y_{\star} - N(\overline{y})|| = ||y_{\star} - x||$. Then we have

$$\begin{aligned} \|\overline{y}(t) - x(t)\| &\leq \|\overline{y}(t) - h(t)\| + \|x(t) - h(t)\| \\ &\leq \|\overline{y} - N(\overline{y})\| + \|x(t) - h(t)\|. \end{aligned}$$

Thus,

$$\|\overline{y} - x\| \le \|\overline{y} - N(\overline{y})\| + \epsilon.$$

So, N is an admissible operator contraction. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda N(y)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.6 does not hold. A consequence of the nonlinear alternative of Frigon that (C1) holds, we deduce that the operator N has a fixed point y^* which is a mild solution of the fractional evolution inclusion problem (6.1) - (6.2).

6.3 Neutral Multivalued Problem with Finite Delay

We give here an extension to previous results for the neutral case (6.3) - (6.4). Firstly, we define its mild solution.

Lemma 6.3.1. The system (6.3) - (6.4) is equivalent to the nonlinear integral equation

$$y(t) = [\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s)y(s) \, ds \qquad (6.12)$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s) ds$$

In other words, every solution of the integral equation (6.12) is also solution of the system (6.3) - (6.4) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (6.3) - (6.4), and using some classical results from fractional calculus to get (6.12).

Definition 6.3.1. We say that the function $y(\cdot) : [-r, b] \to E$ is a mild solution of the neutral functional evolution system (6.3) – (6.4) if $y(t) = \varphi(t)$ for all $t \in H$ and the restriction of $y(\cdot)$ to the interval J is continuous and there exists $f(\cdot) \in L^1(J; E) : f(t) \in F(t, y_{\rho(t,y_t)})$ a.e. in J such that y satisfies the following integral equation

$$y(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s) \, ds.$$

We consider the hypotheses (H_{φ}) , (H1) - (H3) and we need to introduce the following assumptions :

(H4) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\varphi) - A(\overline{s})g(\overline{s},\overline{\varphi})| \le L_{\star} (|s-\overline{s}| + \|\varphi - \overline{\varphi}\|)$$

for all $\varphi, \overline{\varphi} \in C(H; E)$.

(H5) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H6) There exists a constant $0 < L < \frac{1}{\overline{M}_0}$ such that

 $|A(t) g(t,\varphi)| \le L (\|\varphi\| + 1) \text{ for all } t \in J \text{ and } \varphi \in C(H; E).$

Theorem 6.3.1. Suppose that the hypotheses (H1) - (H6) are satisfied and moreover

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1 \tag{6.13}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (6.3) – (6.4) has a least mild solution on [-r, b].

Proof. Transform as below the neutral problem (6.3) - (6.4) into a fixed point problem by considering the multivalued operator $\widetilde{N} : C([-r, b]; E) \to \mathcal{P}(C([-r, b]; E))$ defined by :

$$\widetilde{N}(y) = \left\{ h \in C([-r,b];E) : h(t) = \left\{ \begin{array}{ll} \varphi(t), & \text{if } t \in H; \\ U(t,0) \ [\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds, & \text{if } t \in J. \end{array} \right\}$$

where $f \in S_{F,y} = \{ v \in L^1(J; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in J \}.$

Clearly, the fixed points of the operator \widetilde{N} are mild solutions of the problem (6.3) – (6.4). We remark also that, for each $y \in C([-r, b]; E)$, the set $S_{F,y}$ is nonempty since, by (H2), F has a measurable selection (see [40], Theorem III.6).

Let y be a possible fixed point of the operator \widetilde{N} . Given $t \leq b$, then y should be solution of the inclusion $y \in \lambda \ \widetilde{N}(y)$ for some $\lambda \in (0, 1)$ and there exists $f \in S_{F,y} \Leftrightarrow f(t) \in F(t, y_{\rho(t,y_t)})$ such that, for each $t \in J$, by using (H1) - (H2), and (H5) - (H6) we have for each $t \in [0, b]$

$$\begin{aligned} |y(t)| &\leq |g(t, y_{\rho(t, y_t)})| + |U(t, 0)[\varphi(0) - g(0, \varphi)]| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} U(t, s) f(s, y_{\rho(s, y_s)}) ds \right| \\ &\leq ||A^{-1}(t)|| |A(t)g(t, y_{\rho(t, y_t)})| + \widehat{M}||\varphi|| + ||U(t, 0)||_{B(E)} ||A^{-1}(0)|| |A(0)g(0, \varphi) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s) ||y_{\rho(s, y_s)}|| \right] ds \\ &\leq \widehat{M}||\varphi|| + \overline{M}_0 L \left(||y_{\rho(t, y_t)}|| + 1 \right) + \widehat{M}\overline{M}_0 L \left(||\varphi|| + 1 \right) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left[p(s) + q(s) ||y_{\rho(s, y_s)}|| \right] ds \end{aligned}$$

Since $||y_{\rho(t,y_t)}|| \le |y(t)| + \mathcal{L}^{\varphi}||\varphi||$ we obtain

$$\begin{aligned} |y(t)| &\leq \overline{M}_0 L \left(|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| + 1 \right) + \widehat{M} \|\varphi\| (1 + \overline{M}_0 L) + \frac{M p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi} \|\varphi\| \right) ds \end{aligned}$$

Then

$$\begin{aligned} (1 - \overline{M}_0 L)|y(t)| &\leq \left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right] \|\varphi\| + \overline{M}_0 L + \frac{\overline{M}p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ &+ \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|y(s)| + \mathcal{L}^{\varphi}\|\varphi\|\right) ds \end{aligned}$$

 Set

$$\delta_b := \mathcal{L}^{\varphi} \|\varphi\| + \frac{\left[\overline{M}_0 L \mathcal{L}^{\varphi} + \widehat{M}(1 + \overline{M}_0 L)\right]}{(1 - \overline{M}_0 L)} \|\varphi\| + \overline{M}_0 L + \frac{\widehat{M} p^* b^{\alpha}}{(1 - \overline{M}_0 L)\Gamma(\alpha + 1)}.$$

Thus

$$|y(t)| + \mathcal{L}^{\varphi} \|\varphi\| \leq \delta_b + \frac{\widehat{M}q^*}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (|y(s)| + \mathcal{L}^{\varphi} \|\varphi\|) ds.$$

We consider the function μ defined by

 $\mu(t) := \sup \{ |y(s)| + \mathcal{L}^{\varphi} \|\varphi\| : 0 \le s \le t \}, \quad 0 \le t \le b.$

Let $t^* \in [-r, t]$ be such that $\mu(t) = |y(t^*)| + \mathcal{L}^{\varphi} ||\varphi||$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{\widehat{M}q^*}{(1 - \overline{M}_0 L)\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in [-r, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{\widehat{M}q^*b^{\alpha}}{(1 - \overline{M}_0 L)\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||y|| \le \mu(t)$, we have $||y|| \le \max\{||\varphi||, \Lambda_b\} := \Theta_b$.

We can show as in the previous section that \widetilde{N} is an admissible operator and we shall prove now that $\widetilde{N}: \overline{Y} \to \mathcal{P}(C([-r, b]; E))$ is a contraction.

Let $y, \overline{y} \in C([-r, b]; E)$ and $h \in \widetilde{N}(y)$. Then there exists $f(t) \in F(t, y_{\rho(t,y_t)})$ such that for each $t \in [0, b]$, we have

$$h(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s) \, ds$$

From (H3) it follows that

$$H_d(F(t, y_{\rho(t, y_t)}), F(t, \overline{y}_{\rho(t, y_t)})) \le l_b(t) \|y_{\rho(t, y_t)} - \overline{y}_{\rho(t, y_t)}\|.$$

Hence, there is $\rho \in F(t, \overline{y}_{\rho(t,y_t)})$ such that

$$|f(t) - \varrho| \le l_b(t) \|y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)}\| \quad t \in [0,b].$$

Consider $\mathcal{U}_{\star}: [0, b] \to \mathcal{P}(E)$, given by

$$\mathcal{U}_{\star} = \{ \varrho \in E : |f(t) - \varrho| \le l_b(t) \| y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)} \| \}.$$

Since the multivalued operator $\mathcal{V}(t) = \mathcal{U}_{\star}(t) \cap F(t, \overline{y}_{\rho(t,y_t)})$ is measurable (in [40], see Proposition III.4), there exists a function $\overline{f}(t)$, which is a measurable selection for \mathcal{V} . So, $\overline{f}(t) \in F(t, \overline{y}_{\rho(t,y_t)})$, and we obtain for each $t \in [0, b]$

$$|f(t) - \overline{f}(t)| \le l_b(t) \|y_{\rho(t,y_t)} - \overline{y}_{\rho(t,y_t)}\|$$

Using (H3), (H_{φ}) and Lemma 5.2.1, we obtain

$$|f(t) - \overline{f}(t)| \le l_b(t) |y(t) - \overline{y}(t)|$$

Let us define, for each $t \in [0, b]$

$$\overline{h}(t) = U(t,0)[\varphi(0) - g(0,\varphi)] + g(t,\overline{y}_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)\overline{f}(s)ds$$

Then, by (H1), (H3) and (H4), we have for each $t \in [0, b]$

$$\begin{aligned} |h(t) - \overline{h}(t)| &\leq |g(t, y_{\rho(t, y_t)}) - g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t, s)\|_{B(E)} \left| f(s, y_{\rho(s, y_s)}) - f(s, \overline{y}_{\rho(s, y_s)}) \right| \ ds \\ &\leq \|A^{-1}(t)\| |A(t)g(t, y_{\rho(t, y_t)}) - A(t)g(t, \overline{y}_{\rho(t, y_t)})| \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \ l_b(s) \ \|y_{\rho(s, y_s)} - \overline{y}_{\rho(s, y_s)}\| \ ds. \end{aligned}$$

Using (H_{φ}) and Lemma 5.2.1, we obtain

$$\begin{aligned} |h(t) - \overline{h}(t)| &\leq \overline{M}_0 L_\star |y(t) - \overline{y}(t)| + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)} \|y - \overline{y}\| \\ &\leq \left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)} \right] \|y - \overline{y}\| \end{aligned}$$

Therefore,

$$|h(t) - \overline{h}(t)| \le \left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)}\right] \|y - \overline{y}\|$$

By an analogous relation, obtained by interchanging the roles of y and \overline{y} , it follows that

$$H_d(\widetilde{N}(y),\widetilde{N}(\overline{y})) \le \left[\overline{M}_0 L_\star + \frac{\widehat{M}L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] \|y - \overline{y}\|.$$

So, for an appropriate choice of C_b, L_b^* and b^{α} such that

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1,$$

The operator \widetilde{N} is a contraction and an admissible operator. From the choice of Y there is no $y \in \partial Y^n$ such that $y = \lambda \ \widetilde{N}(y)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.6 does not hold. By the nonlinear alternative due to Frigon we get that (C1) holds, we deduce that the operator \widetilde{N} has a fixed point y^* which is a mild solution of the fractional neutral evolution inclusion problem (6.3) - (6.4).

6.4 Partial Multivalued Problem with Infinite Delay

Before stating and proving the main result, we give first the definition of mild solution of the semilinear fractional evolution problem (6.5) - (6.6).

Lemma 6.4.1. The system (6.5) - (6.6) is equivalent to the nonlinear integral equation

$$y(t) = \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s) \, ds.$$
(6.14)

In other words, every solution of the integral equation (6.14) is also solution of the system (6.5) - (6.6) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (6.5) - (6.6), and using some classical results from fractional calculus to get (6.14).

Definition 6.4.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of (6.5)-(6.6)if $y(t) = \phi(t)$ for all $t \in (-\infty, 0]$ and y satisfies the following integral equation

Definition 6.4.2. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of the evolution system (6.5) - (6.6) if $y(t) = \phi(t)$ for all $t \in (-\infty, 0]$ and the restriction of $y(\cdot)$ to the interval J is continuous and there exists $f(\cdot) \in L^1(J; E) : f(t) \in F(t, y_{\rho(t,y_t)})$ a.e. in J such that y satisfies the following integral equation :

$$y(t) = U(t,0) \ \phi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds \quad \text{for each } t \in J.$$
(6.15)

Set

$$\mathcal{R}(\rho^{-}) = \{\rho(s,\phi) : (s,\phi) \in J \times \mathcal{B}, \rho(s,\phi) \le 0\}.$$

We always assume that $\rho: J \times \mathcal{B} \to \mathbb{R}$ is continuous. Additionally, we introduce the following hypothesis:

 (H_{ϕ}) The function $t \to \phi_t$ is continuous from $\mathcal{R}(\rho^-)$ into \mathcal{B} and there exists a continuous and bounded function $\mathcal{L}^{\phi} : \mathcal{R}(\rho^-) \to (0, \infty)$ such that

$$\|\phi_t\|_{\mathcal{B}} \leq \mathcal{L}^{\phi}(t) \|\phi\|_{\mathcal{B}}$$
 for every $t \in \mathcal{R}(\rho^-)$.

We will need to introduce the following hypothesis which are assumed thereafter :

(H01) There exists a constant $\widehat{M} \ge 1$ such that

$$||U(t,s)||_{B(E)} \le \widehat{M}$$
 for every $(t,s) \in \Delta$.

(H02) The multifunction $F: J \times \mathcal{B} \longrightarrow \mathcal{P}(E)$ is L^1_{loc} -Carathéodory with compact and convex values for each $u \in \mathcal{B}$ and there exist a function $p \in L^1_{loc}(J; \mathbb{R}_+)$ and such that

$$||F(t,u)||_{\mathcal{P}(E)} \leq p(t) + q(t) ||u||_B$$
 for a.e. $t \in J$ and each $u \in \mathcal{B}$.

(H03) For all R > 0, there exists $l_R \in L^1_{loc}(J; \mathbb{R}_+)$ such that

$$H_d(F(t, u) - F(t, v)) \le l_R(t) ||u - v||$$

for each $t \in J$ and for all $u, v \in \mathcal{B}$ with $||u|| \leq R$ and $||v|| \leq R$ and

$$d(0, F(t, 0)) \le l_R(t) \quad \text{a.e.} \quad t \in J.$$

Consider the following space

$$\Omega = \left\{ y : (-\infty, b] \to E : y|_{(-\infty, 0]} \in B \text{ and } y|_{[0, b]} \text{ is continuous } \right\},$$

Theorem 6.4.1. Assume that the hypotheses (H_{φ}) and (H01) - (H03) hold and moreover

$$\frac{\widehat{M}l_b^*b^\alpha}{\Gamma(1+\alpha)} < 1 \tag{6.16}$$

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (6.5) - (6.6) has at least mild solution on [-r, b].

Proof. Transform the problem (6.5) - (6.6) into a fixed-point problem. Consider the multivalued operator $N : \Omega \to \mathcal{P}(\Omega)$ defined by :

$$N(y) = \begin{cases} h \in \Omega : h(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ \phi(0) \\ +\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds, & \text{if } t \in J. \end{cases}$$

where $f \in S_{F,y} = \{ v \in L^1(J; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in J \}.$

Clearly, fixed points of the operator N are mild solutions of the problem (6.5) – (6.6). We remark also that, for each $y \in \Omega$, the set $S_{F,y}$ is nonempty since, by (H2), F has a measurable selection (see [40], Theorem III.6).

For $\phi \in \mathcal{B}$, we will define the function $x(.): (-\infty, b] \to E$ by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \in [0, b]; \\ U(t, 0) \ \phi(0), & \text{if } t \in (-\infty, 0]. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (6.15) if and only if z satisfies $z_0 = 0$ and

$$z(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s) ds \quad \text{for } t \in J.$$

where $f(t) \in F(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ a.e $t \in J$

Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operator $F: \Omega^0 \to \mathcal{P}(\Omega^0)$ by :

$$F(z) = \left\{ h \in \Omega : h(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s) \, ds \quad \text{for } t \in J. \right\}$$
(6.17)

Obviously the operator N has a fixed point is equivalent to F has one, so it turns to prove that F has a fixed point.

Let $z \in \Omega^0$ be a possible fixed point of the operator. By the hypotheses (H01) and (H02), we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|f(s)\| \, ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)} \|_{\mathcal{B}} \right] \, ds \end{aligned}$$

From (H_{ϕ}) , Lemma 1.7.2 and Assumption (A1), we have for each $t \in [0, b]$

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq \|z_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} + \|x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi})\|z_0\|_{\mathcal{B}} + K_b|x(s)| + (M_b + \mathcal{L}^{\phi})\|x_0\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\|U(s,0)\|_{B(E)}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + K_b\widehat{M}|\phi(0)| + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \end{aligned}$$

Using (ii), we get

$$\begin{aligned} \|z_{\rho(s,z_s+x_s)} + x_{\rho(s,z_s+x_s)}\|_{\mathcal{B}} &\leq K_b|z(s)| + K_b\widehat{M}H\|\phi\|_{\mathcal{B}} + (M_b + \mathcal{L}^{\phi})\|\phi\|_{\mathcal{B}} \\ &\leq K_b|z(s)| + (M_b + \mathcal{L}^{\phi} + K_b\widehat{M}H)\|\phi\|_{\mathcal{B}} \end{aligned}$$

It follows that

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left(K_b|z(s)| + (M_b + \mathcal{L}^{\phi} + K_b\widehat{M}H) \|\phi\|_{\mathcal{B}}\right) ds$$

Set $c_b := (M_b + \mathcal{L}^{\phi} + K_b \widehat{M} H) \|\phi\|_{\mathcal{B}}$. Then, we have

$$|z(t)| \leq \frac{\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + \frac{\widehat{M}q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)|+c_b) ds$$

Then

$$K_b|z(t)| + c_b \leq \frac{K_b\widehat{M}p^*b^{\alpha}}{\Gamma(\alpha+1)} + c_b + \frac{K_b\widehat{M}q^*}{\Gamma(\alpha)}\int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds.$$

 Set

$$\delta_b := \frac{K_b \widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} + c_b.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in [0, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. By the previous inequality, we have

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \mu(s) ds \quad \text{for } t \in [0,b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\phi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^\alpha}{\Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||z|| \leq \mu(t)$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$. Since for every $t \in [0, b]$, we have $||z|| \leq \max\{||\phi||, \Lambda_b\} := \Theta_b$.

Set

$$Z = \{ z \in \Omega^0 : \sup_{0 \le t \le b} |z(t)| \le \Theta_b + 1 \}.$$

Clearly, Z is a closed subset of Ω^0 .

We shall show that $N : \overline{Z} \to \mathcal{P}(\Omega^0)$ is a contraction and an admissible operator. First, we prove that F is a contraction ; Let $z, \overline{z} \in \Omega^0$ and $h \in F(z)$. Then there exists $f(t) \in F(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ such that for each $t \in [0, b]$

$$h(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s) \ f(s) \ ds$$

From (H03), it follows that

$$H_d(F(t, z_{\rho(t, y_t)} + x_{\rho(t, y_t)}), F(t, \overline{z}_{\rho(t, y_t)} + x_{\rho(t, y_t)}) \le l_b(t) ||z_{\rho(t, y_t)} - \overline{z}_{\rho(t, y_t)}||.$$

Hence, there is $\rho \in F(t, \overline{y}_{\rho(t, u_t)})$ such that

$$|f(t) - \varrho| \le l_b(t) ||z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}||, t \in [0,b].$$

Consider $\mathcal{U}_{\star}: [0, b] \to \mathcal{P}(E)$, given by

$$\mathcal{U}_{\star} = \{ \rho \in E : |f(t) - \rho| \le l_b(t) \ \|z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}\| \}.$$

Since the multivalued operator $\mathcal{V}(t) = \mathcal{U}_{\star}(t) \cap F(t, \overline{z}_{\rho(t,y_t)})$ is measurable (in [40], see Proposition III.4), there exists a function $\overline{f}(t)$, which is a measurable selection for \mathcal{V} . So, $\overline{f}(t) \in F(t, \overline{z}_{\rho(t,y_t)})$ and we obtain for each $t \in [0, b]$

$$|f(t) - f(t)| \le l_b(t) ||z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}||$$

Using (H_{ϕ}) and Lemma 1.7.2, we obtain

$$|f(t) - \overline{f}(t)| \leq l_b(t) |z(t) - \overline{z}(t)|$$

$$\leq l_b(t) ||z - \overline{z}||$$

Let us define, for each $t \in [0, b]$

$$\overline{h}(t) = U(t,0) \ \varphi(0) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) \ \overline{f}(s) \ ds.$$

Then, we can show that, we have, for each $t \in [0, b]$

$$\begin{aligned} |h(t) - \overline{h}(t)| &\leq \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \|U(t,s)\|_{B(E)} \|f(s) - \overline{f}(s)\| ds \\ &\leq \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} |l_b(s)| \|z - \overline{z}\| ds \\ &\leq \frac{\widehat{M} l_b^* b^\alpha}{\Gamma(\alpha+1)} \|y - \overline{y}\| \end{aligned}$$

Therefore, we have

$$\|h - \overline{h}\| \le \frac{\widehat{M} l_b^* b^\alpha}{\Gamma(\alpha + 1)} \|z - \overline{z}\|.$$

By an analogous relation, obtained by interchanging the roles of y and \overline{y} , it follows that

$$H_d(N(y), N(\overline{y})) \le \frac{\widehat{M}l_b^* b^{\alpha}}{\Gamma(\alpha+1)} \|z - \overline{z}\|.$$

So, for $\frac{\widehat{M}l_b^*b^{\alpha}}{\Gamma(\alpha+1)} < 1$, N is a contraction.

Now we show that F is an admissible operator. Let $z\in \Omega^0$. Set, the space

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}$$

and let us consider the multivalued operator $F: \Omega^0 \to \mathcal{P}(\Omega^0)$

$$F(z) = \left\{ h \in \Omega : h(t) = \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s) \, ds \quad \text{for } t \in J. \right\}$$
(6.18)

where $f \in S_{F,y}^n = \{ v \in L^1([0,b]; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in [0,b] \}.$

From (H01) - (H03) and since F is a multivalued map with compact values, we can prove that for every $z \in \Omega^0$, $F(z) \in \mathcal{P}_{cp}(\Omega^0)$ and there exists $z_* \in \Omega^0$ such that $z_* \in F(z_*)$. Let $h \in \Omega^0, \overline{y} \in \overline{Y}$ and $\epsilon > 0$. Assume that $z_* \in F(\overline{z})$, then we have

$$\begin{aligned} \|\overline{z}(t) - z_{\star}(t)\| &\leq \|\overline{z}(t) - h(t)\| + \|z_{\star}(t) - h(t)\| \\ &\leq \|\overline{z} - N(\overline{z})\| + \|z_{\star}(t) - h(t)\|. \end{aligned}$$

Since h is arbitrary, we may suppose that

$$h \in B(z_{\star}, \epsilon) = \{h \in C([-r, b]; E) : ||h - z_{\star}|| \le \epsilon\}.$$

Therefore,

$$\|\overline{z} - z_{\star}\| \le \|\overline{z} - N(\overline{z})\| + \epsilon.$$

If z is not in $F(\overline{z})$, then $||z_{\star} - F(\overline{z})|| \neq 0$. Since $F(\overline{z})$ is compact, there exists $x \in F(\overline{z})$ such that $||z_{\star} - F(\overline{z})|| = ||z_{\star} - x||$. Then we have

$$\begin{aligned} \|\overline{z}(t) - x(t)\| &\leq \|\overline{z}(t) - h(t)\| + \|x(t) - h(t)\| \\ &\leq \|\overline{z} - F(\overline{z})\| + \|x(t) - h(t)\|. \end{aligned}$$

Thus,

$$\|\overline{z} - x\| \le \|\overline{z} - F(\overline{z})\| + \epsilon.$$

So, F is an admissible operator contraction. From the choice of z there is no $z \in \partial Z^n$ such that $z = \lambda F(z)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.6 does not hold. A consequence of the nonlinear alternative of Frigon that (C1) holds, we deduce that the operator F has a fixed point y^* which is a mild solution of the fractional evolution inclusion problem (6.5) - (6.6).

6.5 Neutral Multivalued Problem with Infinite Delay

We give here an extension to previous results for the neutral case (6.7) - (6.8). Firstly, we define its mild solution.

Lemma 6.5.1. The system (6.7) - (6.8) is equivalent to the nonlinear integral equation

$$y(t) = [\phi(0) - g(0,\phi)] + g(t, y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} A(s) y(s) \, ds \tag{6.19}$$

+
$$\frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} f(s) ds$$

In other words, every solution of the integral equation (6.19) is also solution of the system (6.7) - (6.8) and vice versa.

Proof. It can be proved by applying the integral operator to both sides of the system (6.7) - (6.8), and using some classical results from fractional calculus to get (6.19).

Definition 6.5.1. We say that the function $y(\cdot) : (-\infty, b] \to E$ is a mild solution of the fractional neutral functional evolution system (6.7) – (6.8) if $y(t) = \phi(t)$ for all $t \leq 0$ and the restriction of $y(\cdot)$ to the interval J is continuous and there exists $f(\cdot) \in L^1(J; E) : f(t) \in F(t, y_{\rho(t,y_t)})$ a.e. in J such that y satisfies the following integral equation

$$y(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s) \ ds \tag{6.20}$$

We consider the hypotheses (H_{ϕ}) , (H01) - (H03) and we need to introduce the following assumptions :

(H04) There exists a constant $L_{\star} > 0$ such that

$$|A(s)g(s,\phi) - A(\overline{s})g(\overline{s},\phi)| \le L_{\star} (|s - \overline{s}| + \|\phi - \phi\|)$$

for all $s, \overline{s} \in J, \phi, \overline{\phi} \in \mathcal{B}$.

(H05) There exists a constant $\overline{M}_0 > 0$ such that

$$||A^{-1}(t)||_{B(E)} \le \overline{M}_0 \quad \text{for all } t \in J.$$

(H06) There exists a constant $0 < L < \frac{1}{\overline{M}_0 K_b}$ such that

 $|A(t) g(t, \varphi)| \leq L (||\phi|| + 1)$ for all $t \in J$ and $\phi \in \mathcal{B}$.

Theorem 6.5.1. Suppose that the hypotheses (H01) - (H06) are satisfied and moreover

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1$$
(6.21)

where $p^* = \sup p(s)$, $q^* = \sup q(s)$ and $l_b^* = \sup l_b(s)$. Then the problem (6.7) – (6.8) has at least mild solution on $(-\infty, b]$.

Proof. Transform as below the neutral problem (6.7) - (6.8) into a fixed point problem by considering the multivalued operator $\widetilde{N} : \Omega \to \mathcal{P}(\Omega)$ defined by :

$$\widetilde{N}(y) = \begin{cases} h \in \Omega : h(t) = \begin{cases} \phi(t), & \text{if } t \le 0; \\ U(t,0) \ [\phi(0) - g(0,\phi)] + g(t,y_{\rho(t,y_t)}) \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \ U(t,s)f(s) \ ds, & \text{if } t \in J. \end{cases}$$

where $f \in S_{F,y} = \{ v \in L^1(J; E) : v(t) \in F(t, y_{\rho(t,y_t)}) \text{ for a.e. } t \in J \}.$

Clearly, the fixed points of the operator \widetilde{N} are mild solutions of the problem (6.7)-(6.8). We remark also that, for each $y \in \Omega$, the set $S_{F,y}$ is nonempty since, by (H02), F has a measurable selection (see [40], Theorem III.6).

For $\phi \in \mathcal{B}$, we consider the function $x(.): (-\infty, b] \to E$ defined as below by

$$x(t) = \begin{cases} \phi(t), & \text{if } t \leq 0; \\ U(t,0) \ \phi(0), & \text{if } t \in J. \end{cases}$$

Then $x_0 = \phi$. For each function $z \in \Omega$, set

$$y(t) = z(t) + x(t)$$

It is obvious that y satisfies (6.20) if and only if z satisfies $z_0 = 0$ and

$$z(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s)ds.$$

where $f(t) \in F(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ a.e $t \in J$ Let

$$\Omega^0 = \{ z \in \Omega : z_0 = 0 \}.$$

Define the operator $\widetilde{F}:\Omega^0\to \mathcal{P}(\Omega^0)$ by :

$$\widetilde{F}(z) = \{h \in \Omega : h(t) = g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - U(t,0)g(0,\phi) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s)ds.\}$$
(6.22)

Obviously the operator \widetilde{N} has a fixed point is equivalent to \widetilde{F} has one, so it turns to prove that \widetilde{F} has a fixed point.

Let $z \in \Omega^0$ be be a possible fixed point of the operator. Then, using (H01) - (H02), (H05) - (H06) we have for each $t \in [0, b]$

$$\begin{aligned} |z(t)| &\leq |g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})| + |U(t,0)g(0,\phi)| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) ds \, ds \right| \\ &\leq ||A^{-1}(t)|| \, |A(t)g(t, z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)})| + ||U(t,0)||_{B(E)} ||A^{-1}(0)|| \, |A(0)g(0,\phi)| \\ &+ \left| \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s) f(s, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) ds \, ds \right| \\ &\leq \overline{M}_0 L(||z_{\rho(t,z_t+x_t)} + x_{\rho(t,z_t+x_t)}||_{\mathcal{B}} + 1) + \widehat{M}\overline{M}_0 L(||\phi||_{\mathcal{B}} + 1) \\ &+ \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} \left[p(s) + q(s) ||z_{\rho(s,y_s)} + x_{\rho(s,y_s)}|| \right] ds \end{aligned}$$

Since $||z_{\rho(t,y_t)} + x_{\rho(t,y_t)}||_{\mathcal{B}} \le K_b |z(t)| + c_b$ we obtain

$$|z(t)| \leq \overline{M}_0 L \left(K_b | z(t) | + c_b + 1 \right) + \overline{M}_0 L \widehat{M} (\|\phi\| + 1) + \frac{\widehat{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \left(|z(s)| + c_b \right) ds$$

Then

$$(1 - \overline{M}_0 L K_b)|z(t)| \leq \overline{M}_0 L (c_b + 1) + \overline{M}_0 L \widehat{M}(\|\phi\| + 1) + \frac{\overline{M} p^* b^{\alpha}}{\Gamma(\alpha + 1)} \\ + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} (|z(s)| + c_b) ds$$

 Set

$$\delta_b := c_b + \frac{K_b \overline{M}_0 L\left((c_b + 1) + \widehat{M}(\|\phi\| + 1)\right)}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha + 1)} + \frac{K_b \widehat{M} p^* b^{\alpha}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha + 1)}$$

Thus

$$K_b|z(t)| + c_b \leq \delta_b + \frac{K_b \widehat{M} q^*}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} (K_b|z(s)| + c_b) ds.$$

We consider the function μ defined by

$$\mu(t) := \sup \{ K_b | z(s) | + c_b : 0 \le s \le t \}, \quad 0 \le t \le b.$$

Let $t^* \in (-\infty, t]$ be such that $\mu(t) = K_b |z(t^*)| + c_b$. If $t^* \in [0, b]$, by the previous inequality, we get

$$\mu(t) \leq \delta_b + \frac{K_b \widehat{M} q^*}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \mu(s) ds \quad \text{for } t \in [0, b].$$

If $t^* \in (-\infty, 0]$, then $\mu(t) = \|\varphi\|$ and the previous inequality holds. And Lemma 1.8.1 implies that there exists a positive constant $\delta_b = \delta_b(\alpha)$ such that

$$\mu(t) \leq \delta_b \times \left[1 + \frac{K_b \widehat{M} q^* b^{\alpha}}{(1 - \overline{M}_0 L K_b) \Gamma(\alpha + 1)} \right]$$
$$= \Lambda_b$$

Since $||z|| \le \mu(t)$, we have $||z|| \le \max\{||\phi||, \Lambda_b\} := \Theta_b$.

We can show as in the previous section that \widetilde{N} is an admissible operator and we shall prove now that $\widetilde{N}: \overline{Z} \to \mathcal{P}(\Omega^0)$ is a contraction.

Let $z, \overline{z} \in \Omega^0$ and $h \in \widetilde{F}(z)$. Then there exists $f(t) \in F(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ such that for each $t \in [0, b]$

$$h(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)f(s) \, ds$$

From (H03) it follows that

$$H_d(F(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}), F(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)}) \le l_b(t) \|z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}\|.$$

Hence, there is $\varrho \in F(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ such that

$$|f(t) - \varrho| \le l_b(t) ||z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}|| \quad t \in [0,b].$$

Consider $\mathcal{U}_{\star}: [0, b] \to \mathcal{P}(E)$, given by

$$\mathcal{U}_{\star} = \{ \varrho \in E : |f(t) - \varrho| \le l_b(t) \| z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)} \| \}.$$

Since the multivalued operator $\mathcal{V}(t) = \mathcal{U}_{\star}(t) \cap F(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})$ is measurable (in [40], see Proposition III.4), there exists a function $\overline{f}(t)$, which is a measurable selection for \mathcal{V} . So, $\overline{f}(t) \in F(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})$, and we obtain for each $t \in [0, b]$

$$|f(t) - \overline{f}(t)| \le l_b(t) ||z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}||$$

Using (H03), (H_{φ}) and Lemma 1.7.2, we obtain

$$|f(t) - \overline{f}(t)| \le l_b(t) |z(t) - \overline{z}(t)|$$

Let us define, for each $t \in [0, b]$

$$\overline{h}(t) = U(t,0)[\phi(0) - g(0,\phi)] + g(t,\overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)}) + \frac{1}{\Gamma(\alpha)} \int_0^t (t-s)^{\alpha-1} U(t,s)\overline{f}(s) ds$$

Then, by (H01), (H03) and (H04), we have for each $t \in [0, b]$

$$\begin{aligned} |h(t) - h(t)| &\leq |g(t, z_{\rho(s,y_s)} + x_{\rho(s,y_s)}) - g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t,s)\|_{B(E)} |f(s) - \overline{f}(s)| ds \\ &\leq \|A^{-1}(t)\| |A(t)g(t, z_{\rho(t,y_t)} + x_{\rho(t,y_t)}) - A(t)g(t, \overline{z}_{\rho(t,y_t)} + x_{\rho(t,y_t)})| \\ &+ \frac{1}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} \|U(t,s)\|_{B(E)} |f(s) - \overline{f}(s)| ds \\ &\leq \overline{M}_0 L_\star \|z_{\rho(t,y_t)} - \overline{z}_{\rho(t,y_t)}\| + \frac{\widehat{M}}{\Gamma(\alpha)} \int_0^t (t - s)^{\alpha - 1} |l_b(s)| \|z_{\rho(s,y_s)} - \overline{z}_{\rho(s,y_s)}\| ds. \end{aligned}$$

Using (H_{ϕ}) and Lemma 1.7.2, we obtain

$$\begin{aligned} |h(t) - \overline{h}(t)| &\leq \overline{M}_0 L_\star ||z - \overline{z}|| + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)} ||z - \overline{z}|| \\ &\leq \left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)} \right] ||z - \overline{z}|| \end{aligned}$$

Therefore,

$$|h(t) - \overline{h}(t)| \le \left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha + 1)}\right] ||z - \overline{z}||$$

By an analogous relation, obtained by interchanging the roles of z and \overline{z} , it follows that

$$H_d(\widetilde{N}(z), \widetilde{N}(\overline{z})) \le \left[\overline{M}_0 L_\star + \frac{\widehat{M}L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] \|z - \overline{z}\|.$$

So, for an appropriate choice of $\overline{M}_0 L_\star, L_b^\star$ and b^α such that

$$\left[\overline{M}_0 L_\star + \frac{\widehat{M} L_b^* b^\alpha}{\Gamma(\alpha+1)}\right] < 1,$$

The operator \widetilde{N} is a contraction and an admissible operator. From the choice of Z there is no $z \in \partial Z^n$ such that $z = \lambda \widetilde{N}(z)$ for some $\lambda \in (0, 1)$. Then the statement (C2) in Theorem 1.10.6 does not hold. By the nonlinear alternative due to Frigon we get that (C1) holds, we deduce that the operator \widetilde{N} has a fixed point z^* which is a mild solution of the fractional neutral evolution inclusion (6.7) - (6.8).

6.6 Examples

To illustrate the previous results, we give in this section four applications: **Example 1.** Consider the partial differential inclusion

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) \in \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-r}^{0}a_{1}(s-t)u\left[s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]ds, \\ -r \leq t \leq b, \ \xi \in [0,\pi], \\ u(t,0) = u(t,\pi) = 0, \qquad -r \leq t \leq b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -r < \theta \leq 0, \ \xi \in [0,\pi], \end{cases}$$
(6.23)

where $a_1(t,\xi)$ is a continuous function and is uniformly Hölder continuous in t $0 < \alpha \leq 1$; $a_1 : [-r,0] \to \mathbb{R}$; $a_2 : [0,\pi] \to \mathbb{R}$; $u_0 \in C(H;E)$; $\rho_1 : [0,+\infty) \to \mathbb{R}$ are continuous functions; $\rho_2 : [0,+\infty) \to \mathcal{P}(\mathbb{R})$ is a multivalued map with compact convex values. To study this system, we consider the space $E = L^2([0, \pi], \mathbb{R})$ and the operator $A : D(A) \subset E \to E$ given by Aw = w'' with

$$D(A) := \{ w \in E : w'' \in E, w(0) = w(\pi) = 0 \}$$

It is well known that A is the infinitesimal generator of an analytic semigroup $\{T(t)\}_{t\in[0,b]}$ on E, with compact resolvent. On the domain D(A), we define the operators $A(t): D(A) \subset E \to E$ by

$$A(t)x(\xi) = Ax(\xi) + a_0(t,\xi)x(\xi)$$

By assuming that $a_0(.,.)$ is continuous and that $a_0(t,\xi) \leq -\delta_0$ ($\delta_0 > 0$) for every $t \in \mathbb{R}, \xi \in [0,\pi]$, and specific case $\alpha = 1$ it follows that the system

$$u'(t) = A(t)u(t) \quad t \ge s,$$
$$u(s) = x \in E,$$

has an associated evolution family given by

$$U(t,s)x(\xi) = \left[T(t-s)exp\left(\int_s^t a_0(\tau,\xi)d\tau\right)x\right](\xi).$$

From this expression, it follows that U(t,s) is a compact linear operator and that

$$||U(t,s)|| \le e^{-(1+\delta_0)(t-s)}$$
 for every $(t,s) \in [0,b] \times [0,b]$; $s \le t$.

Theorem 6.6.1. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, $\rho_1 : [0, +\infty) \to \mathbb{R}$ is a continuous function, $\rho_2 : [0, +\infty) \to \mathcal{P}(\mathbb{R})$ is a multivalued map with compact convex values and the functions $a_1 : [-r, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (6.23).

Proof. From the assumptions, we have that

$$F(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (6.23) into the abstract system (6.1) - (6.2). Moreover, the function F is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 6.2.1.

From Remark 1.7.1, we have the following result.

Corollary 6.6.1. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a mild solution of (6.23) on [-r, b].

Example 2. Consider the partial differential inclusion

$$\begin{aligned}
 ^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-r}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\
 \in \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\
 + \int_{-r}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\
 -r \leq t \leq 0, \ \xi \in [0,\pi], \\
 v(t,0) = v(t,\pi) = 0, \qquad -r \leq t \leq 0, \\
 v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -r < \theta \leq 0, \ \xi \in [0,\pi], \end{aligned}$$
(6.24)

where $a_3: [-r, 0] \to \mathbb{R}$ is a continuous function.

Theorem 6.6.2. Let $\varphi \in C(H; E)$. Assume that the condition (H_{φ}) holds, $\rho_1 : [0, +\infty) \to \mathbb{R}$ is a continuous function, $\rho_2 : [0, +\infty) \to \mathcal{P}(\mathbb{R})$ is a multivalued map with compact convex values and the functions $a_1, a_3 : [-r, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (6.24).

Proof. From the assumptions, we have that

$$F(t,\psi)(\xi) = \int_{-r}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-r}^{0} a_3(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_{0}^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (6.24) into the abstract system (6.3) - (6.4). Moreover, the function F is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 6.3.1.

From Remark 1.7.1, we have the following result.

Corollary 6.6.2. Let $\varphi \in C(H; E)$ be continuous and bounded. Then there exists a mild solution of (6.24) on [-r, b].

Example 3. Consider the partial differential inclusion

$$\begin{cases} {}^{c}D_{0}^{\alpha}u(t,\xi) \in \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0}a_{1}(s-t)u\left[s - \rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi}a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right]ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \end{cases}$$
(6.25)
$$u(t,0) = u(t,\pi) = 0, \qquad 0 \le t \le b, \\ u(\theta,\xi) = u_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$

where $a_1: (-\infty, 0] \to \mathbb{R}$ is a continuous function.

Theorem 6.6.3. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_1 : [0, +\infty) \to \mathbb{R}$ is a continuous function, $\rho_2 : [0, +\infty) \to \mathcal{P}(\mathbb{R})$ is a multivalued map with compact convex values and the functions $a_1 : (-\infty, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (6.25).

Proof. From the assumptions, we have that

$$F(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^{\pi} a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (6.25) into the abstract system (6.5) - (6.6). Moreover, the function F is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 6.4.1.

From Remark 1.7.2, we have the following result.

Corollary 6.6.3. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (6.25) on $(-\infty, b]$.

Example 4. Consider the partial differential inclusion

$$\begin{cases} {}^{c}D_{0}^{\alpha} \left[u(t,\xi) - \int_{-\infty}^{0} a_{3}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds \right] \\ \in \frac{\partial^{2}u(t,\xi)}{\partial\xi^{2}} + a_{0}(t,\xi)u(t,\xi) \\ + \int_{-\infty}^{0} a_{1}(s-t)u\left(s-\rho_{1}(t)\rho_{2}\left(\int_{0}^{\pi} a_{2}(\theta)|u(t,\theta)|^{2}d\theta\right),\xi\right)ds, \\ 0 \le t \le b, \ \xi \in [0,\pi], \\ v(t,0) = v(t,\pi) = 0, \qquad 0 \le t \le b, \\ v(\theta,\xi) = v_{0}(\theta,\xi), \qquad -\infty < \theta \le 0, \ \xi \in [0,\pi], \end{cases}$$
(6.26)

where $a_3: (-\infty, 0] \to \mathbb{R}$ is a continuous functions.

Theorem 6.6.4. Let $\phi \in \mathcal{B}$. Assume that the condition (H_{ϕ}) holds, $\rho_1 : [0, +\infty) \to \mathbb{R}$ is a continuous function, $\rho_2 : [0, +\infty) \to \mathcal{P}(\mathbb{R})$ is a multivalued map with compact convex values and the functions $a_1, a_3 : (-\infty, 0] \to \mathbb{R}$ and $a_2 : [0, \pi] \to \mathbb{R}$ are continuous. Then there exists a mild solution of (6.26).

Proof. From the assumptions, we have that

$$F(t,\psi)(\xi) = \int_{-\infty}^{0} a_1(s)\psi(s,\xi)ds,$$
$$g(t,\psi)(\xi) = \int_{-\infty}^{0} a_3(s)\psi(s,\xi)ds,$$

$$\rho(s,\psi) = s - \rho_1(s)\rho_2\left(\int_0^\pi a_2(\theta)|\psi(0,\xi)|^2d\theta\right),$$

are well defined functions, which permit to transform system (6.26) into the abstract system (6.7) - (6.8). Moreover, the function F is bounded linear operator. Now, the existence of mild solutions can be deduced from a direct application of Theorem 6.5.1.

From Remark 1.7.2, we have the following result.

Corollary 6.6.4. Let $\phi \in \mathcal{B}$ be continuous and bounded. Then there exists a mild solution of (6.26) on $(-\infty, b]$.

Conclusion

In this thesis, we have present some results of existence, even uniqueness and controllability of mild solutions on a bounded and unbounded interval for different classes of firt order and Caputo's fractional derivative order of differential and integrodifferential, perturbed and nonperturbed, partial functional and neutral functional *evolution* equations and inclusions with finite and infinite state-dependent delay using various nonlinear alternatives on Fréchet and Banach spaces depending on the fixed point argument and combined with semigroups theory.

We have generalized the various evolution problems in the thesis of Baghli [20] for the Caputo's fractional derivative order and we have considered the case when the delay is depending on the solution following the paper of Baghli-Benchohra-Nieto [?] and Baghli [19].

Our first work, in this thesis was proving the existence of mild solutions on unbounded interval for the first order perturbed evolution equations with infinite state-dependent delay with the Theorem 3.5, page 8 of our first paper [10] and we have extended the result to the first order neutral perturbed evolution equations with infinite state-dependent delay with the Theorem 4.2, page 14 of our first paper [10].

Then in *our second work*, we have give in this thesis the controllability of mild solutions on unbounded interval for the first order evolution equations with infinite state-dependent delay on the Theorem 2, page 388 of our second paper [11] and we have extended the result to the first order neutral evolution equations with infinite state-dependent delay with the Theorem 3, page 394 of our first paper [11].

The others results about fractional evolution equations and inclusions are submitted or in preparation.

CONCLUSION

Perspectives

Our perspectives are to study the existence, even uniqueness and the controllability of mild solutions on the whole positif real line for the different classes of fractional evolution equations studied in this thesis when the delay is finite and infinite and also is depending on the solution. We look also for the existence and the controllability of mild solutions for fractional perturbed and nonperturbed evolution inclusions with finite and infinite state-dependent delay.

PERSPECTIVES

Bibliography

- N. Abada, R. P. Agarwal, M. Benchohra and H. Hammouche, Existence results for nondensely defined impulsive semilinear functional differential equations with state-dependent delay, *Asian-Eur. J. Math.*, 1 (4) (2008), 449-468.
- [2] S. Abbas and M. Benchohra, Partial hyperbolic differential equations with finite delay involving the Caputo fractional derivative, *Commun. Math. Anal.*, 7 (2) (2009), 62-72.
- [3] S. Abbas and M. Benchohra, Darboux problem for perturbed partial differential equations of fractional order with finite delay, *Nonlinear Anal.: Hybrid Systems*, **3** (2009), 597-604.
- [4] S. Abbas, M. Benchohra and A. N. Vityuk, On fractional order derivatives and Darboux problem for implicit differential equations. *Frac. Calc. Appl. Anal.*, **15** (2) (2012), 168-182.
- [5] R. P. Agarwal, S. Baghli-Bendimerad and M. Benchohra, Controllability of mild solutions on semiinfinite interval for classes of semilinear functional and neutral functional evolution equations with infinite delay, *Applied Mathematics and Optimization*, **60** (2) (2009), 253– 274.
- [6] R. P. Agarwal, M. Belmekki and M. Benchohra, A survey on semilinear differential equations and inclusions involving Riemann-Liouville derivative, Adv. Differ. Equations, to appear.
- [7] N. U. Ahmed, Semigroup theory with applications to systems and control, Pitman Res. Notes Math. Ser., 246. Longman Scientific & Technical, Harlow John Wiley & Sons, Inc., New York, 1991.
- [8] N. U. Ahmed, Dynamic systems and control with applications, World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2006.
- [9] E. Ait Dads and K. Ezzinbi, Boundedness and almost periodicity for some state-dependent delay differential equations, *Electron. J. Differential Equations*, **2002** (67) (2002), 1-13.
- [10] D. Aoued and S. Baghli-Bendimerad, Mild solutions for perturbed evolution equations with in nite state-dependent delay, *Electronic Journal of Qualitative Theory of Differential Equations* 2013, 59, 1–24.
- [11] D. Aoued and S. Baghli-Bendimerad, Controllability of mild solutions for evolution equations with infinite state-dependent delay, *Europian Journal of Pure and Applied Mathematics*, 9 (4) (2016), 383-401.

- [12] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Fractional partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.
- [13] D. Aoued, S. Baghli-Bendimerad and M. Benchohra, Fractional perturbed partial functional and neutral functional evolution equations with infinite state-dependent delay, submitted.
- [14] **D.** Aoued, S. Baghli-Bendimerad and M. Benchohra, Controllability of mild solutions for fractional partial functional and neutral functional evolution equations with statedependent delay, submitted.
- [15] **D. Aoued**, S. Baghli-Bendimerad and M. Benchohra, Multivalued fractional partial functional and neutral functional fractional evolution equations with state-dependent delay, submitted.
- [16] A. Anguraj, M. M. Arjunan and E. M. Hernàndez, Existence results for an impulsive neutral functional differential equation with state-dependent delay, *Appl. Anal.*, 86 (7) (2007), 861-872.
- [17] O. Arino, K. Boushaba and A. Boussouar, A mathematical model of the dynamics of the phytoplankton-nutrient system. Spatial heterogeneity in ecological models, *Nonlinear Anal. RWA*, 1 (1) (2000), 69-87.
- [18] C. Avramescu, Some remarks on a fixed point theorem of Krasnoselskii, Electron. J. Qual. Theory Differ. Equ., 5 (2003), 1-15.
- [19] S. Baghli-Bendimerad, Global mild solution for functional evolution inclusions with statedependent delay, Journal of Advanced Research in Dynamical and Control Systems, 5 (4) (2013), 1–19.
- [20] S. Baghli-Bendimerad, Résultats d'existence et d'unicité des solutions faibles pour quelques classes d'équations et d'inclusions d'évolution à retard, Thèse de Doctorat, Université de Sidi Bel-Abbès, 2008.
- [21] S. Baghli-Bendimerad and M. Benchohra, Uniqueness results for partial functional differential equations in Fréchet spaces, *Fixed-Point Theory*, 9 (2) (2008), 395–406.
- [22] S. Baghli-Bendimerad and M. Benchohra, Existence results for semilinear neutral functional differential equations involving evolution operators in Fréchet spaces, *Georgian Mathematical Journal*, Volume 17 (3) (2010), 423-436.
- [23] S. Baghli-Bendimerad and M. Benchohra, Global Uniqueness results for partial functional and neutral functional evolution equations with infinite delay, *Differential Integral Equations*, 23 (1)&(2) (2010), 31–50.
- [24] S. Baghli-Bendimerad, M. Benchohra and Kh. Ezzinbi, Controllability results for semilinear functional and neutral functional evolution equations with infinite delay, *Surveys in Mathematics and its applications*, 4 (2009), 15–39.
- [25] S. Baghli-Bendimerad and M. Benchohra, Perturbed functional and neutral functional evolution equations with infinite delay in Fréchet spaces, *Electronic Journal of Differential* Equations, 2008 (69) (2008), 1–19.

- [26] S. Baghli-Bendimerad and M. Benchohra, Multivalued evolution equations with infinite delay in Fréchet spaces, E. J. Qualitative Theory of Diff. Equ., 2008 (33) (2008), 1–24.
- [27] S. Baghli-Bendimerad, M. Benchohra and J. J. Nieto, Global uniqueness results for partial functional and neutral functional evolution equations with state-dependent delay, *Journal* of Advanced Research in Differential Equations, 2 (3) (2010), 35-52.
- [28] D. Baleanu, K. Diethelm, E. Scalas and J. J. Trujillo, Fractional calculus models and numerical methods, World Scientific Publishing, New York, 2012.
- [29] A. Belarbi, M. Benchohra and A. Ouahab, Uniqueness results for fractional functional differential equations with infinite delay in Fréchet spaces, *Appl. Anal.*, 85 (2006), 1459-1470.
- [30] M. Benchohra, J. Henderson and S. K. Ntouyas, *Impulsive differential equations and in*clusions, Hindawi Publishing Corporation, Volume 2, New York, 2006.
- [31] M. Benchohra, J. Henderson, S. K. Ntouyas and A. Ouahab, Existence results for functional differential equations of fractional order, J. Math. Anal. Appl., 338 (2008), 1340-1350.
- [32] M. Benchohra, L. Górniewicz and S. K. Ntouyas, Controllability of some nonlinear systems in Banach spaces: The fixed point theory approach, Pawel Wlodkowicz University College, Plock, 2003.
- [33] M. Benchohra and S. K. Ntouyas, Existence of mild solutions on semiinfinite interval for first order differential equation with nonlocal condition, *Comm. Math. Univ. Carolinae*, 41 (3) (2000), 485-491.
- [34] A. Bensoussan, G. Da Prato, M. C. Delfour and S. K. Mitter, Representation and control of infinite dimension systems, Volume 2, Systems and Control: Foundations and Applications, Birkhauser, Boston, 1993.
- [35] B. Bonilla, M. Rivero, L. Rodrìguez-Germá and J. J. Trujillo, Fractional differential equations as alternative models to nonlinear differential equations, *Appl. Math. Comput*, 187 (2007), 79-88.
- [36] T. A. Burton, A fixed-point theorem of Krasnoselskii, Appl. Math. Lett., 11 (1) (1998), 85-88.
- [37] T. A. Burton and C. Kirk, A fixed point theorem of Krasnoselskii type, Math. Nach., 189 (1998), 23-31.
- [38] N. Carmichael and M.D. Quinn, An approach to nonlinear control problems using the fixed point methods, degree theory and pseudo-inverses, *Numer. Funct. Anal. Optim.*, 7 (1984-1985), 197-219.
- [39] A. Carpinteri and F. Mainardi, Fractals and fractional calculus in continuum mechanics, 378 CISM Courses and Lectures, Springer-Verlag, Berlin, 1997.
- [40] C. Castaing and M. Valadier, Convex Analysis and Measurable Multifunctions, Lecture Notes in Math., 580, Springer-Verlag, New York, 1977.

- [41] C. Corduneanu and V. Lakshmikantham, Equations with unbounded delay, Nonlinear Anal., 4 (1980), 831-877.
- [42] R. Curtain and H. J. Zwart, An Introduction to infinite dimensional linear systems theory, Springer-Verlag, New-Yok, 1995.
- [43] M. A. Darwish and S. K. Ntouyas, Semilinear functional differential equations of fractional order with state-dependent delay. *Electron. J. Differential Equations*, **2009** (38) (2009), 1-10.
- [44] M. Dawidowski and I. Kubiaczyk, An existence theorem for the generalized hyperbolic equation $z''_{xy} \in F(x, y, z)$ in Banach space, Ann. Soc. Math. Pol. Ser. I, Comment. Math., **30** (1) (1990), 41-49.
- [45] K. Deimling, *Multivalued differential equations*, Walter de Gruyter, Berlin-New York, 1992.
- [46] K. Diethelm, *The analysis of fractional differential equations*, Lecture Notes in Math., Springer, Berlin, 2010.
- [47] K. J. Engel and R. Nagel, One-parameter semigroups for linear evolution equations, Springer-Verlag, New York, 2000.
- [48] A. Freidman, *Partial differential equations*, Holt, Rinehat and Winston, New York, 1969.
- [49] M. Frigon, Fixed point results for generalized contractions in gauge spaces and applications, Proc. Amer. Math. Soc., 128 (10) (2000), 2957-2965.
- [50] M. Frigon, Fixed point results for multivalued contractions on gauge spaces. Set valued mappings with applications in nonlinear analysis, Ser. Math. Anal. Appl., 4, Taylor & Francis, London, (2002), 175–181.
- [51] M. Frigon, Fixed point and continuation results for contractions in metric and gauge spaces. Fixed point theory and its applications, 89–114, Banach Center Publ., 77, Polish Acad. Sci., Warsaw, 2007.
- [52] M. Frigon and A. Granas, Résultats de type Leray-Schauder pour des contractions sur des espaces de Fréchet, Ann. Sci. Math. Québec, 22 (2) (1998), 161-168.
- [53] X. Fu, Controllability of neutral functional differential systems in abstract space, Appl. Math. Comput., 141 (2003), 281–296.
- [54] X. Fu, Controllability of abstract neutral functional differential systems with unbounded delay, Appl. Math. Comput., 151 (2004), 299–314.
- [55] J. A. Goldstein, Semigroups of linear operators and applications, Oxford, New York, 1985.
- [56] L. Górniewicz, Topological fixed point theory of multivalued mappings, Math. Appl., 495, Kluwer Academic Publishers, Dordrecht, 1999.
- [57] A. Granas and J. Dugundji, *Fixed point theory*, Springer-Verlag, New York, 2003.
- [58] I. Gyori and F. Hartung, Exponential stability of a state-dependent delay system, Discrete Contin. Dyn. Syst., 18 (4) (2007), 773-791.

- [59] J. K. Hale, Theory of functional differential equations, Springer-Verlag, New York, 1977.
- [60] J. Hale and J. Kato, Phase space for retarded equations with infinite delay, *Funkcial. Ekvac.*, **21** (1978), 11-41.
- [61] J. K. Hale, Theory of functional differential equations, Springer-Verlag, New York 1977.
- [62] J. K. Hale and S. M. Verduyn Lunel, Introduction to functional differential equations, Appl. Math. Sci., 99, Springer-Verlag, New York, 1993.
- [63] F. Hartung, Linearized stability for a class of neutral functional differential equations with state-dependent delay, Nonlinear Anal., 69 (5-6) (2008), 1629-1643.
- [64] F. Hartung, Differentiability of solutions with respect to parameters in neutral differential equations with state-dependent delay, J. Math. Anal. Appl., **324** (1) (2006), 504-524.
- [65] F. Hartung, Linearized stability in periodic functional differential equations with statedependent delay. J. Comput. Appl. Math., 174 (2) (2005), 201-211.
- [66] S. Heikkila and V. Lakshmikantham, Monotone iterative technique for nonlinear discontinuous differential equations, Marcel Dekker Inc., New York, 1994.
- [67] H. R. Henriquez, Existence of periodic solutions of neutral functional differential equations with unbounded delay, *Proyecciones*, **19** (3) (2000), 305-329.
- [68] E. Hernandez, Regularity of solutions of partial neutral functional differential equations with unbounded delay, *Proyecciones*, **21** (1) (2002), 65-95.
- [69] E. Hernández, A. Prokopczyk and L. Ladeira, A note on partial functional differential equations with state-dependent delay, *Nonlinear Anal. Real World Applications*, 7 (2006), 510-519.
- [70] E. Hernandez, R. Sakthivel and S. Tanaka Aki, Existence results for impulsive evolution differential equations with state-dependent delay, *Electron. J. Differential Equations*, 2008 (28) (2008), 1–11.
- [71] R. Hilfer, Applications of fractional calculus in physics, World Scientific, Singapore, 2000.
- [72] Y. Hino and S. Murakami, Total stability in abstract functional differential equations with infinite delay, *Electronic J. of Qualitative Theory of Diff. Equa.*, Lecture Notes in Math., 1473, Springer-Verlag, Berlin, (1991).
- [73] Y. Hino and S. Murakami, Total stability in abstract functional differential equations with infinite delay, Proceedings of the 6th Colloquium on the Qualitative Theory of Differential Equations (Szeged, 1999), 13, 9 pp. (electronic) Proc. Colloq. Qual. Theory Differ. Equ., Electron. J. Qual. Theory Differ. Equ., Szeged, 2000.
- [74] Y. Hino, S. Murakami and T. Naito, Functional differential equations with infinite delay, Lecture Notes in Math., 1473, Springer-Verlag, Berlin, 1991.
- [75] Sh. Hu and N. Papageorgiou, Handbook of multivalued analysis, Volume I: Theory, Kluwer, Dordrecht, Boston, London, 1997.

- [76] F. Kappel and W. Schappacher, Some considerations to the fundamental theory of infinite delay equations, J. Differential Equations, 37 (1980), 141-183.
- [77] M. Kamenskii, V. Obukhovskii and P. Zecca, Condensing multivalued maps and semilinear differential inclusions in Banach spaces, Walter de Gruyter & Co., Berlin, 2001.
- [78] A. A. Kilbas and S. A. Marzan, Nonlinear differential equations with the Caputo fractional derivative in the space of continuously differentiable functions, *Differential Equations*, 41 (2005), 84-89.
- [79] A. A. Kilbas, H. M. Srivastava and J. J. Trujillo, Theory and applications of fractional differential equations, North-Holland Math. Stud., 204, Elsevier Science B.V., Amsterdam, (2006).
- [80] M. Kisielewicz, *Differential inclusions and optimal control*, Kluwer, Dordrecht, The Netherlands, 1991.
- [81] V. Kolmanovskii and A. Myshkis, Introduction to the theory and applications of functionaldifferential equations, Math. Appl., 463, Kluwer Academic Publishers, Dordrecht, 1999.
- [82] V. Lakshmikantham, S. Leela and J. Vasundhara, Theory of fractional dynamic systems, Cambridge Academic Publishers, Cambridge, 2009.
- [83] V. Lakshmikantham, L. Wen and B. Zhang, Theory of differential equations with unbounded delay, Mathematics and its Applications, Kluwer Academic Publishers, Dordrecht, 1994.
- [84] S. G. Krein, Linear differential equations in Banach spaces, Amer. Math. Soc., Providence, 1971.
- [85] W. S. Li, Y. K. Chang and J. J. Nieto, Solvability of impulsive neutral evolution differential inclusions with state- dependent delay, *Math. Comput. Model.*, 49 (2009), 1920-1927.
- [86] Y. F. Luchko, M. Rivero, J. J. Trujillo and M. P. Velasco, Fractional models, non-locality and complex systems, *Comp. Math. Appl.*, **59** (2010), 1048-1056.
- [87] K. S. Miller and B. Ross, An introduction to the fractional calculus and differential equations, John Wiley, New York, 1993.
- [88] K. B. Oldham and J. Spanier, *The fractional calculus*, Academic Press, New York, London, 1974.
- [89] X. Li and J. Yong, Optimal control theory for infinite dimensional systems, Birkhauser, Berlin, 1995.
- [90] A. Pazy, Semigroups of linear operators and applications to partial differential equations, Springer-Verlag, New York, 1983.
- [91] I. Podlubny, Fractional differential equation, Academic Press, San Diego, 1999.
- [92] A.V. Rezounenko and J. Wu, A non-local PDE model for population dynamics with stateselective delay: Local theory and global attractors, J. Comput. Appl. Math., 190 (1-2) (2006), 99-113.

- [93] S. G. Samko, A. A. Kilbas and O. I. Marichev, Fractional integrals and derivatives. Theory and applications, Gordon and Breach, Yverdon, 1993.
- [94] K. Schumacher, Existence and continuous dependence for differential equations with unbounded delay, Arch. Rational Mech. Anal., 64 (1978), 315-335.
- [95] V. E. Tarasov, Fractional dynamics: Application of fractional calculus to dynamics of particles, Fields and media, Springer, Heidelberg; Higher Education Press, Beijing, 2010.
- [96] A. A. Tolstonogov, Differential inclusions in a Banach space, Kluwer Academic Publishers, Dordrecht, 2000.
- [97] A. N. Vityuk and A. V. Golushkov, Existence of solutions of systems of partial differential equations of fractional order, *Nonlinear Oscil.*, **7** (3) (2004), 318-325.
- [98] A. N. Vityuk, and A. V. Mikhailenko, On a class of fractional-order differential equations. Nonlinear Oscil., 11 (2008), 307-319.
- [99] D. R. Willé and C. T. H. Baker, Stepsize control and continuity consistency for statedependent delay-differential equations, J. Comput. Appl. Math., 53 (2) (1994), 163-170.
- [100] J. Wu, Theory and applications of partial functional differential equations, Appl. Math. Sci. 119, Springer-Verlag, New York, 1996.
- [101] K. Yosida, Functional analysis, 6th edn. Springer-Verlag, Berlin, 1980.
- [102] G. M. Zaslavsky, Chaos and fractional dynamics, 511 Lect. Notes in Phys., Oxford University Press, Oxford, 2005.
- [103] J. Zabczyk, Mathematical control theory, Birkhauser, Berlin, 1992.